MITSUBISHI





(Read these precautions before using this product.)

Before using this product, please read this manual and the relevant manuals carefully and pay full attention to safety to handle the product correctly.

The precautions given in this manual are concerned with this product only. For the safety precautions of the programmable controller system, refer to the user's manual for the CPU module used.

In this manual, the safety precautions are classified into two levels: " MARNING" and " A CAUTION".

WARNING	Indicates that incorrect handling may cause hazardous conditions, resulting in death or severe injury.
	Indicates that incorrect handling may cause hazardous conditions, resulting in minor or moderate injury or property damage.

Under some circumstances, failure to observe the precautions given under " A CAUTION" may lead to serious consequences.

Observe the precautions of both levels because they are important for personal and system safety. Make sure that the end users read this manual and then keep the manual in a safe place for future reference.

[Design Precautions]

<u>∕</u> NARNING
For the operating status of each station after a communication failure, refer to this manual.
Malfunction due to a communication failure may result in an accident.
 When controlling a running programmable controller (data modification) by connecting a peripheral device or GX Developer to a CPU module or by connecting a programmable controller to a special
function module, create an interlock circuit in the sequence program so that the entire system will function safely all the time.
For other forms of control (such as program modification or operating status change) of a running programmable controller, read the relevant manuals carefully and ensure that the operation is safe before proceeding.
Especially, when a remote programmable controller is controlled by an external device, immediate action cannot be taken if a problem occurs in the programmable controller due to a communication failure.
To prevent this, configure an interlock circuit in the sequence program, and determine corrective actions to be taken between the external device and CPU module in case of a communication
failure.
 Do not install the control lines or communication cables together with the main circuit lines or power cables. Keep a distance of 100mm (3.94 inches) or more between them. Failure to do so may result in malfunction due to noise.
[Installation Precautions]
Use the programmable controller in an environment that meets the general specifications in the

- Use the programmable controller in an environment that meets the general specifications in the user's manual for the CPU module used. Failure to do so may result in electric shock, fire, malfunction, or damage to or deterioration of the product.
- Fully insert the protection on the bottom of the module into the hole in the base unit and press the module into position.

(To fix an AnS series module to the base unit, tighten the screws within the specified torque range). Incorrect mounting may cause malfunction, failure or drop of the module.

- Shut off the external power supply (all phases) used in the system before mounting or removing the module. Failure to do so may result in damage to the product.
- Do not directly touch any conductive parts and electronic components of the module.
 Doing so can cause malfunction or failure of the module.

[Wiring Precautions]

WARNING

• Completely turn off the externally supplied power used in the system when installing or placing wiring.

Failure to do so may result in electric shock or damage to the product.

 Correctly solder coaxial cable connectors. Incomplete soldering may result in malfunction.
 Prevent foreign matter such as dust or wire chips from entering the module. Such foreign matter can cause a fire, failure, or malfunction.
 Place the cables in a duct or clamp them. If not, dangling cable may swing or inadvertently be pulled, resulting in damage to the module or cables or malfunction due to poor contact.
 When disconnecting the cable from the module, do not pull the cable by the cable part. For the cable with connector, hold the connector part of the cable. For the cable connected to the terminal block, loosen the terminal screw. Pulling the cable connected to the module may result in malfunction or damage to the module or

cable.

[Startup and Maintenance Precautions]

 Please read this manual thoroughly and confirm the safety before starting online operations (especially, program modifications, forced outputs, and operating status modifications), which are performed by connecting the GX Developer via the MELSECNET network system to a running CPU module of other station. Improper operation may damage machines or cause accidents.
 Do not disassemble or modify the module. Doing so may cause failure, malfunction, injury, or a fire.
 Use any radio communication device such as a cellular phone or PHS (Personal Handy-phone System) more than 25cm (9.85 inches) away in all directions from the programmable controller. Failure to do so may cause malfunction.
 Shut off the external power supply (all phases) used in the system before mounting or removing the module. Failure to do so may result in damage to the product.
 Do not touch any terminal while power is on. Doing so may cause malfunctions.
 Shut off the external power supply (all phases) used in the system before cleaning the module or retightening the terminal screws or module mounting screws. Failure to do so may cause the module to fail or malfunction. Undertightening can cause drop of the screw, short circuit or malfunction. Overtightening can damage the screw and/or module, resulting in drop, short circuit, or malfunction.
 Before handling the module, touch a grounded metal object to discharge the static electricity from the human body. Failure to do so may cause the module to fail or malfunction.
Disposal Precautions]

• When disposing of this product, treat it as industrial waste.

CONDITIONS OF USE FOR THE PRODUCT

(1) Mitsubishi programmable controller ("the PRODUCT") shall be used in conditions;

i) where any problem, fault or failure occurring in the PRODUCT, if any, shall not lead to any major or serious accident; and

ii) where the backup and fail-safe function are systematically or automatically provided outside of the PRODUCT for the case of any problem, fault or failure occurring in the PRODUCT.

(2) The PRODUCT has been designed and manufactured for the purpose of being used in general industries.

MITSUBISHI SHALL HAVE NO RESPONSIBILITY OR LIABILITY (INCLUDING, BUT NOT LIMITED TO ANY AND ALL RESPONSIBILITY OR LIABILITY BASED ON CONTRACT, WARRANTY, TORT, PRODUCT LIABILITY) FOR ANY INJURY OR DEATH TO PERSONS OR LOSS OR DAMAGE TO PROPERTY CAUSED BY the PRODUCT THAT ARE OPERATED OR USED IN APPLICATION NOT INTENDED OR EXCLUDED BY INSTRUCTIONS, PRECAUTIONS, OR WARNING CONTAINED IN MITSUBISHI'S USER, INSTRUCTION AND/OR SAFETY MANUALS, TECHNICAL BULLETINS AND GUIDELINES FOR the PRODUCT. ("Prohibited Application")

Prohibited Applications include, but not limited to, the use of the PRODUCT in;

- Nuclear Power Plants and any other power plants operated by Power companies, and/or any
 other cases in which the public could be affected if any problem or fault occurs in the PRODUCT.
- Railway companies or Public service purposes, and/or any other cases in which establishment of a special quality assurance system is required by the Purchaser or End User.
- Aircraft or Aerospace, Medical applications, Train equipment, transport equipment such as Elevator and Escalator, Incineration and Fuel devices, Vehicles, Manned transportation, Equipment for Recreation and Amusement, and Safety devices, handling of Nuclear or Hazardous Materials or Chemicals, Mining and Drilling, and/or other applications where there is a significant risk of injury to the public or property.

Notwithstanding the above, restrictions Mitsubishi may in its sole discretion, authorize use of the PRODUCT in one or more of the Prohibited Applications, provided that the usage of the PRODUCT is limited only for the specific applications agreed to by Mitsubishi and provided further that no special quality assurance or fail-safe, redundant or other safety features which exceed the general specifications of the PRODUCTs are required. For details, please contact the Mitsubishi representative in your region.

*The manual number is given on the bottom left of the back cover.

Print Date	*Manual Number	Revision
Dec., 1991	IB(NA)66350 - A	First edition
Jul., 1993	IB(NA)66350 - B	Correction CONTENTS, Section 1, 1.1, 1.2.2 to 1.2.4, 1.3.1, 1.3.2, 2.1.2, 2.2.2, 3.1.2, 3.2.2, 4, 4.1.1, 4.1.2, 5.2, 5.3, 5.3.1, 5.3.2, 5.3.7, 5.3.8, 5.5.1, 5.6, 6.2, 6.2.1 to 6.2.3, 6.3, 6.4, 8.8.2, 8.5.1, 8.5.2, 8.6, 8.7.2, 9.3.1, 10.3.1
Oct., 2003	IB(NA)66350 - C	Equivalent to Japanese version H Overall reexamination
Apr., 2007	IB(NA)66350 - D	Correction The following link modules are added. A1SJ71AP23Q/R23Q, A1SJ71AT23BQ Addition ABOUT MANUALS, COMPLIANCE WITH THE EMC AND LOW VOLTAGE DIRECTIVES, Section 1.2.5 Correction SAFETY PRECAUTIONS, Chapter 1, Section 1.1, 1.2.3, 1.3.1, 1.2.4, 1.3.2, 2.1.2, 2.2.3, 3.1.2, 3.2.3, 4.1.2, 4.1.3, 4.2.2, 4.2.3, 4.3.2, 4.3.3, 4.4.2, 4.4.3, 5.1, 5.2, 5.3.1, 5.3.2, 5.3.3, 5.3.8, 5.4.1, 5.5.1, 6.1.2, 6.2.1, 6.2.2, 6.2.3, 7.2, 7.4.2, 7.4.3, 7.5, 7.6.1, 7.6.3, 7.6.4, 7.7.1, 7.7.3, 7.8, 7.9.1, 7.9.2, 7.9.3, 7.9.4, 7.10.1, 7.10.2, 7.10.3, 7.10.4, 7.10.5, 7.11.1, 8.1, 8.3, 8.4, 8.4.1, 8.4.2, 8.4.3, 8.5.1, 8.5.2, 8.7.2, 8.7.3, Chapter 9, Section 9.1, 9.2.1, 9.2.2, 9.3.1, 9.3.2, 9.4, 9.5, 9.6, 9.7, 9.7.1, 9.8, 9.8.2, 9.9, 10.1.1, 10.1.2, 10.1.3, 10.3.1, 10.3.2, 10.3.3, 10.3.4, 10.4, 10.5
Oct., 2007	IB(NA)66350 - E	Correction Section 4.3.3, 5.2
Sep., 2010	IB(NA)66350 - F	The following CPU modules are added. Q02CPU-A, Q02HCPU-A, Q06HCPU-A Correction SAFETY PRECAUTIONS, Section 5.1, 5.3.1, 5.4.1, 5.5.2, 10.3.3, 10.4 Addition CONDITIONS OF USE FOR THE PRODUCT
Feb., 2012	IB(NA)66350 - G	Correction SAFETY PRECAUTIONS, COMPLIANCE WITH EMC AND LOW VOLTAGE DIRECTIVES, Section 5.1, 10.3.3

Japanese Manual Version IB-68277-N

This manual confers no industrial property rights or any rights of any other kind, nor does it confer any patent licenses. Mitsubishi Electric Corporation cannot be held responsible for any problems involving industrial property rights which may occur as a result of using the contents noted in this manual.

© 1991 MITSUBISHI ELECTRIC CORPORATION

INTRODUCTION

Thank you for choosing the Mitsubishi MELSEC-A Series of General Purpose Programmable Controllers. Before using the equipment, please read this manual carefully to develop full familiarity with the functions and performance of the A series programmable controller you have purchased, so as to ensure correct use.

CONTENTS

1 OVERVIEW 1 - 1 to 1 - 16
1.1 Contents of This Manual ••••••1 - 2
1.2 Basic Information about Data Link Systems ••••••••••••••••••••••••••••••••••••
1.2.1 Master, local, and remote I/O stations ••••••••••••••••••••••••••••••••••••
1.2.2 Outline of the MELSECNET and MELSECNET/B data link systems ••••••••••••••••1 - 4
1.2.3 Differences between the MELSECNET and MELSECNET/B data link systems ••••••••1 - 5
1.2.4 Differences among the MELSECNET mode, MELSECNET mode, and MELSECNET
composite mode ••••••1 - 6
1.2.5 Differences between QCPU and A/QnACPU local stations ••••••••••••••••••••••••••••••••••••
1.3 Applicable Link Modules and General Names ••••••••••••••••••••••••••••••••••••
1.3.1 Applicable link modules ••••••• 1 - 13
1.3.2 General names of CPU modules 1.3.2 General names of CPU modules
2 TWO-TIER SYSTEM IN THE MELSECNET DATA LINK SYSTEM 2 - 1 to 2 - 13
2.1 Outline of the MELSECNET Data Link System ••••••••••••••••••••••••••••••••2 - 1
2.1.1 Configuration of the data link system •••••••••••••••••••••••••••••••••••2 - 1
2.1.2 Features of the data link system •••••••2 - 3
2.2 MELSECNET Data Link System ••••••2 - 7
2.2.1 Overall configuration •••••••2 - 7
2.2.2 Precautions when operating the data link system •••••••••••••••••••••••2 - 9
2.2.3 Applicable modules ••••••• 2 - 10
3 TWO-TIER SYSTEM IN THE MELSECNET/B DATA LINK SYSTEM 3 - 1 to 3 - 8
3.1 Outline of the MELSECNET/B Data Link System ••••••••••••••••••••••••••••••••••••
3.1.1 Configuration of the data link system ••••••••••••••••••••••••••••••••••••
3.1.2 Features of the data link system ••••••••••••••••••••••••••••••••••••
3.2 MELSECNET/B Data Link System •••••••3 - 5
3.2.1 Overall configuration •••••••3 - 5
3.2.2 Precautions when operating the data link system ••••••••••••••••••••••••••••••••••••
3.2.3 System devices ••••••••••••••••••••••••••••••••••••
4 - 104 - 30
4.1 Three-Tier System Using the MELSECNET Data Link System ••••••••••••••••••••••••••••••••••••
4.1.1 System configuration ••••••4 - 2
4.1.2 Precautions when operating data link system ••••••••••••••••••••••••••••••••••••

4.2	Data	Link System when the Second Tier is MELSECNET and the Third Tier is MELSECNET/B
	•••••	······ 4 - 10
4.2	2.1	System configuration ••••••••••••••••••••••••••••••••••••
4.2	2.2	Precautions when using data link ••••••••••••••••••••••••••••••••••••
4.2	2.3	System devices ••••••••••••••••••••••••••••••••••••
4.3	Data	Link System when the Second Tier is MELSECNET/B and the Third Tier is MELSECNET
	•••••	
4.3	3.1	System configuration ••••••••••••••••••••••••••••••••••••
4.3	3.2	Precautions when operating the data link system ••••••••••••••••••••••••••••••••••••
4.3	3.3	System devices ••••••••••••••••••••••••••••••••••••
4.4	Thre	e-Tier System using the MELSECNET/B Data Link System ••••••••••••••••••••••••••••••••••••
4.4	4.1	System configuration ••••••••••••••••••••••••••••••••••••
4.4	4.2	Precautions when using data link ••••••••••••••••••••••••••••••••••••
4.4	4.3	System devices ••••••••••••••••••••••••••••••••••••

5 SPECIFICATIONS

5 - 1 to 5 - 42

5.1	5.1 General Specifications••••••••5 -		
5.2	5.2 Performance Specifications •••••••		· 2
5.3	Fun	ctions ••••••	- 4
5.	3.1	Cyclic transmission function ••••••5 -	· 6
5.	3.2	Transient transmission function •••••• 5 - 7	16
5.	3.3	Automatic return function •••••• 5 - 7	19
5.	3.4	Loopback function ••••••• 5 - 2	20
5.	3.5	Fault detection function ••••••• 5 - 2	24
5.	3.6	Self-diagnostics function •••••• 5 - 2	28
5.	3.7	Extensive use of link relays (B) and link registers (W) in a three-tier system ••••••••• 5 - 2	29
5.	3.8	MELSECNET mode and MELSECNET composite mode	32
5.4	Opti	cal Fiber Cable Specifications •••••• 5 - 3	36
5.	4.1	SI/GI-type optical fiber cable •••••• 5 - 3	36
5.5	Coa	xial Cable Specifications ••••••• 5 - 3	37
5.	5.1	Coaxial cable specifications	37
5.	5.2	Connector for the coaxial cable •••••• 5 - 3	38
5.6	Twis	sted Pair Cable Specifications •••••• 5 - 4	41

6 LINK DATA SEND/RECEIVE PROCESSING AND PROCESSING TIME

6 - 1 to 6 - 22

6.1	Link	Data Communication Processing ••••••6 - 1
6.	1.1	Communication processing outline ••••••6 - 1
6.	1.2	Link refresh execution timing ••••••6 - 2
6.	1.3	Link data during a communication error ••••••6 - 5
6.2	Trar	nsmission Delay Time in Two-Tier System••••••6 - 7
6.2	2.1	Transmission delay time in a two-tier system •••••••••••••••••••••••6 - 8
6.2	2.2	Link refresh time •••••• 6 - 12
6.2	2.3	Link data communication time (link scan) ••••••••••••••••••••••••••••••••••••
6.3	Trar	nsmission Delay Time in Three-Tier System ••••••••••••••••••••••••••••••••••••
6.4	Tim	e to Access Another Station from a Peripheral Device ••••••••••••••••••••••••••••••••••6 - 22

7 DATA LINK SETTINGS

7.1 Data	a Link Settings Overview •••••••7 - 1
7.2 Sett	ing of the Number of Modules ••••••7 - 2
7.3 Net	work Refresh Parameters •••••••7 - 4
7.4 Link	Parameters ••••••••••••••••••••••••••••••••••••
7.4.1	Link parameters to be set for the MELSECNET mode ••••••••••••••••••••••••••••••••••••
7.4.2	Link parameters to be set for the MELSECNET mode7 - 12
7.4.3	Link parameters to be set for the MELSECNET composite mode ••••••••••••••••••••••••••••••••••••
7.5 Mor	nitoring Time Setting ••••••• 7 - 17
7.6 Con	nmon Element ••••••••••••••••••••••••••••••••••••
7.6.1	Maximum number of link points per station ••••••••••••••••••••••••••••••••••••
7.6.2	Determining the link relay (B) assignment range ••••••••••••••••••••••••••••••••••••
7.6.3	Determining the link register (W) assignment range ••••••••••••••••••••••••••••••••••••
7.6.4	Determining the input (X) and output (Y) assignment range7 - 23
7.7 Link	Parameters in the MELSECNET Mode ••••••••••••••••••••••••••••••••••••
7.7.1	Local system assignment and link parameter setting example ••••••••••••••••••••••••••••••••••••
7.7.2	Remote I/O system assignment and link parameter setting example
7.7.3	Local/remote I/O system assignment and link parameter setting example 7 - 35
7.8 Link	Parameters in the MELSECNET Mode 7 - 42
7.9 Link	Parameters in the MELSECNET Composite Mode ••••••••••••••••••••••••••••••••••••
7.9.1	Local system assignment •••••• 7 - 48
7.9.2	Remote I/O system assignment •••••• 7 - 50
7.9.3	Local/remote I/O system assignment •••••• 7 - 52
7.9.4	Link parameter setting example •••••• 7 - 56
7.10 Thre	ee-Tier System Assignment ••••••• 7 - 61
7.10.1	Common element ••••••• 7 - 61
7.10.2	Using the MELSECNET mode in the second tier ••••••••••••••••••••••••••••••••••••
7.10.3	Using the MELSECNET mode in the second tier ••••••••••••••••••••••••••••••••••••
7.10.4	Using the MELSECNET composite mode in the second tier ••••••••••••••••••••••••••••••••••••
7.10.5	Link parameter setting example •••••• 7 - 72
7.11 Ass	ignment of Inputs and Outputs to the Master Station in a Remote I/O System •••••••••• 7 - 81
7.11.1	I/O assignment restrictions •••••• 7 - 81
7.11.2	I/O assignment example •••••• 7 - 83

8 PROCEDURES TO OPERATION

8 - 1 to 8 - 23

8.1	Pre	paratory Steps before Operation ••••••8 - 1
8.2	Sett	ing the Link Module Station Numbers ••••••8 - 2
8.2	2.1	Setting the link module station numbers in the MELSECNET data link system ••••••8 - 2
8.2	2.2	Setting the link module station numbers in the MELSECNET/B data link system ••••••8 - 5
8.3	Sett	ing Communication Speed ••••••8 - 7
8.4	Opti	ical Fiber Cable/Coaxial Cable Wiring •••••••8 - 8
8.4	4.1	Precautions for wiring ••••••8 - 8
8.4	4.2	Optical fiber cables connections •••••• 8 - 10
8.4	4.3	Coaxial cable connections •••••• 8 - 12
8.5	Shie	elded Twisted Pair Cable Wiring •••••• 8 - 14
8.	5.1	Precautions for wiring •••••• 8 - 14

8.5	5.2	Connecting shielded twisted pair cables
8.6	Star	tup Procedure •••••••••• 8 - 16
8.7	Self-	diagnostics Test ····································
8.7	7.1	Self-loopback test ••••••• 8 - 17
8.7	7.2	Station to station test •••••• 8 - 19
8.7	7.3	Forward loop test and reverse loop test ••••••• 8 - 21

9 PROGRAMMING

9 - 1 to 9 - 52

9.1	Precautions for Creating Programs ••••••9 -						
9.2	Spec	cial Link Relays •••••••••••••••9 - 4					
9.	2.1	Special link relays enabled only for the master station ••••••••••••••••••••••••••••••••••••					
9.	2.2	Special link relays enabled only for local stations ••••••9 - 9					
9.3	Spec	cial Link Registers ••••••••••••••••••••••••••••••••••••					
9.	3.1	Special link registers enabled only for the master station ••••••••••••••••••••••••••••••••••••					
9.	3.2	Special link registers effective only for local stations ••••••••••••••••••••••••••••••••••••					
9.4	Data	Link Program Using Link Inputs (X) and Link Outputs (Y) •••••••••••••••••••••••••••••••••••					
9.5	Data	Link Program Using Link Relays (B) •••••••••••••••••••••••••••••••••••					
9.6	Data	Link Program Using Link Registers (W) •••••••••••••••••••••••••••••••••••					
9.7	Read	d/Write Program for a Word Device from the Master Station to a Local Station •••••••• 9 - 36					
9.8	Read	d/Write Program from a Remote I/O Station to a Special Function Module					
9.	8.1	Read program (RFRP instruction) ••••••• 9 - 42					
9.	8.2	Write program (RTOP instruction) •••••• 9 - 46					
9.9	Fault	t Detection Program •••••••••• 9 - 50					

10 TROUBLESHOOTING

10 - 1 to 10 - 28

10.1 Link	Monitor by GPP ••••••••••••••••••••••••••••••••••
10.1.1	Master station link monitor ••••••••••••••••••••••••••••••••••••
10.1.2	Local station link monitor ••••••••••••••••••••••••••••••••••••
10.1.3	Remote I/O station link monitor ••••••••••••••••••••••••••••••••••••
10.2 Mor	nitoring the Special Link Relays and Special Link Registers ••••••••••••••••••••••••10 - 13
10.3 Proc	cedure for Troubleshooting ••••••10 - 14
10.3.1	Troubleshooting flowchart ••••••10 - 14
10.3.2	Flowchart for when "the data link is disabled throughout the entire system" ••••••10 - 15
10.3.3	Flowchart for when "the data link is disabled at a specific station" ••••••••••••10 - 17
10.3.4	Flowchart for when "the data communication error occurs" ••••••••••••••••10 - 19
10.3.5	Flowchart for when "the unspecified number of slave stations become faulty" ••••••10 - 24
10.4 ERF	ROR LED •••••••10 - 25
10.5 Rep	lacing a Faulty Station of Link Module ••••••10 - 26

Index - 1 to Index - 2

ABOUT MANUALS

The following manual is also related to this product. Order it by referring to the table below as necessary.

Related manual

Manual name	Manual No. (Model code)
MELSECNET, MELSECNET/B Local Station Data Link Module User's Manual This manual explains specifications, functions, preparatory procedures before operation, programming, and troubleshooting of the MELSECNET or MELSECNET/B local station data link module. (Sold separately)	SH-080670ENG (13JR98)

COMPLIANCE WITH EMC AND LOW VOLTAGE DIRECTIVES

(1) Method of ensuring compliance

To ensure that Mitsubishi programmable controllers maintain EMC and Low Voltage Directives when incorporated into other machinery or equipment, certain measures may be necessary. Please refer to one of the following manuals.

- User's manual for the CPU module used
- User's manual (hardware) for the CPU module or base unit used
- (2) Additional measures

To ensure that this product maintains EMC and Low Voltage Directives, please refer to one of the manuals listed under (1).

1 OVERVIEW

This manual describes the performance, functions, and programming procedure for the MELSEC-A Series MELSECNET, MELSECNET/B Data Link System.

When applying the following program examples to the actual system, make sure to examine the applicability and confirm that it will not cause system control problems.

(1) MELSECNET data link system

MELSECNET data link system is a system to connect link modules via optical fiber cable or coaxial cable and to control them.

The system can be configured in the MELSEC-A series.

The only difference between a system using optical fiber cable and a system using coaxial cable is a distance between link modules. Other functions/operations are the same.

(2) MELSECNET/B data link system

MELSECNET/B data link system is a system to connect link modules via inexpensive twisted pair cable and control them.

The system can be configured in the MELSEC-A series.

(3) Operation mode

The following three operation modes are available with the MELSECNET and MELSECNET/B data link system. The operation mode is determined by the configuration of the connected data link modules and the link parameter settings.

- MELSECNET mode
- MELSECNET I mode
- MELSECNET II composite mode

Refer to the manual for the individual link modules for details on the performance, functions, and operation of the link modules connected to the data link system.



1.1 Contents of This Manual

This manual explains the MELSECNET data link system and the MELSECNET/B data link system.

The MELSECNET data link system is explained in Chapter 2 and the MELSECNET/B data link system is explained in Chapter 3.

Other chapters explain both systems.

POINT

 To avoid misunderstanding, the table in the top right of a page indicates which system is being explained. In addition, when the MELSECNET data link and the MELSECNET/B data link partly differ, <u>the item mainly explains the MELSECNET data link, and the description unrelated to MELSECNET/B data link is underlined.</u> When using the MELSECNET/B data link, skip the underlined sentences/ phrases.
 This manual is described based on the screen examples of the SW□GP-GPPA.

When using GX Developer or any other peripheral, some of the screen displays or functions may be different from those of this manual. Refer to the manual for the product in use.

MELSEC-A

1.2 Basic Information about Data Link Systems

This section gives the basic information for using the MELSECNET and the MELSECNET/ B data link systems.

Read this section carefully before going on to Chapter 2.

1.2.1 Master, local, and remote I/O stations

In a data link system, link modules are classified into master stations, local stations, and remote I/O stations as shown below:

(1)	Master stations	Controls slave stations (local stations and remote I/O stations) connected to data link system. Sets link parameter for the MELSECNET data link system and the MELSECNET/B data link system. A data link system must contain one master station.
(2)	Local stations	Controls I/O and special function modules of the host station in a program of the host station.
(3)	Remote I/O stations	Controls the I/O module and special function module of the host station in a program of master station via MELSECNET(/B). As for the I/O module and special function module connected to remote I/O station, online I/O module cannot be replaced.

POINTS

(1)	Set the link parameters for the MELSECNET and MELSECNET/B data link
	systems to the master stations only.
	The link parameters need not be set to the local and remote I/O stations.
(2)	When the MELSECNET and MELSECNET/B data link systems are used with
	the QnA or AnUCPU, the "number of modules" and "network refresh
	parameters" must be set to the master and local stations.
	For the network parameter setting method, refer to the operating manual of
	the used GPP function software package or GX Developer.
	In addition, the QnACPU can confirm the contents of the link special relays
	(M9200 to M9255) and link special registers (D9200 to D9255) using the
	special relays (SM1200 to SM1255) and special registers (SD1200 to
	SD1255).
(3)	The software versions of the master station link modules that allow remote I/O
	stations to be connected in the MELSECNET/B data link systems are
	indicated below.
	A1SJ71T21B "B" or later
	A1SJ71AT21B "A" or later
	AJ71AT21B "B" or later

REMARK

1) *: Online I/O module replacement means replacing an I/O module without occurrence of "UNIT VERIFY ERROR" while the programmable controller is ON.

1.2.2 Outline of the MELSECNET and MELSECNET/B data link systems

Use a MELSEC-A series programmable controller CPU to establish a data link system via optical fiber, coaxial or twisted-wire pair cables.

The MELSECNET data link system is a system to connect link modules via optical fiber cables or coaxial cables.

The MELSECNET/B data link system is a system to connect link modules via twisted-wire pair cables.

(1) MELSECNET data link system

In the MELSECNET data link systems, up to 64 slave stations (local and remote I/O stations) can be connected to a master station.

By using optical fiber or coaxial cables, connect slave station No.1 to the master station and slave station No. 2 to slave station 1 ..., creating a loop system until station n is connected to the master station.



(2) MELSECNET/B data link systems

In the MELSECNET/B data link system, up to 31 slave stations (local and remote I/O stations) can be connected to the master station.

There is no restriction on order of connection and station order of the master station, local station and remote I/O station.

In addition, it is unnecessary to create a loop system, since each station has electrical connection.



1.2.3 Differences between the MELSECNET and MELSECNET/B data link systems

The following table shows differences between the MELSECNET and MELSECNET/B data link systems.

Table 1	.1	Differences	between	the MEL	SECNET	data lini	<pre>svstem</pre>	and ME	LSECNET/B	data	link s	svstem
												, ,

		MELSECNET dat	MELSECNET/B data link system		
	Optical data	link system			
Item	SI or H-PCF type optical fiber cable	GI type optical fiber cable	Coaxial cable data link system	Shielded twisted pair cable data link system	
Communication speed		1.25MB	125kBPS/250kBPS/500kBPS/1MBPS		
Transmission path		Duplex I	оор	Bus system	
Overall cable distance	Max.10 km (Between stations: 1 km)	Max.10 km (Between stations: 2 km)	Max. 10km (Between stations: 500m)	Varies depending on the communications speed 125kBPS: 1200m 250kBPS: 600m 500kBPS: 400m 1MBPS: 200m	
Number of stations	Up to 65 (1	master station, 64	Up to 32 (1 master station, 31 local/ remote I/O stations)		
Modulation method		CMI met	NRZI method		
RAS functions	The loopback fund disconnection, and of the host station	ction activated by d d the diagnostic fur	The diagnostic function such as a link line check of the host station.		
Cable used	Optical fi	ber cable	Coaxial cable	Shielded twisted pair cable	
Transmission loss	Max.12 dB/km	Max.3 dB/km	-	-	
Sending level	-17 to -11dBm -17 to -10dBm (peak value) (peak value)		-	-	
Receiving level	-32 to -11dBm (peak value)	-29 to -10dBm (peak value)	-	-	

1.2.4 Differences among the MELSECNET mode, MELSECNET I mode, and MELSECNET I composite mode

The data link system has the MELSECNET mode, the MELSECNET II mode, or the MELSECNET II composite mode.

In the MELSECNET mode, the data link system operates within the range of B/W0 to 3FF regardless of the link module used for the master station.

In the MELSECNET II mode, the following link modules are used as the master and local stations and B/W400 and later are used for data link.

- AnACPUP21/R21
- AnACPU+AJ71AP21/R21/T21B
- AnUCPU+AJ71AP21/R21/T21B
- QnACPU+AJ71AP21/R21/T21B
- A2ASCPU(S1)+A1SJ71AP21/R21/T21B
- A2USHCPU-S1+A1SJ71AP21/R21/T21B
- Q2AS(H)CPU(S1)+A1SJ71AP21/R21/T21B
- QCPU-A+A1SJ71AP21/R21/T21B
- QCPU+A1SJ71AP23Q/R23Q/T23BQ(Local station only)

The following describes the major differences between the MELSECNET mode and the MELSECNET II mode:

- The data link device range has been increased from B/W0 to 3FF (1024 points) to B/W0 to FFF (4096 points) in the MELSECNET II mode.
- The maximum number of link points per station (master station and local station) has been increased from 1024 bytes to 2048 bytes in the MELSECNET II mode.
- Connection of remote I/O stations is impossible in the MELSECNET II mode.

The MELSECNET I composite mode has the almost same function as that of the MELSECNET I mode, and it is a mode which can connect to link module and remote I/O station of the MELSECNET mode.

The following describes the major differences among the three modes:

- (1) Link modules that can be used as the master station (Refer to Table 1.2 for the link module model name.)
 - (a) MELSECNET mode

MELSECNET mode-compatible data link modules

(A MELSECNET II mode-compatible data link module can also be used as the master station for MELSECNET mode.)

- (b) MELSECNET I mode MELSECNET I mode-compatible link modules
- (c) MELSECNET II composite mode MELSECNET II mode-compatible data modules

- (2) Link modules that can be used as the local station (Refer to Table 1.2 for the link module model name.)
 - (a) MELSECNET mode MELSECNET mode- and MELSECNET II mode-compatible data link modules
 - (b) MELSECNET II mode MELSECNET II mode-compatible data link modules
 - (c) MELSECNET II composite mode MELSECNET mode- and MELSECNET II mode-compatible data link modules
- (3) Possibility of connecting remote I/O stations
 - (a) MELSECNET mode Possible to connect
 - (b) MELSECNET I mode Impossible to connect
 - (c) MELSECNET I composite mode Possible to connect

(4) Data link device range

The following describes the data link device range for each operation mode. The data link range for inputs (X) and outputs (Y) is determined by the data link module used as the master station. The maximum value is indicated below. For example, if an A2NCPU or A2ACPU is used as the master station, the X/Y range that can be used for data link is 0 to 1FF (512 points).

- (a) MELSECNET mode
 - X/Y : 0 to 7FF (2048 points)
 - B : 0 to 3FF (1024 points)
 - W : 0 to 3FF (1024 points)

The data link range for link relays (B) and link registers (W) is B/W0 to 3FF (1024 points) when a MELSECNET II compatible data link module is connected to the local station.

(b) MELSECNET II mode

X/Y: 0 to 7FF (2048 points)

- B : 0 to FFF (4096 points)
- W : 0 to FFF (4096 points)
- (c) MELSECNET II composite mode
 - X/Y: 0 to 7FF (2048 points)
 - B : 0 to FFF (4096 points)
 - W : 0 to FFF (4096 points)

The data link range for link relays (B) and link registers (W) that can be data linked by the MELSECNET-compatible local station is B/W0 to 3FF (1024 points).

- (5) Link parameter types and the number of link points per station
 - (a) MELSECNET mode
 - 1) Link parameter is only one type.
 - 2) Maximum number of link points per station
 Master station and local station : 1024 bytes/station
 Remote I/O station : 512 bytes/station (I/O is 512 points of X/Y0 to 1FF per station)
 - (b) MELSECNET I mode
 - Link parameter
 Link parameter is divided into two types (first/second half link parameter).
 Data link is also possible by setting the first half link parameter only.
 - 2) Maximum link points per station
 - For stations with only first half link parameters set: Master station and local station: 1024 bytes/station
 - For stations with both first and second link parameters set: Master station and local station: 2048 bytes/station
 - (c) MELSECNET II composite mode
 - Link parameter
 Link parameter is divided into two types (first/second half link parameter).
 (When setting first half link parameter only, its function is the same as that of the MELSECNET mode.)
 - First half link parameters should be assigned to master stations, local stations, and remote I/O stations.
 - Second half link parameters should be assigned to master and local stations that are compatible with MELSECNET II mode only.
 Second half link parameters cannot be assigned to remote I/O stations or MELSECNET mode-compatible local stations.
 - 2) Maximum link points per station
 - For stations with only first half link parameters set: Master station and local station: 1024 bytes/station
 Remote I/O station : 512 bytes/station (I/O is 512 points of X/Y0 to 1FF per station)
 - For stations with both first and second link parameters set: Master station and local station: 2048 bytes/station

The information mentioned above (1) to (5) is summarized in Table 1.2.

Table 1.2 MELSECNET (II) Data Link Function Overview
--

ltem			Operation mode						
			MELSECNET data link system						
	nem		MELSECNET mode	MELSEC	CNET II te mode	MELSECN	ET II mode		
Data li can be use	nk modules tha d as a master s	at station	A0J2HCPUP21/R21 AnNCPUP21/R21 AnNCPUP21-S3 A2NCPUP21-S4 AnACPUP21-S4 AnACPUP21-S3 A2ACPUP21-S4 AnACPUP4J71AP21(S3)/R21 AnUCPU+AJ71AP21(S3)/R21 AnUCPU+AJ71AP21(S3)/R21 AnSCPU+A1SJ71AP21/R21 AnASCPU+A1SJ71AP21/R21 QnASCPU+A1SJ71AP21/R21 QCPU-A+A1SJ71AP21/R21 A80BD-A2USH-S1+A1SJ71AP21/R21 A2CCPUP21/R21	AnACPUP21/R21 AnACPUP21-S3 A2ACPUP21-S4 AnACPU+AJ71AP21(S3)/R21 AnUCPU+AJ71AP21(S3)/R21 QnACPU+AJ71AP21(S3)/R21 AnASCPU+A1SJ71AP21/R21 QnASCPU+A1SJ71AP21/R21 QCPU-A+A1SJ71AP21/R21 A80BD-A2USH-S1+A1SJ71AP21/R21					
Data link modules that can be used as a local station			A0J2HCPUP21/R21 AnNCPUP21/R21 AnNCPUP21-S3 A2NCPUP21-S4 Computer+A70BD-J71AP22 PC9800+A98BD-J71AP22 AnACPUP21/R21 AnACPUP21-S3 A2ACPUP21-S4	AnACPU+AJ71AP21(S3)/R21 AnUCPU+AJ71AP21(S3)/R21 QnACPU+AJ71AP21(S3)/R21 AnSCPU+A1SJ71AP21/R21 AnASCPU+A1SJ71AP21/R21 QnASCPU+A1SJ71AP21/R21 QCPU-A+A1SJ71AP21/R21 QCPU+A1SJ71AP23Q/R23Q A80BD-A2USH-S1+ A1SJ71AP21/R21 A2CCPUP21/R21		AnACPUP21/R21 AnACPUP21-R3 A2ACPUP21-S3 A2ACPUP21-S4 AnACPU+AJ71AP21(S3)/R21 AnUCPU+AJ71AP21(S3)/R21 QnACPU+AJ71AP21(S3)/R21 AnASCPU+A1SJ71AP21/R21 QCPU-A+A1SJ71AP21/R21 QCPU+A1SJ71AP23Q/R23Q A80BD-A2USH-S1+ A1SJ71AP21/R21			
Connection	of remote I/O s	stations	Possible	Pos	sible	Impo	ssible		
		X/Y	0 to 7FF	0 to 7FF		0 to 7FF			
	Parameter (first half)	В	0 to 3FF	0 to	3FF	0 to FFF			
Device	(III St Hall)	W	0 to 3FF	0 to	3FF	0 to FFF			
range	Parameter	X/Y	-		-	-			
	(second	В	-	α to	α to FFF		α to FFF		
	half)	W	-	α to	FFF α to F	FFF			
Link parameter type			1 type	2 types (Link parameters, first and second halves)		2 types (Link parameters, first and second halves)			
Max. number of link points per station	Master/lc statior	ocal า	1024 bytes/station	Setting of first half only Setting of both first and second halves	1024 bytes/ station 2048 bytes/ station	Setting of first half only Setting of both first and second halves	1024 bytes/ station 2048 bytes/ station		
	Remote I/O station		512 bytes/station (I/O: 512 points (X/Y0 to 1FF) /station)				-		

Operation mode							
MELSECNET/B data link system							
MELSECNET mode	MELSECNET II composite mode		MELSECNET I mode				
AnNCPU+AJ71AT21B AnACPU+AJ71AT21B AnUCPU+AJ71AT21B QnACPU+AJ71AT21B AnSCPU+A1SJ71AT21B AnASCPU+A1SJ71AT21B QnASCPU+A1SJ71AT21B QCPU-A+A1SJ71AT21B A80BD-A2USH-S1+A1SJ71AT21B		AnACPU+/ AnUCPU+/ QnACPU+/ AnASCPU+A QnASCPU+A QCPU-A+A A80BD-A2USH-S	AJ71AT21B AJ71AT21B AJ71AT21B A1SJ71AT21B A1SJ71AT21B A1SJ71AT21B S1+A1SJ71AT21B S1+A1SJ71AT21B				
AnNCPU AnACPU AnUCPU QnACPU+, AnSCPU+, QnASCPU- QCPU-A+, QCPU-A+, QCPU+A1 A80BD-A2USH		AnACPU+AJ71AT21B AnUCPU+AJ71AT21B QnACPU+AJ71AT21B AnASCPU+A1SJ71AT21B QnASCPU+A1SJ71AT21B QCPU-A+A1SJ71AT21B QCPU-A+A1SJ71AT23BQ A80BD-A2USH-S1+A1SJ71AT21B					
Possible	sible	Impossible					
0 to 7FF	0 to	7FF	0 to 7FF				
0 to 3FF	0 to	3FF	0 to FFF				
0 to 3FF	0 to	3FF	0 to FFF				
-	-	-	-	-			
-	FFF	α to FFF					
-	FFF	α to FFF					
1 type 2 types (Link parameters, first and second halves)			2 types (Link parameters, first and second halves)				
 1024 bytes/station	Setting of first half only	1024 bytes/station	Setting of first half only	1024 bytes/station			
	both first 2048 and second bytes/station halves		both first and second halves	2048 bytes/station			
512 bytes/station (I/O :512 points (X/Y0 to 1FF) /station)							

Table 1.2 MELSECNET (${\rm I\hspace{-0.5mm}I}$) Data Link Function Overview (Continued)

 α : "The last number in first half" + 1 ("0" if the first half range is "0".)

1.2.5 Differences between QCPU and A/QnACPU local stations

 When using an QCPU local station This module replaces or adds a part of the existing MELSECNET (II) data link system to QCPU.



(2) Differences between QCPU and A/QnACPU local stations When replacing an A/QnACPU local station with a QCPU one, pay attention to the following point.

For details, refer to the MELSECNET, MELSECNET/B Local Station Data Link Module User's Manual.

ltem	Description			
	QCPU local station	A/QnACPU local station		
Link refresh	Refreshes data with the sequence program. Automatically refreshes data at either of the Upon completion of link scan Link parameter setting (refresh parameters) is not required. • Only after execution of the END instruction program For the AnUCPU, QnACPU, A2US(H)CPU	 Automatically refreshes data at either of the following timing. Upon completion of link scan Only after execution of the END instruction in the sequence program For the AnUCPU, QnACPU, A2US(H)CPU(S1), 		
	Does not refresh data when the CPU module is in STOP status.	Q2AS(H)CPU(S1) and QCPU-A, refresh ranges can be changed with refresh parameters.		
Operation after power OFF → ON, or resetting CPU module (CPU module is in STOP.)	Starts data communication with other stations by executing the program for refresh (Y10 = ON) with the CPU module set to RUN. Until then, the master station treats the QCPU local station as a faulty station (relevant bit in D9228 to D9231 is turned ON).	Starts data communications with other stations.		

Table 1.3 Differences between QCPU and A/QnACPU local stations

Itom	Description	
liem	QCPU local station	A/QnACPU local station
	Handles the received instruction with the sequence program.	
LRDP/LWTP instruction receive processing	If the LRDP/LWTP instruction is received when the CPU module is in STOP status, sends an error response to the master station (4: LRDP/LWTP inexecutable on the station).	The system handles the received instruction. (The program for receiving LRDP/LWTP instruction is not required.)
Access from peripheral to host station	The following are not available. Buffer memory batch monitor/test Network diagnostics of GX Developer 	Not particularly restricted.
Access from peripheral to other stations *1	 Unable to access other stations. Master station → QCPU local station QCPU local station → Master station 	 The master station can access A/QnACPU local stations. A/QnACPU local stations can access the master station.
Network diagnostics of GX Developer	Unable to use the network diagnostics of GX Developer. The data link status or fault location can be checked by refreshing the special relay (for link) and special register (for link) of the local module into CPU module devices.	Can use the network diagnostics of GX Developer.
Forward loop test Reverse loop test	Place the QCPU local station into RUN status (Y10 = ON) to conduct the test. If the test is conducted in STOP status (Y10 = OFF), the master station treats the QCPU local station as a faulty station (relevant bit in D9228 to D9231 is turned ON). However, the test is normally conducted.	Place the A/QnACPU local station into STOP status to conduct the test.
	LRDP instruction receive request (Buffer memory address: ACH)	LRDP instruction completion (M9204)
Replacement for special relay (for link)	LWTP instruction receive request (Buffer memory address: AEH)	LWTP instruction completion (M9205)
	Hardware failure (RUN LED: OFF)	Link card failure (M9211)
	Link status (X0)	Link status (M9240)

*1 When replacing with a QCPU local station, the following alternative solution can be applied to GOT communications.

Table 1.4 Alternative solut	ion for GOT com	munications
-----------------------------	-----------------	-------------

Before replacement	Alternative solution
GOT is connected to master station to access A/QnACPU local station.	Send/receive the link data of the devices that are used for access from the GOT.
GOT is connected to A/QnACPU local station to access master station.	 If the number of link points is insufficient, install another local module to the QA1S6 B extension base unit where the QCPU local station is mounted.

1.3 Applicable Link Modules and General Names

1.3.1 Applicable link modules

This manual explains the link modules shown below.

The link modules compatible with MELSECNET ${\rm I\!I}\,$ mode also support the MELSECNET mode.

MELSECNET data link system

- (1) CPU modules
 - (a) Link modules compatible with MELSECNET mode A0J2HCPUP21/R21 A1NCPUP21/R21 A1NCPUP21-S3 A2NCPUP21/R21(S1) A2NCPUP21-R21-S1 A2NCPUP21/R21-S1 A2NCPUP21-S4 A3NCPUP21/R21 A3NCPUP21/R21
 (b) Link modules compatible with MELSECNET To med
 - (b) Link modules compatible with MELSECNET II mode A2ACPUP21/R21(S1) A2ACPUP21-S3 A2ACPUP21/R21-S1 A2ACPUP21-S4 A3ACPUP21/R21 A3ACPUP21-S3
- (2) The following link modules are installed to a base unit.
 - (a) Link modules compatible with MELSECNET mode (Installed in the CPU slot) AJ72P25/R25 (for remote I/O station) AJ72P25-S3 (for remote I/O station)
 - (b) Link modules compatible with MELSECNET II mode (Installed in an I/O slot) AJ71AP21/R21 (for master station/local station) A1SJ71AP21/R21 (for master station/local station) AJ71AP21-S3 (for master station/local station)
- (3) The following link modules are used independently, without being installed to a base unit.
 - (a) Link modules compatible with MELSECNET mode A0J2P25/R25 (for remote I/O station) A0J2P25-S3 (for remote I/O station)

MELSECNET/B data link system

- (1) The following link modules are installed to a base unit.
 - (a) Link module compatible with MELSECNET mode (Installed in the CPU slot) A1SJ72T25B (for remote I/O station) AJ72T25B (for remote I/O station)
 - (b) Link module compatible with MELSECNET I mode (Installed in an I/O slot) A1SJ71AT21B (for master station/local station) AJ71AT21B (for master station/local station)

POINT

Although the following link modules can be connected to the MELSECNET data link system, they are not discussed in this manual. Graphic operation terminal A6BSW-S3, S4, S5 (bypass switch) Personal computer board Refer to the manual for these link modules for details.

1.3.2 General names of CPU modules

This manual describes CPU modules using the following generic names.

(1)	AnNCPU	
	AnNCPU is general	name for the following link modules:
	1) A1NCPU	3) A2NCPU-S1
	2) A2NCPU	4) A3NCPU

(2)	AnACPU	
	AnACPU is generation	al name for the following link modules:
	1) A2ACPU	3) A3ACPU
	2) A2ACPU-S1	

(3) AnUCPU

AnUCPU is general name for the following link modules:

T) AZUCPU	3) A3UCPU
2) A2UCPU-S1	4) A4UCPU

(4) QnACPU

QnACPU is general name for the following link modules:

1) Q2ACPU	3) Q3ACPU
2) Q2ACPU-S1	4) Q4ACPU

(5) ACPU

ACPU is general name for all the CPUs listed in items (1) to (4), A0J2HCPU and A2CCPU.

(6) AnSCPU

AnSCPU is general name for the following link modules: 1) A1SJHCPU 3) A2SHCPU 2) A1SHCPU

(7) AnASCPU

AnASCPU is general name for the following link modules: 1) A2ASCPU 3) A2USHCPU-S1 2) A2ASCPU-S1

(8) QnASCPU

QnASCPU is general name for the following link modules:

 1) Q2ASCPU
 3) Q2ASHCPU

 2) Q2ASCPU-S1
 4) Q2ASHCPU-S1

(9) AnNCPUP21/R21

AnNCPUP21/R21 is general name for the following link modules:

- 1) A1NCPUP21/R21 5) A2NCPUP21-S3
- 2) A1NCPUP21-S3 6) A2NCPUP21-S4
- 3) A2NCPUP21/R21 7) A3NCPUP21/R21
- 4) A2NCPUP21/R21-S1 8) A3NCPUP21-S3

(10) AnACPUP21/R21

AnACPUP21/R21 is general name for the following link modules:
1) A2ACPUP21/R21 4) A2ACPUP21-S4
2) A2ACPUP21/R21-S1 5) A3ACPUP21/R21

- 3) A2ACPUP21-S3 6) A3ACPUP21-S3
- 3) AZACFUF21-33 0) ASACFU

(11) QCPU-A

QCPU-A is general name for the following CPU modules:1) Q02CPU-A3) Q06HCPU-A2) Q02HCPU-A

(12) QCPU

QCPU is general name for the following CPU modules:

- 1) Q02CPU 4) Q12HCPU
- 2) Q02HCPU 5) Q25HCPU
- 3) Q06HCPU

2 TWO-TIER SYSTEM IN THE MELSECNET DATA LINK SYSTEM

This section describes the MELSECNET data link system.

- 2.1 Outline of the MELSECNET Data Link System
- 2.1.1 Configuration of the data link system

The MELSECNET data link system connects link modules via optical fiber or coaxial cables.

In the MELSECNET data link system, up to 64 slave (local and remote I/O) stations can be connected to a link module used as the master station.

There is no restriction on combinations of local and remote I/O stations.



(1) Master station

The master station is the link module which controls the whole MELSECNET data link system.

The total number of slave stations (up to 64) connected to MELSECNET data link system and the device (B, W, X, Y) to perform data link are set to the programmable controller CPU of the master station using link parameter.

The master station controls data communications in the MELSECNET data link system by using the set link parameters.

(2) Slave stations

There are two kinds of slave stations: local stations and remote I/O stations.

(a) Local stations

When two or more programmable controller CPUs are used for data link, local stations are used to increase the number of I/O points and the program capacity in a large-scale system.

(b) Remote I/O stations

Remote I/O stations are used to reduce wiring costs when data must be frequently input/output from/to devices that are far away from a programmable controller CPU.

The programmable controller CPU in the master station controls the input and output of remote I/O stations.

The number of I/O points is 512 points (X/Y0 to 1FF) per station.

REMARK

Master stations, local stations, and remote I/O stations are expressed in the following symbols.

- Master station.....M
- Local station.....L (Local station No. n: Ln)
- Remote I/O station......R (remote I/O station No. n: Rn)

MELSEC-A

2.1.2 Features of the data link system

The following describes features of MELSECNET data link system.

(1) Cyclic transmission function

The cyclic transmission is a function to periodically communicate data between a master station and slave stations (local stations and remote I/O stations). Either 1 : n or 1 : 1 data communications are enabled by the cyclic transmission.

(a) 1 : n data communications

This is data communications between the master station and all local stations/ between local stations.

ON/OFF data and 16-bit data can both be communicated:

- 1) ON/OFF data is communicated using the link relay (B).
- 2) 16-bit data is communicated using the link register (W).



Fig 2.1 Flow of data by B/W communication

(b) 1:1 data communications

This is 1 : 1 data communications between the master station and a local station/ between the master station and a remote I/O station.

ON/OFF data can be communicated using inputs (X) and outputs (Y).



Fig 2.2 Flow of input (X) and output (Y) communication data

(2) Transient transmission function

The transient transmission function reads/writes data from/to a device in a local station by the master station programmable controller CPU or communicates data between a peripheral device connected to a programmable controller CPU and a programmable controller CPU in another station.

The transient transmission includes the following.

(a) Communication between master station and local station The device (T, C, D, W) of a local station is read/written from the programmable controller CPU of the master station.

The master station uses LRDP and LWTP^{*1} instructions in a sequence program.

(b) Communication between master station and remote I/O station The content of buffer memory of the special function module connected from remote I/O station is read/written from the programmable controller CPU of the master station.

The master station uses RFRP and RTOP^{*1} instructions in a sequence program.

(c) Access of peripheral device and special function module to/from other station Access of the peripheral device and special function module connected to the programmable controller CPU to/from other station is performed. The accessible station varies as shown in Table 2.1, depending on which one of the master station, local station and remote I/O station is connected to peripheral device or mounts special function module.

For the function executable in the peripheral device and special function module, refer to the manual for the peripheral device and special function module in use.

Access target station	Station where peripheral device is connected		
Access larger station	Master Station	Local Station	Remote I/O Station
Master Station	Accessible	Accessible	Accessible
Local Station	Accessible	Not accessible	Not accessible
Remote I/O Station	Accessible*2	Not accessible	Not accessible

 Table 2.1 Accessible Stations With a Peripheral Device

REMARK

* 1

* 2

 When the AnACPU(P21/R21), AnUCPU, A2US(H)CPU(S1) or QCPU-A is used, the LRDP/LWTP/RFRP/RTOP instructions of the dedicated instructions can also be used.

(For details of the instructions, refer to the Type AnSHCPU/AnACPU/ AnUCPU/QCPU-A (A Mode) Programming Manual (Dedicated Instructions) IB-66251.)

- When the QnACPU, Q2AS(H)CPU(S1) is used, the ZNRD/ZNWR/ RFRP/RTOP instructions of the data link instructions can also be used. (For details on the instructions, refer to the QnACPU Programming Manual (Common Instructions).)
- Not accessible when using GX Developer.

MELSEC-A

- (3) Improved RAS (Reliability, Availability, Serviceability) functions
 - (a) Loopback function

If a cable breaks or the power supply to a slave station is turned off, the affected slave station is disconnected from the data link so that normal link operations are able to continue for other stations.

In the MELSECNET data link system, the optical fiber cables or coaxial cables are doubled to make the loopback function possible. This double configuration allows data link operations to continue by switching the loop from forward to reverse if a cable breaks or the power to a slave station is turned off. (Refer to Section 5.3.4.)

(b) Automatic return function

If an error occurs in a slave station, the corresponding station is disconnected from the data link system.

When a slave station (local station, remote I/O station) is disconnected from the link due to an occurrence of a problem, it is automatically connected into the link when the station recovers the normal operating conditions. (Refer to Section 5.3.3.)

- (c) Error detection
 - The data link operation status is stored in special relays (M) and special registers (D) in a programmable controller CPU. Read these special relays (M) and special registers (D) to check the data link operation status.
 - 2) Use the link monitor function of a peripheral device (A6GPP, A6PHP, A6HGP) to check the data link operation status.
- (d) Self-diagnostics function

The self-diagnostics function checks the link module hardware, the optical fiber cable or coaxial cable connection status, etc.

REMARK

The errors that make the RAS functions valid are only cable breakage, slave station power-off, data link setting error, and the errors that can be detected by the self-diagnostics of the CPU module.

The RAS functions may not work depending on the fault of the data link module.

(4) Using MELSECNET mode and MELSECNET I mode-compatible modules in the same network

The link module which can be connected to MELSECNET data link system includes the MELSECNET mode-compatible link module and the MELSECNET II mode-compatible link module.

The MELSECNET data link system even allows MELSECNET mode-compatible modules and MELSECNET II mode-compatible modules to be used in the same network.

The MELSECNET ${\rm I\!I}$ mode-compatible link module can also be connected to the data link system of the operating MELSECNET mode.

The MELSECNET data link system has parameters for setting its operation mode so that various combinations of system configuration are possible.

The operation mode includes MELSECNET mode, MELSECNET II mode, and MELSECNET II composite mode.

Refer to Section 1.2 for details on the differences between these modes.

MELSEC-A

2.2 MELSECNET Data Link System

2.2.1 Overall configuration

(1) Two-tier system

In a two-tier system, up to 64 stations (local and remote I/O stations) can be connected to the master station with optical fiber cable or coaxial cable. In a two-tier system, a master station is referred to as "master station" and a local station or a remote I/O station is referred to as "slave station".

(2) System configuration

(a) Fig 2.3 shows the configuration of the two-tier system.








(b) The following table shows the configurations of the master station/local station and the remote I/O station.

2.2.2 Precautions when operating the data link system

The following describes the precautions for performing data link.

 Optical fiber cable and coaxial cable cannot be mixed in the same loop. The same type of cable must be used for links within the same loop of the two-tier system. Optical fiber cable and coaxial cable cannot be mixed.



Fig 2.4 Possibility of connection between link modules

(2) Number of link device points

The number of link device points (X, Y, B, W) that can be used for each local or remote I/O station is limited. (For details, refer to Section 7.3.)

(3) Link parameter settings

Set the link parameters in the master station to operate data link. The link parameters include the number of slave stations connected to the MELSECNET data link system, the link device assignments, and the monitoring time.

(a)	Number of slave stations	The total number of local and remote I/O stations connected to the MELSECNET data link system.
(b)	Link device assignment	It is a setting of the range for link data communications for each station (master, local, and remote I/O stations). For details, refer to Chapter 7.
(c)	Monitoring time	The maximum allowable time that local stations and remote I/O stations take to determine whether the master station is operating normally. (For details, refer to Section 7.5.)

(4) Operation mode in the MELSECNET data link system Operation mode is determined according to the type of link module which is connected to the MELSECNET data link system and link parameter setting. For details, refer to Section 1.2.4.

2.2.3 Applicable modules

The following describes link modules which can be used in the MELSECNET.

(1) Data link module using optical fiber cable

The data link modules that can be connected with optical fiber cable are listed in Table 2.2.

O: Applicable

		Appli	cable										
		opt	ical					Two-tier	system				
Module	Model	fiber	cable	Description	MELS	ECNET	mode	MELSE	ECNET	MEI	SECNE	ТΠ	Remarks
		SI	GI		М		R	ш m м	lode	com M	posite m		-
	A0J2HCPUP21	01	01		IVI	-		IVI	-	IVI	-	IX.	
	A2CCPUP21	0											
	A1NCPUP21	1											
	A1NCPUP21-S3		0										
	A2NCPUP21	0		CDLL modulo with the link									
	A2NCPUP21-S3		0	function	0	0					0		
	A2NCPUP21-S1	0											Line the station
CPU module	A2NCPUP21-S4		0										number setting
	A3NCPUP21	0											switch to set the
	A3NCPUP21-S3		0										or local station.
	A2ACPUP21	0											
	A2ACPUP21-S3		0										
	A2ACPUP21-S1	0		CPU module with the link	0	0			0	0	0		
	A2ACPUP21-S4		0	function	0	0			0	0	0		
	A3ACPUP21	0											
	A3ACPUP21-S3		0										
	A1SJ71AP21	0		Module for data link, used with any of the following CPUs:	0	0		*1	*1	*1	0		Installed in an I/O slot in a base unit.
	A1SJ71AP21-S3		0	AnSCPU, AnUSCPU, QnASCPU, QCPU-A		Ū		0	0	0	Ŭ		
	AJ71AP21	0		Module for data link, used	0	0		*2	*2	*2	0		
	AJ71AP21-S3		0	the link function (ACPU)	0	0		0	0	0	0		
_	A0J2P25	0		Compact type module for									
Data link	A0J2P25-S3		0	remote I/O stations									
module	AJ72P25	0		Module for remote I/O			0					0	Installed in the CPU
	AJ72P25-S3		0	mounted on the following base units: A32B(-S1), A35B, A38B									slot of a main base unit.
	A1SJ71AP23Q	0		Module for data link, used with a QCPU module		0			0		0		 Used only for local stations. Installed in an I/O slot in an extension base unit.

REMARK

- (1) The M, L and R stations in Table 2.2 indicate the following stations.
 - a) M station..... Master station
 - b) L station..... Local station
 - c) R station..... Remote I/O station
- (2) *1: Applicable only when used in combination with the A2US(H)CPU(S1), Q2AS(H)CPU(S1) or QCPU-A.
- (3) *2: Applicable only when used in combination with the AnACPU, AnUCPU or QnACPU.

(2) Data link module using coaxial cable

The data link modules that can be connected with coaxial cable are listed in Table 2.3.

											O: Applicable	
Module	Model	Model Description		ELSECNET		MELSEC		MEL	SECNE	ΤI	Remarks	
			mode			NET II mode		composite m		node		
			М	L	R	М	L	М	L	R		
	A0J2HCPUR21											
CPU module	A2CCPUR21											
	A1NCPUR21											
	A2NCPUR21	CPU module with the link	0	0					0		Use the station number	
	A2CPUR21-S1										setting switch to set the	
	A2NCPUR21-S1	-									selection of master or	
	A3NCPUR21										local station.	
	A2ACPUR21											
	A2ACPUR21-S1	CPU module with the link	0	0		0	0	0	0			
	A3ACPUR21											
	A1SJ71AR21	Module for data link, used with any of the following CPUs: AnSCPU, AnUSCPU, QnASCPU, QCPU-A	0	0		*1 O	*1 O	*1 O	0		Installed in an I/O slot in	
	AJ71AR21	Module for data link, used with a CPU module without the link function (ACPU)	0	0		*2 O	*2 O	*2 O	0		a base unit.	
Data link	A0J2R25	Compact type module for remote I/O stations										
module	AJ72R25	Module for remote I/O stations, which is to be mounted on the following base units: A32B(-S1), A35B, A38B			0					0	Installed in the CPU slot of a main base unit.	
	A1SJ71AR23Q	Module for data link, used with a QCPU module		0			0		0		 Used only for local stations. Installed in an I/O slot in an extension base unit. 	

Table 2.3 Data Link Modules Connectable with Coaxial Cable

REMARK

- (1) The M, L and R stations in Table 2.3 indicate the following stations.
 - a) M station..... Master station
 - b) L station..... Local station
 - c) R station..... Remote I/O station
- (2) *1: Applicable only when used in combination with the A2US(H)CPU(S1), Q2AS(H)CPU(S1) or QCPU-A.
- (3) *2: Applicable only when used in combination with the AnACPU, AnUCPU or QnACPU.

MEMO

3 TWO-TIER SYSTEM IN THE MELSECNET/B DATA LINK SYSTEM

This section describes the MELSECNET/B data link system.

3.1 Outline of the MELSECNET/B Data Link System

3.1.1 Configuration of the data link system

The MELSECNET/B data link system is a system to connect link modules using shielded twisted pair cable.

Up to 31 slave stations (local station and remote I/O station) can be used in a system which has one link module as the master station.



Master station Local station

(1) Master station

Master station is the link module that controls the whole MELSECNET/B data link system.

The number of connected slave stations (Max. 31) and the device (B, W, X, Y) ranges for data communications are set with link parameters at the programmable controller CPU in the master station.

The master station controls data communications in a MELSECNET/B data link system in accordance with these set link parameters.

(2) Slave stations

Slave stations include a local station and a remote I/O station.

(a) Local station

When two or more programmable controller CPUs are used for data link, local stations are used to increase the number of I/O points and the program capacity in a large-scale system.

(b) Remote I/O station

Remote I/O stations are used to reduce wiring cost when data must be frequently input/output from/to devices that are far away from the programmable controller CPU.

The programmable controller CPU in the master station controls the input and output of remote I/O stations.

The number of I/O points is 512 points (X/Y0 to 1FF) per station.

REMARK

Master stations, local stations, and remote I/O stations are expressed in the following symbols.

- Master station..... M
- Local station..... L (Local station No. n: Ln)
- Remote I/O station...... R (remote I/O station No. n: Rn)

3.1.2 Features of the data link system

The features of the MELSECNET/B data link system are described below.

(1) Cyclic transmission function

The cyclic transmission is a function to periodically communicate data between a master station and slave stations (local stations and remote I/O stations) Either 1 : n or 1 : 1 data communications are enabled by the cyclic transmission function.

(a) 1 : n data communications

This is data communications between the master station and all local stations/ between local stations.

ON/OFF data and 16-bit data can both be communicated:

- 1) ON/OFF data is communicated using the link relay (B).
- 2) 16-bit data is communicated using the link register (W).



Fig 3.1 Flow of data by B/W communication

(b) 1:1 data communications

This is 1 : 1 data communications between a master station and a local station. ON/OFF data can be communicated using inputs (X) and outputs (Y).



Fig 3.2 Flow of input (X) and output (Y) communication data

(2) Transient transmission function

The transient transmission function (a) reads/writes data from/to a device in a local station by using a master station programmable controller CPU, and (b) communicates data between a peripheral device connected to a programmable controller CPU and a programmable controller CPU in another station. The transient transmission function executes the following types of processings:

(a) Communications between a master station and a local station Read/write for devices (T, C, D and W) in a local station by a master station programmable controller CPU.

LRDP/LWTP^{*1} instructions in a sequence program are used for this processing at the master station.

(b) Communications between a master station and a remote I/O station Read/write of data from/to the buffer memory of a special-function module connected to a remote I/O station is performed from a master station programmable controller CPU.

RFRP/RTOP^{*1} instructions in a sequence program are used at the master station.

(c) Access between a peripheral device or special-function module and another station

Another station is accessed from a peripheral device or special-function module connected to a programmable controller CPU.

As Table 3.1 shows, the accessible station varies depending on the peripheral device or special-function module installed in a master, local or remote I/O station.

However, the stations in Table 3.1 are basically accessible.

The manual of the used peripheral device or special-function module gives details about the functions that can be executed by that peripheral device or special-function module.

Access target station	Station where a peripheral device is connected								
	Master Station	Local Station	Remote I/O Station						
Master Station	Accessible	Accessible	Accessible						
Local Station	Accessible	Not accessible	Not accessible						
Remote I/O Station	Accessible*2	Not accessible	Not accessible						

T-1-1-040	Alexandra Marcalla I. I. A. Devila		
Table 3.1 Communicating Sta	ations Available to Peripi	neral Devices and Specia	II-Function Modules

REMARK

*1	 When the AnACPU(P21/R21), AnUCPU, A2US(H)CPU(S1) or QCPU-A is
	used, the LRDP/LWTP/RFRP/RTOP instructions of the dedicated instructions
	can also be used.
	(For details of the instructions, refer to the Type AnSHCPU/AnACPU/
	AnUCPU/QCPU-A (A Mode) Programming Manual (Dedicated Instructions)
	IB-66251.)

- When the QnACPU, Q2AS(H)CPU(S1) is used, the ZNRD/ZNWR/RFRP/ RTOP instructions of the data link instructions can also be used. (For details on the instructions, refer to the QnACPU Programming Manual (Common Instructions).)
- *2 Not accessible when using GX Developer.
- (3) Improved RAS (Reliability, Availability, Serviceability) functions
 - (a) Automatic return function When a slave station (local station) where an error occurred returns to the linkenabled state, the station automatically restarts the data link operation. (Refer to Section 5.3.3.)
 - (b) Error detection
 - The data link operating state is stored in special relays (M) and special registers (D) in a programmable controller CPU. The data link operating state can be checked by reading the data stored in the special relays (M) and special registers (D).
 - 2) The data link operating state can also be checked by using the link monitor function at a peripheral device (A6GPP, A6PHP, or A6HGP).
 - (c) Self-diagnostics function

The self-diagnostics function of the data link module checks the link module hardware and shielded twisted pair cable connections.

- (4) Three operating modes can be selected by setting the CPU module To satisfy various system configuration requirements, different operating modes can be selected for the MELSECNET/B data link system by setting link parameters. The MELSECNET/B data link system has the following three operation modes: MELSECNET mode, MELSECNET II mode, and MELSECNET II composite mode. Section 1.2 gives details about the differences among these modes.
- (5) Switching the communication speed is enabled. The communication speed can be set to 125kbps, 250kbps, 500kbps, or 1Mbps. The total link distance can be changed by switching the communication speed. For the relationship between the communication speed and total link distance, refer to Table 1.1.

3.2 MELSECNET/B Data Link System

3.2.1 Overall configuration

(1) Two-tier system

The two-tier system is a system to connect up to 31 local stations and remote I/O stations to a master station via shielded twisted pair cables. In a two-tier system, a master station is referred to as "master station" and a local station or a remote I/O station is referred to as "slave station".

(2) System configuration

Fig 3.3 shows the configuration of the two-tier system.



Fig 3.3 Two-tier system

POINT	
(1) Remote l/ used.	O stations cannot be connected if the MELSECNET I mode is
(2) The conn remote I/(system.	ection order or station order of master stations, local stations and O stations can be freely decided in the MELSECNET/B data link

3.2.2 Precautions when operating the data link system

The following describes the precautions for performing data link.

- Number of link device points for one station Note that the number of link device (X, Y, B, W) points that can be used at a local station and a remote I/O station is limited. (For details, refer to Section 7.3.)
- (2) Link parameter setting Set the link parameters in the master station to perform data link. The link parameters include the number of slave stations connected in the MELSECNET/B data link system, the link device assignment, and the watchdog monitoring time.

(a)	Number of slave stations	The total number of the local stations and remote I/O stations connected to the MELSECNET/B data link system.
(b)	Link device assignment	The range of link data communications is set at each master station, local station, and remote I/O station. For details, refer to Chapter 7.
(c)	Monitoring time	It is used for local stations and remote I/O stations to judge whether a master station is operating normally. For details, refer to Section 7.5.

- (3) The operating mode of a MELSECNET/B data link system The operating mode is determined by the type of link modules connected to the MELSECNET/B data link system and the link parameter settings. For details, refer to Section 1.2.4.
- (4) Shielded twisted pair cable
 The shielded twisted pair cable used for the MELSECNET/B data link system is
 KNPEV-SB0.5SQ×1P.
 For details, refer to Section 5.6.

3.2.3 System devices

Table 3.2 shows the link modules that can be used in the MELSECNET/B Data Link System.

												0
						Ap	plicab	le syst	em			
		Description		Two-tier system								
Link module	CPU module			MELSECNET			MELSEC		MELSECNET		NET	Remarks
				mode		NE mo	NET II mode		II composite			
		Program capacity	Number of I/O points	М	L	R	M	L	М	L	R	
	A0J2HCPU	8k steps	336 points									
	A1NCPU	6k steps	256 points									
	A2NCPU	14k steps	512 points	0	0					0		
	A2NCPU-S1	14k steps	1024 points									
	A3NCPU	30k steps	2048 points									
	A2ACPU	14k steps	512 points							0		
	A2ACPU-S1	14k steps	1024 points									
AJ71AT21B	A3ACPU	30k steps	2048 points		0							
	A2UCPU	14k steps	512 points						0			
	A2UCPU-S1	14k steps	1024 points									
	A3UCPU	30k steps	2048 points	0			0	0				
	A4UCPU	30k steps	4096 points									
	Q2ACPU	28k steps	512 points									
	Q2ACPU-S1	60k steps	1024 points									
	Q3ACPU	92k steps	2048 points									Use the station
	Q4ACPU	128k steps	4096 points									number setting switch
		8k steps	256 points	0						0		master or local station.
	A2SHCPU	14k steps	512 points	0	0					0		
	A2ASCPU	14k steps	512 points									
	A2ASCPU-S1	14k steps	1024 points						0	0		
	A2USHCPU-S1	30k steps	1024 points					0				
A1SJ71	Q2ASCPU	28k steps	512 points	0	0		0					
AT21B	Q2ASCPU-S1	60k steps	1024 points									
	Q2ASHCPU	28k steps	512 points									
	Q2ASHCPU-S1	60k steps	1024 points									
	Q02CPU-A	30k steps										
	Q02HCPU-A	30k steps	4096 points	0	0		0	0	0	0		
	Q06HCPU-A	60k steps										
	Q02CPU	28k steps										
	Q02HCPU	28k steps										Used only for local stations
A1SJ71AT23QB	Q06HCPU	60k steps	4096 points		0			0		0		Installed in an I/O slot
	Q12HCPU	124k steps										in an extension base unit.
	Q25HCPU	252k steps										

Table 3.2 List of link modules

O: Available

3. TWO-TIER SYSTEM IN THE MELSECNET/B DATA LINK SYSTEM

MELSEC-A

Table 3.2 List of link modules (Continued)

O: Available

						Ap	plicab	le syst	em			
		Description			Two-tier system							
Link module	CPU module				LSECI mode	NET MELSEC NETI mode		MELSECNET II composite mode			Remarks	
		Program capacity	Number of I/O points	М	L	R	М	L	М	L	R	
AJ72T25B	-	Module for remote I/O stations, which is to be mounted to the following base unit: A32B(-S1), A35B, A38B				0					0	Installed in the CPU slot
A1SJ72T25B	-	Module for remote I/O stations, which is to be mounted to the following base unit: A1S32B, A1S33B, A1S35B, A1S38B)	of a main base unit.

REMARK

- (1) The M, L and R stations in Table 3.2 indicate the following stations.
 - a) M station..... Master station
 - b) L station.....Local station
 - c) R station......Remote I/O station

4 COMPOSITION OF A THREE-TIER SYSTEM

In a three-tier system, a local station in the second tier is used as a master station for the third tier and connects with the slave stations.

MELSECNET data link system and MELSECNET/B data link system can be used for the second and third tiers. The combinations are shown in Fig 4.1.



POINT	
(1) When the	AnUCPU, QnACPU, A2ASCPU(S1), A2USHCPU-S1,
Q2AS(H)	CPU(S1), or QCPU-A is used, the three-tier system can be
configured	d with the MELSECNET/B data link system.
When oth	er CPU modules are used, the three-tier system cannot be
configured	d only with the MELSECNET/B data link system.
(2) Local and	remote I/O stations in the third tier are referred to sub-slave stations
and are co	ontrolled by the master station in the third tier.

4.1 Three-Tier System Using the MELSECNET Data Link System

When the second and third tiers are configured with MELSECNET data link system, connect the stations using optical fiber cables or coaxial cables.

- (a) Up to 64 local and remote I/O stations can be connected to a master station for the second tier.
- (b) For the third tier, up to 64 local and remote I/O stations can be connected to the master station, which is the local station for the second tier.

4.1.1 System configuration



Fig 4.2 shows the configuration of the three-tier system.

Fig 4.2 Three-tier system configured with MELSECNET data link system

In the three-tier system, optical fiber cables and coaxial cables can be used together for the second and third tiers.

Combinations of the optical fiber cables and coaxial cables for each tier are shown below.



POINT

- (1) In MELSECNET data link system, up to three tiers can be configured.
- (2) Remote I/O stations cannot be connected if the MELSECNET II mode is used.

4.1.2 Precautions when operating data link system

This section explains precautions for three-tier system using MELSECNET data link.

(1) Setting link parameters

In the three-tier system, setting link parameters to CPU modules for the master stations in the second and third tiers is required.

For link parameter setting, refer to Section 5.3.7 and Chapter 7.



- (2) Restriction on use of link modules
 - (a) For the CPU module other than the AnUCPU, A2ASCPU(S1), A2USHCPU-S1, QnACPU, Q2AS(H)CPU(S1), and QCPU-A

Only one of the following link modules can be used with a CPU module. When configuring the three-tier system, use a CPU module having a link function with any of the following link modules.

- AJ71AP21
- AJ71AP21-S3
- A1SJ71AP21
- AJ71AR21
- A1SJ71AR21



(b) For the CPU module any of the AnUCPU, A2ASCPU(S1), A2US(H)CPU(S1), QnACPU, Q2AS(H)CPU(S1), and QCPU-A Up to two of the following link modules (one as a master station and the other as a local station) can be used with a CPU module.

(The two modules cannot be used only for master stations or local stations.)

- AJ71AP21
- AJ71AP21-S3
- A1SJ71AP21
 AJ71AR21

- A1SJ71AR21



(3) When using a CPU module with link function as a master station for the third tier In a three-tier system including a CPU module with link function and a data link module, the CPU module can be used as a master station for the third tier and the data link module as a local station in the second tier.



Note that the following AnACPUP21/R21 versions and later can be used as a master station for the third tier.

When configuring a three-tier system with a version earlier than that, use the AnACPUP21/R2 as a local station in the second tier, and a data link module as a master station for the third tier.

CPU module with link function	Version
A2ACPUP21	107C <u>X</u>
A2ACPUR21	107B <u>X</u>
A2ACPUP21-S1	107B <u>Y</u>
A2ACPUR21-S1	107B <u>Y</u>
A3ACPUP21	107C <u>Z</u>
A3ACPUR21	107B <u>Z</u>

4.1.3 System devices

													-					O: Available
								Арр	licabl	le sys	tem							
								MELS	ECN	ET da	ta link	ĸ						
					Seco	nd tier							Thire	d tier				
Module	Model				MEL	SEC	MEL	SEC	NET				MEL	SEC	MEL	SEC	NET	Remarks
		MEL	_SECI	NET	NE	ТΠ	Пc	ompo	site	MEL	_SEC	NET	NE	ТΠ	Пc	ompo	site	
			moue		mo	de		mode			moue		mo	de		mode	•	
		М	L	R	М	L	М	L	R	L/m	l	r	L/m	l	L/m	l	r	
	A0J2HCPUP21																	These connet he wood
	A0J2HCPUR21	0	0					0			~					0		I nese cannot be used
	A2CCPUP21	0	0					0			0					0		the third tier.
	A2CCPUR21																	
	A1NCPUP21																	
	A1NCPUP21-S3																	
	A1NCPUR21																	
	A2NCPUP21																	
	A2NCPUP21-S3																	
	A2NCPUR21		-					~		~	~					~		
	A2NCPUP21-S1	0	0					0		0	0					0		
CPU	A2NCPUP21-S4																	
module	A2NCPUR21-S1																	
function	A3NCPUP21																	Use the station number
	A3NCPUP21-S3																	setting switch to set the
	A3NCPUR21																	local station.
	A2ACPUP21																	
	A2ACPUP21-S3																	
	A2ACPUR21																	
	A2ACPUP21-S1																	
	A2ACPUP21-S4	0	0		0	0	0	0		0	0		0	0	0	0		
	A2ACPUR21-S1																	
	A3ACPUP21																	
	A3ACPUP21-S3	1																
	A3ACPUR21																	

Table 4.1 Link modules available for the three-tier system

REMARK

- (1) The definitions of L/m station, ℓ station, and r station in Table 4.1 are as follows.
 - a) L/m station Local station in the second tier/master station in the third tier
 - b) ℓ station Local station in the third tier
 - c) r station Remote I/O station in the third tier



MELSEC-A

Table 4.1 Link modules available for the three-tier system (Continued)

O: Available

									Ар	olicab	le sys	tem							
									MELS	SECN	ET da	ıta linl	K						
						Seco	nd tie	r						Thir	d tier				
Module	Мо	del	MEI	_SEC mode	NET 9	MEL NE mo	SEC TII ode	MEI II c	LSEC compo mode	NET osite	MEI	LSEC mode	NET 9	MEL NE mo	SEC TII ode	MEI II c	LSEC compo mode	NET site	Remarks
			М	L	R	М	L	М	L	R	L/m	l	r	L/m	l	L/m	l	r	1
	A1SJHCPU																		
	A1SHCPU		0	0					0			0					0		
	A2SHCPU	A19 171 AD21																	
	A2ASCPU	/R21																	
	A2ASCPU-S1		0	0		0					0				0				
	A2USHCPU-		0	0		0	0	0	0		0			0	0	0			
	S1																		
	A0J2HCPU																		
	A1NCPU																		
	A2NCPU		0	0					0			0					0		
	A2NCPU-S1																		
	A3NCPU																		
	A2ACPU	AJ71AP21/																	
	A2ACPU-S1	AJ71AP21-S3	0	0		0	0	0	0			0			0		0		
	A3ACPU																		Use the station number
	A2UCPU																		setting switch to set the
CDU	A2UCPU-S1		-				_	_	_			_		_	-	_	_		selection of master or
module	A3UCPU		0	0		0	0	0	0		0	0		0	0	0	0		iocal station.
+	A4UCPU																		
link module	Q2ASCPU																		-
modulo	Q2ASCPU-																		
	S1	A1SJ71AP21	0	0		0	0	0	0		0			0	0				
	Q2ASHCPU	/R21	U	U		U					U				U				
	Q2ASHCPU- S1																		
	Q2ACPU																		
	Q2ACPU-S1	AJ71AP21/	0																
	Q3ACPU	AJ71AP21-S3	0	0		0	0	0	0		0	0		0	0	0	0		
	Q4ACPU																		
	Q02CPU-A																		-
	Q02HCPU-A	A1SJ71AP21/	0	0		0	0	0	0		0	0		0	0	0	0		
	Q06HCPU-A	1121																	
	Q02CPU																		
	Q02HCPU																		Used only for local stations
	Q06HCPU	A1SJ71AP23		0			0		0			0			0		0		Installed in an I/O slot
	Q12HCPU	U/K23U																	in an extension base
	Q25HCPU																		unit.

O: Available

Table 4.1 Link modules available for the three-tier system (Continued)

								Арр	olicab	le sys	tem							
								MELS	SECN	ET da	ta link	(
					Seco	nd tier	-						Thire	d tier				
Module	Model	MEI	_SEC	NET	MEL NE	SEC T II	MEI II c	_SEC	NET site	MEL	_SECI mode	NET	MEL NE	SEC	MEI IC	LSEC	NET site	Remarks
					mo	ode		mode		1 (110		_	mo	bde	1 /	mode		
		IVI	L	ĸ	IVI	L	IVI	L	ĸ	L/m	l	ſ	L/m	l	L/m	l	I	
	A1NCPUP21-																	
	S3																	
	A2NCPUP21-																	
CPU module with link	A2NCPUR21																	Use the station number
	A2NCPUP21- R21 S1									*1 O								setting switch to set the selection of master or
	A2NCPUP21-									_								local station.
	A2NCPUR21-																	
tunction +	S1 A3NCPUP21																	
module	A3NCPUP21-																	
	A3NCPUR21																	
	A2ACPUP21																	
	A2ACPUR21																	
	A2ACPUP21- AJ71AP21/ S1 R21									*2			*2		*2			
	A2ACPUR21- S1 A21/1AP21-S3									0			0		0			
	A3ACPUP21																	
	A3ACPUR21																	
	A0J2P25																	
	A0J2P25-S3																	
Data link	A0J2R25]																
module	AJ72P25			0					0			0						
	AJ72P25-S3																	
	AJ72R25																	

*1: MELSECNET and MELSECNET ${\rm I\!I}$ composite modes can be used for the second tier.

*2: MELSECNET, MELSECNET II, and MELSECNET II composite modes can be used for the second tier.

O: Available

Table 4.1 Link modules available for the three-tier system (Continued)

Applicable system MELSECNET data link Second tier Third tier Module Remarks Model MELSEC MELSECNET MELSEC MELSECNET MELSECNET MELSECNET NET II II composite NET II I composite mode mode mode mode mode mode Μ L R Μ L Μ L R L/m L/m L/m r l r l l A2ASCPU +A1SJ71AP21 A2ASCPU +A1SJ71AR21 A2ASCPU-S1 +A1SJ71AP21 A2ASCPU-S1 +A1SJ71AR21 A2USHCPU-S1 +A1SJ71AP21 A2USHCPU-S1 A1SJ71AP2 +A1SJ71AR21 A1SJ71AR2 Q02CPU-A +A1SJ71AP21 Q02CPU-A +A1SJ71AR21 Q02HCPU-A +A1SJ71AP21 CPU Use the station Q02HCPU-A module number setting +A1SJ71AR21 *2 *2 *2 4 switch to set the link 0 0 Ο Q06HCPU-A selection of master modules +A1SJ71AP21 or local station. (two) Q06HCPU-A +A1SJ71AR21 A2UCPU +AJ71AP21(S3) A2UCPU +AJ71AR21 A2UCPU-S1 +AJ71AP21(S3) A2UCPU-S1 AJ71AP21 +AJ71AR21 AJ71AP21-A3UCPU **S**3 AJ71AR21 +AJ71AP21(S3) A3UCPU +AJ71AR21 A4UCPU +AJ71AP21(S3) A4UCPU +AJ71AR21

*2: MELSECNET, MELSECNET II or MELSECNET II composite modes can be used for the second tier.

4.2 Data Link System when the Second Tier is MELSECNET and the Third Tier is MELSECNET/B

In the system that the second tier is a MELSECNET data link system and the third tier is a MELSECNET/B data link system, the second tier is connected with optical fiber cables or coaxial cables, and the third tier is connected with shielded twisted pair cables.

- (a) Up to 64 local and remote I/O stations can be connected to a master station for the second tier.
- (b) For the third tier, up to 31 local and remote I/O stations can be connected to the master station, which is the local station for the second tier.

4.2.1 System configuration

Fig 4.3 shows the configuration of the three-tier system.



Fig 4.3 Three-tier system when the second tier is MELSECNET data link system

POINT

- (1) Remote I/O stations cannot be connected if the MELSECNET II mode is used.
- (2) MELSECNET/B data link system has no restriction on the order of stations (including master stations).

4.2.2 Precautions when using data link

This section explains precautions for configuring a three-tier system.

(1) Setting link parameters

In the three-tier system, setting link parameters to CPU modules for the master stations in the second and third tiers is required.

For link parameter setting, refer to Section 5.3.7 and Chapter 7.



- (2) Restriction on use of link modules
 - (a) For the CPU module other than the AnUCPU, A2ASCPU(S1), A2USHCPU-S1, QnACPU, Q2AS(H)CPU(S1), and QCPU-A

Only one of the following link modules can be used with a CPU module. When configuring the three-tier system, use a CPU module having a link function with any of the following link modules.

- AJ71AP21 Second tier Second tier • AJ71AP21-S3 A А 3 - A1SJ71AP21 Ĺ NC PU - AJ71AR21 A Α - A1SJ71AR21 Iр π - AJ71AT21B B - A1SJ71AT21B Third tier Third tier
- (b) For the CPU module any of the AnUCPU, A2ASCPU(S1), A2USHCPU-S1, QnACPU, Q2AS(H)CPU(S1), and QCPU-A

Up to two of the following link modules (one as a master station and the other as a local station) can be used with a CPU module.

(The two modules cannot be used only for master stations or local stations.)

[For local station]

- AJ71AP21
- AJ71AP21-S3
- A1SJ71AP21
- AJ71AR21
- A1SJ71AR21

[For master station]

- AJ71AT21B
- A1SJ71AT21B



4.2.3 System devices

																		O: Available
								Арр	olicab	le sys	tem							
				MELS	ECN	ET da	ta link	(N	1ELSE	ECNE	T/B d	ata lin	ık		
					Secor	nd tier							Third	d tier				
Module	Model		050		MEL	SEC	MEI	SEC	NET		050		MEL	SEC	MEL	SEC	NET	Remarks
		WEL	-SECI		NE	ТⅡ	ПΟ	ompo	site	IVIEI	_SEU	NEI	NE	ГΠ	пс	ompo	site	
			moue		mc	ode		mode			mode		mo	de		mode	•	
		М	L	R	М	L	М	L	R	L/m	l	r	L/m	l	L/m	l	r	
	A0J2HCPUP21																	
	A0J2HCPUR21	~						~										These cannot be used
	A2CCPUP21	0	0					0										the third tier
	A2CCPUR21																	
	A1NCPUP21																	
	A1NCPUP21-S3																	
	A1NCPUR21																	
	A2NCPUP21																	
	A2NCPUP21-S3																	
	A2NCPUR21		~					~										
	A2NCPUP21-S1	0	0					0										
CPU	A2NCPUP21-S4																	
module	A2NCPUR21-S1																	
function	A3NCPUP21																	Use the station number
lanouon	A3NCPUP21-S3																	setting switch to set the
	A3NCPUR21																	local station
	A2ACPUP21																	
	A2ACPUP21-S3																	
	A2ACPUR21																	
	A2ACPUP21-S1																	
	A2ACPUP21-S4	0	0		0	0	0	0										
	A2ACPUR21-S1																	
	A3ACPUP21																	
	A3ACPUP21-S3	1																
	A3ACPUR21																	
			_															
		11/1/																

Table 4.2 Link modules available for the three-tier system

(1) The definitions of L/m station, ℓ station, and r station in Table 4.2 are as follows.

- a) L/m station Local station in the second tier/master station in the third tier
- b) ℓ station Local station in the third tier
- c) r station Remote I/O station in the third tier



Table 4.2 Link modules available for the three-tier system (Continued)

														•			,		O: Available
									Ар	olicab	e sys	tem							
					MELS	SECN	ET da	ta link	(Ν	/IELS	ECNE	T/B d	ata lir	ık		
						Seco	nd tier	r						Thir	d tier	1			-
Module	Mo	odel	ME	LSEC	NET	MEL	SEC	MEI	SEC	NET	MEL	SEC	NET	MEL	SEC	MEI	LSEC	NET	Remarks
				mode	9	NE mo	I∐ ode	шс	ompc mode	site		mode	9	NE mo	I∐ ode	шс	compc mode	osite	
			м	1	R	м	1	м	1	R	l /m	0	r	l /m	0	l /m	0	r	
	A0J2HCPU			-			-		-		2/111	ĸ	·	2	ĸ	2,	Ł	·	
	A1NCPU																		
	A2NCPU	AJ71AP21/ R21	0	0					0										
	A2NCPU-S1	AJ71AP21-S3							-										
	A3NCPU	-																	
	A2ACPU																		-
	A2ACPU-S1	-																	Lise the station number
	A3ACPU	-																	setting switch to set the
	A2UCPU																		selection of master or
	A2UCPU-S1	AJ71AP21/																	local station.
	A3UCPU	R21	0	0		0	0	0	0										
	A4UCPU	AJ71AP21-S3																	
CPU	Q2ACPU																		
module +	Q2ACPU-S1																		
link	Q3ACPU																		
module	Q4ACPU																		
	Q02CPU																		• Used only for local
	Q02HCPU	4 4 0 174 4 500																	stations.
	Q06HCPU	Q/R23Q		0			0		0										Installed in an I/O slot
	Q12HCPU																		in an extension base
	Q25HCPU																		unit.
	A1SJHCPU																		
	A1SHCPU	A1SJ71AT21 B																	
	A2SHCPU	1=																	Use the station number
	A1NCPU		1									0					0		setting switch to set the selection of master or
	A2NCPU																		local station.
	A2NCPU-S1	AJ/TATZTB																	
	A3NCPU	-																	

O: Available

Table 4.2 Link modules available for the three-tier system (Continued)

									Арр	olicab	le sys	tem							
Module					MELS	ECN	ET da	ta link	(N	/ELSI	ECNE	T/B d	lata lin	ık		
						Secor	nd tier							Thire	d tier				-
Module	Mo	odel	MEI	LSEC	NET	MEL	.SEC τπ	MEL	SEC	NET	MEL	SEC	NET	MEL	SEC	MEL	SEC	NET	Remarks
				mode	;	m	л ш ode	шс	mode	e		mode	•	mc	и de	шс	mode	e	
			М	L	R	М	L	М	L	R	L/m	D	r	L/m	D	L/m	Q	r	-
	A2ASCPU											v			÷		v		
	A2ASCPU-S1																		
	A2USHCPU-																		
	S1																		
	Q2ASCPU	A1S.I71AT21B																	
	Q2ASCPU-																		
	S1																		
	Q2ASHCPU																		
	Q2ASHCPU-																		
	S1		-																
CPU module	A2ACPU	-										0			0		0		
	A2ACPU-S1	-																	setting switch to set the
	A3ACPU	4																	selection of master or
	A2UCPU																		local station.
module +	A2UCPU-S1																		
link	A3UCPU	AJ71AT21B																	
module	A4UCPU																		
	Q2ACPU																		
	Q2ACPU-S1																		
	Q3ACPU																		
	Q4ACPU																		
	Q02CPU-A																		
	Q02HCPU-A	A1SJ71AT21B										0			0		0		
	Q06HCPU-A																		
	Q02CPU																		
	Q02HCPU																		Used only for local stations
	Q06HCPU	A1SJ71AT23B										0			0		0		Installed in an I/O
	Q12HCPU	Q																	slot in an extension
	Q25HCPU																		base unit.
Link	A0J2P25															<u> </u>			
	A0J2P25-S3		•																
	A0J2R25		•		0					0									
	AJ72P25									Ũ									
module	AJ72R25																		
	A1SJ72T25B				<u> </u>	<u> </u>		<u> </u>									<u> </u>		
	AJ72T25B		•										0					0	

Table 4.2 Link modules	available for the thr	ee-tier system (Continued)

O: Available

									Арр	olicab	e sys	tem							 -
					MELS	SECN	ET da	ta link	(N	1ELSI	ECNE	T/B d	ata lin	k		-
						Secor	nd tier							Third	d tier				
Module	Mod	el	MEL	SEC	NET	MEL	SEC	MEI	SEC	NET	MEL	SEC	NET	MEL	SEC	MEL	SEC	NET	Remarks
				mode		NE mc	I∐ nde	шс	ompo mode	site		mode		NE mc	I∐ nde	шс	ompc mode	site	
						M					1 / 100			1. (1 / 100	inioae		
			IVI	L	ĸ	IVI	L	IVI	L	к	L/m	l	ſ	L/m	l	L/m	l	ſ	
	A1NCPUP21																		
	A1NCPUP21-																		
	AZNCPUPZI																		
	S3																		
	A2NCPUR21																		
											*1								
	S1	AJ71AT21B									0								
CPU	A2NCPUP21-																		
module	S4																		
with link function + link	A2NCPUR21																		
	-S1																		
	A3NCPUP21																		
module	A3NCPUP21-																		
	S3																		
	A3NCPUR21																		Use the station number
	A2ACPUP21																		setting switch to set the
	A2ACPUR21																		selection of master or
	A2ACPUP21-																		local station.
	S1 /	AJ71AT21B									*2			*2		*2			
	A2ACPUR21-										0			0		0			
	A3ACPUR21																		
	A2ASCPU																		
	1																		
	A2ASCPU																		
CPU	+A1SJ71AR2																		
module	1																		
module + link modules	A2ASCPU-	A1SJ71AT21									*2			*2*3		*2*3			
	S1	5									0			0		0			
(two)	1																		
	A2ASCPU-																		
	S1																		
	+A1SJ71AR2																		
	1																		

*1: MELSECNET and MELSECNET ${\rm I\!I}$ composite modes can be used for the second tier.

*2: MELSECNET, MELSECNET ${\rm I\!I}$, and MELSECNET ${\rm I\!I}$ composite modes can be used for the second tier.

*3: Applicable when the A1SJ71AT21B is used.

												-		•		,			O: Available
									Арр	olicabl	e sys	tem							
					MELS	ECN	ET da	ta link				Ν	IELS	ECNE	T/B d	ata lin	ık		-
						Secor	nd tier							Thire	d tier				-
Module	Мос	del	MEL	SECI	NET	MEL	SEC	MEL	SEC	NET	MEL	SEC	NET	MEL	SEC	MEL	SEC	NET	Remarks
				mode	1	NE	T II de	Πc	ompo mode	site		mode	•	NE	T II de	Πc	ompc mode	site	
				r .	_					_							Inoue	, 	-
			М	L	R	М	L	М	L	R	L/m	l	r	L/m	l	L/m	l	r	
	A2USHCPU-																		
	S1																		
	A205HCP0-																		
	+A1SJ71AR21																		
	Q2ASCPU																		
	+A1SJ71AP21																		
	Q2ASCPU																		
	+A1SJ71AR21																		
	Q2ACPU-S1																		
	+A1SJ71AP21																		
	Q2ACPU-S1																		
	+A1SJ71AR21																		
	Q2ASHCPU																		
CPU	+A1SJ71AP21																		
module	Q2ASHCPU																		Use the station
+	+A1SJ71AR21	A1SJ71AT21									*2			*2*3		*2*3			number setting switch
link modules	Q2ASHCPU-	в									0			0		0			to set the selection of master or local station
(two)	51 +A1SJ71AP21																		
. ,																			
	S1																		
	+A1SJ71AR21																		
	Q02CPU-A																		
	+A1SJ71AP21																		
	Q02CPU-A																		
	+A1SJ71AR21																		
	Q02HCPU-A																		
	+A1SJ71AP21																		
	Q02HCPU-A																		
	+A1SJ71AR21																		
	Q06HCPU-A																		
	+A1SJ71AP21																		
	Q06HCPU-A																		
	+A15J/1AR21																		

Table 4.2 Link modules available for the three-tier system (Continued)

*2: MELSECNET, MELSECNET II, and MELSECNET II composite modes can be used for the second tier.

*3: Applicable when the A1SJ71AT21B is used.

O: Available

*1: MELSECNET and MELSECNET ${\rm I\!I}$ composite modes can be used for the second tier.

4.3 Data Link System when the Second Tier is MELSECNET/B and the Third Tier is MELSECNET

In the system that the second tier is a MELSECNET/B data link system and the third tier is a MELSECNET data link system, the second tier is connected with shielded twisted pair cables, and the third tier is connected with optical fiber cables or coaxial cables.

- (a) Up to 31 local and remote I/O stations can be connected to a master station for the second tier.
- (b) For the third tier, up to 64 local and remote I/O stations can be connected to the master station, which is the local station for the second tier.

4.3.1 System configuration





Fig 4.4 Three-tier system when the second tier is MELSECNET/B data link system

POINT

- (1) Remote I/O stations cannot be connected in MELSECNET I mode.
- (2) MELSECNET/B data link system has no restriction on the order of stations (including master stations).

4.3.2 Precautions when operating the data link system

This section explains precautions for configuring a three-tier system.

(1) Setting link parameters

In the three-tier system, setting link parameters to CPU modules for the master stations in the second and third tiers is required.

For link parameter setting, refer to Section 5.3.7 and Chapter 7.



- (2) Restriction on use of link modules
 - (a) For the CPU module other than the AnUCPU, A2ASCPU(S1), A2USHCPU-S1, QnACPU, Q2AS(H)CPU(S1), and QCPU-A

Only one of the following link modules can be used with a CPU module.

When configuring the three-tier system, use a CPU module having a link function with any of the following link modules.



 (b) For the CPU module any of the AnUCPU, A2ASCPU(S1), A2USHCPU-S1, QnACPU, Q2AS(H)CPU(S1), and QCPU-A
 Up to two of the following link modules (one as a master station and the other as a local station) can be used with a CPU module.

(The two modules cannot be used only for master stations or local stations.) [For local station]

- AJ71AP21
- AJ71AP21-S3
- A1SJ71AP21
- AJ71AR21
- A1SJ71AR21

[For master station]

- AJ71AT21B
- A1SJ71AT21B



(3) When using a CPU module with link function as a master station for the third tier When a CPU module with link function is used with the AJ71AT21B in a three-tier system, the CPU module can be used as a master station for the third tier and the AJ71AT21B can be used as a local station in the second tier.



Note that the following AnACPUP21/R21 versions and later can be used as a master station for the third tier.

When a version earlier than that is used, construction of a three-tier system where the AJ71AT21B is in the second tier and the CPU module is in the third tier is not possible.

CPU module with link function	Version						
A2ACPUP21	107C <u>X</u>						
A2ACPUR21	107B <u>X</u>						
A2ACPUP21-S1	107B <u>Y</u>						
A2ACPUR21-S1	107B <u>Y</u>						
A3ACPUP21	107C <u>Z</u>						
A3ACPUR21	107B <u>Z</u>						

4.3.3 System devices

													•					O: Available
	Model	Applicable system																
Module		MELSECNET/B data link MELSECNET data link																
		Second tier Third tier																
		MELSECNET mode		MELSEC NET II mode		MELSECNET I composite mode		MELSECNET mode		MELSEC NET II mode		MELSECNET II composite mode		NET	Remarks			
														site				
		М	L	R	М	L	М	L	R	L/m	l	r	L/m	l	L/m	l	r	
	A0J2HCPUP21	1																T he second second second
	A0J2HCPUR21										~					~		I hese cannot be used
	A2CCPUP21	1									0					0		the third tier.
	A2CCPUR21																	
	A1NCPUP21																	
	A1NCPUP21-S3																	
	A1NCPUR21																	
	A2NCPUP21																	
	A2NCPUP21-S3																	
	A2NCPUR21										~					~		
	A2NCPUP21-S1									0	0					0		
CPU	A2NCPUP21-S4																	
module	A2NCPUR21-S1																	Lise the station
function	A3NCPUP21																	number setting switch
	A3NCPUP21-S3																	to set the selection of
	A3NCPUR21	1																master or local station.
	A2ACPUP21																	
	A2ACPUP21-S3																	
	A2ACPUR21																	
	A2ACPUP21-S1																	
	A2ACPUP21-S4									0	0		0	0	0	0		
	A2ACPUR21-S1																	
	A3ACPUP21	1																
	A3ACPUP21-S3																	
	A3ACPUR21	1																

Table 4.3 Link modules available for the three-tier system

REMARK

(1) The definitions of L/m station, ℓ station, and r station in Table 4.3 are as follows.

a) L/m station Local station in the second tier/master station in the third tier

- b) ℓ station Local station in the third tier
- c) r station Remote I/O station in the third tier

MELSEC-A
Г

Table 4.3 Link modules available for the three-tier system (Continued	I)
---	----

O: Available

-

									Арр	olicab	le sys	tem							
				Ν	IELSI	ECNE	T/B d	ata lir	ık				MELS	SECN	ET da	ita link	(
						Seco	nd tier	r						Thire	d tier	1			
Module	Мос	lel	MEI	LSEC	NET	MEL	SEC	MEI	SEC	NET	MEL	SEC	NET	MEL	SEC	MEI	SEC	NET	Remarks
				mode	9	NE	I∐ Ide	ШC	compo mode	site		mode	•	NE	I∐ Ide	Шc	ompo mode	osite	
			NA		Б	N		M	1110000	, 	1 /m	0	_	1./m		1 /m	Â	, _	-
		1	IVI	L	ĸ	IVI	L	IVI	L	ĸ	L/III	l	1	L/III	l	L/III	l	'	
		AJ71AP21/ R21										0					0		
	A2NCPU-S1	AJ71AP21-S3										0					0		
	A2ACPU																		
	A2ACPU-S1																		
	A3ACPU	-																	Use the station
	A2UCPU	-																	to set the selection of
	A2UCPU-S1	A 171AP21/																	master or local station.
	A3UCPU	R21										0			0		0		
	A4UCPU	AJ71AP21-S3										-			-		-		
	Q2ACPU	_																	
	Q2ACPU-S1	-																	
	Q3ACPU																		
0.011	Q4ACPU																		
module	Q02CPU																		l la sel amb i fan la sel
+	Q02HCPU																		 Used only for local stations.
link	Q06HCPU	A1SJ71AP23										0			0		0		Installed in an I/O
module	Q12HCPU	Q/N20Q																	slot in an extension
	Q25HCPU																		base unit.
	A1SJHCPU																		
	A1SHCPU																		
	A2SHCPU																		
	A2ASCPU																		
	A2ASCPU-S1	A1SJ71AT21	0																
	A2USHCPU-S1	В	U	0															
	Q2ASCPU																		
	Q2ASCPU-S1																		
	Q2ASHCPU	_																	
	Q2ASHCPU-S1																		
	A1NCPU																		Use the station
	A2NCPU	AJ71AT21B	0	0					0										number setting switch
	A2NCPU-S1		Ŭ																to set the selection of
	A3NCPU				1									1				1	master or local station.

O: Available

									Арр	olicab	le sys	tem							
				Ν	/IELSI	ECNE	T/B d	ata lir	ık				MELS	ECN	ET da	ta link	(
						Seco	nd tier							Thire	d tier	1			
Module	Mod	lel	MEI	SEC	NET	MEL	SEC	MEL	SEC	NET	MEI	SEC	NET	MEL	SEC	MEL	SEC	NET	Remarks
				mode	9	NE ma	I∐ ode	шс	ompo mode	site		mode	•	NE mc	I∐ ode	шс	ompo mode	site	
			M		D	м	1	N4		D	L/m	0	r	1 /m		l /m		r	
			IVI	L	Ň	IVI	-	111	L	IX.	L/111	ł	'	L/111	ł	L/111	ł		
	A2ACPU-S1	-																	
	A3ACPU																		
	A2UCPU	-																	
	A2UCPU-S1	-																	
	A3UCPU	AJ71AT21B	0	0		0	0	0	0										Installed in an I/O slot
	A4UCPU	-																	in an base unit.
0.011	Q2ACPU	-																	
module	Q2ACPU-S1	-																	
+	Q3ACPU																		
link	Q4ACPU	-																	
module	Q02CPU-A																		
	Q02HCPU-A	A1SJ71AT21 B	0	0					0										
	Q06HCPU-A																		
	Q02CPU																		Used only for local
	Q02HCPU	A19 171 AT23																	stations.
	Q06HCPU	BQ		0			0		0										Installed in an I/O
	Q12HCPU																		slot in an extension base unit.
	Q25HCPU																		
	A0J2P25																		
	A0J2P25-S3																		
Data link	A0J2R25												0					0	
module	AJ72P25																		
	AJ/2R25																		
	A15J72125B				0					0									
	AJ/2T25B																		

O: Available

Table 4.3 Link modules available for the three-tier system (Continued)

									Арр	olicab	le sys	tem									
				Ν	1ELSE	ECNE	T/B d	ata lin	ık				MELS	SECN	ET da	ta link	(
						Secor	nd tier							Thir	d tier	1					
Module	Model		MEL	SEC	NET	MEL	SEC	MEL	SEC	NET	MEL	SEC	NET	MEL	SEC	MEI	SEC	NET	Remarks		
				mode	•	NE	TI	I C	ompo	osite		mode	•	NE	TI	Πс	ompo	site			
					_	me	Jue		moue	-			1	IIIC	Jue		moue				
			М	L	R	М	L	М	L	R	L/m	l	r	L/m	l	L/m	l	r			
	A1NCPUP21																				
	A1NCPUP21-S3																				
	A1NCPUR21																				
	A2NCPUP21																				
	A2NCPUP21-S3																				
	A2NCPUR21										*1										
	A2NCPUP21-S1										0										
CPU	A2NCPUP21-S4																				
module	A2NCPUR21-S1																				
with link	A3NCPUP21																				
function	A3NCPUP21-S3 AJ	71AT21B																			
+ link	A3NCPUR21																				
module	A2ACPUP21																				
	A2ACPUP21-S3																				
	A2ACPUR21																				
	A2ACPUP21-S1												*0			*0		*0			
	A2ACPUP21-S4												~2			~2		~2			
	A2ACPUR21-S1										0			0		0			Lise the station		
	A3ACPUP21																		number setting switch		
	A3ACPUP21-S3																		to set the selection of		
	A3ACPUR21																		master or local		
	A2ASCPU																		station.		
	+A1SJ71AP21																				
	A2ASCPU																				
	+A1SJ71AR21																				
	A2ACPU-S1																				
	+A15J71AP21																				
CPU	424CPU-S1 +A1SJ71AR21																				
module																					
function	+A1SJ71AP21 A1	SJ71AT21									*2			*2*3		*2*3					
+	A2USHCPU-S1 B										0			0		0					
link	+A1SJ71AR21																				
modules	Q2ASCPU																				
((100)	+A1SJ71AP21																				
	Q2ASCPU																				
	+A15J/1AK21																				
	4415.1714P21																				
	02ASCPU-S1																				
	+A1SJ71AR21																				

*1: MELSECNET and MELSECNET ${\rm I\!I}\,$ composite modes can be used for the second tier.

*2: MELSECNET, MELSECNET II, and MELSECNET II composite modes can be used for the second tier.

*3: Applicable when the A1SJ71AT21B is used.

Table 4.3 Link modules available for the three-tier system (Continued)

O: Available

							T/D /		App	olicab	le sys	tem				C . P . I			
				N	IELSE		1/B d	ata lir	IK				MELS	SECN		ita link	<		
						Secor	nd tier	-						Thire	d tier	r			
Module	Mod	el	ME	SEC	NET	MEL	SEC	MEI	SEC	NET		SEC		MEL	SEC	MEI	LSEC	NET	Remarks
				mode		NE	ГΠ	Пα	ompo	site				NE	ТΠ	Пc	compo	osite	
				moue		mc	de		mode	•		moue		mo	ode		mode	9	
			м	L I	R	м	I	м	I	R	I /m	D	r	l /m	D	I /m	0	r	
		1		-			-		-		2/111	e	· ·	2/111	ł	2/111	k		
	+A1SJ71AP21																		
	Q2ASHCPU																		
	+A1SJ71AR21																		
	Q2ASHCPU-S1																		
	+A1SJ71AP21																		
	Q2ASHCPU-S1																		
	+A1SJ71AR21																		
	Q02CPU-A	1																	
	+A1SJ71AP21	A1SJ71AT21									*2			*2*3		*2*3			
	Q02CPU-A	В									0			0		0			
	+A1SJ71AR21																		
	Q02HCPU-A																		
	+A1SJ71AP21																		
	Q02HCPU-A																		
	441S.I71AP21																		
	+A1SJ71AR21																		
	A2UCPU						-												
	+AJ71AP21(S3)																		
CPU	A2UCPU																		
module	+AJ71AR21																		Lise the station
with link	A2UCPU-S1																		number setting switch
function	+AJ71AP21(S3)																		to set the selection of
+	A2UCPU-S1																		master or local
link	+AJ/1AR21																		station.
(two)	A3UCPU																		
((100)																			
	+AJ71AR21																		
	A4UCPU																		
	+AJ71AP21(S3)																		
	A4UCPU																		
	+AJ71AR21	A 171 AT21P									*1			*1		*1			
	Q2ACPU	AJ/TATZTB									0			0		0			
	+AJ71AP21(S3)																		
	Q2ACPU																		
	+AJ71AR21																		
	Q2ACPU-S1																		
	+AJ/ 1AP21(S3)																		
	Q2ACPU-S1																		
	+AJ71AP21(S3)																		
	Q3ACPU																		
	+AJ71AR21																		
	Q4ACPU																		
	+AJ71AP21(S3)																		
	Q4ACPU	1																	
	+AJ71AR21																		

*1: MELSECNET and MELSECNET ${\rm I\!I}$ composite modes can be used for the second tier.

*2: MELSECNET, MELSECNET II, and MELSECNET II composite modes can be used for the second tier.

*3: Applicable when the A1SJ71AT21B is used.

4.4 Three-Tier System using the MELSECNET/B Data Link System

When the second and third tiers are configured with MELSECNET/B data link system, connect the stations using shielded twisted pair cables.

- (a) Up to 31 local and remote I/O stations can be connected to a master station for the second tier.
- (b) For the third tier, up to 31 local and remote I/O stations can be connected to the master station, which is the local station for the second tier.

4.4.1 System configuration

Fig 4.5 shows the configuration of the three-tier system.



Fig 4.5 Three-tier system configured with MELSECNET/B data link system

POINT	
(1) Remote I/(2) MELSECI(including)	O stations cannot be connected in MELSECNET II mode. NET/B data link system has no restriction on the order of stations master stations).

4.4.2 Precautions when using data link

This section explains precautions for configuring a three-tier system.

(1) Setting link parameters

In the three-tier system, setting link parameters to CPU modules for the master stations in the second and third tiers is required.

For link parameter setting, refer to Section 5.3.7 and Chapter 7.



(2) Restriction on use of link modules

Up to two of the following link modules (one as a master station and the other as a local station) can be used with a CPU module.

(The two modules cannot be used only for master stations or local stations.)

- AJ71AT21B
- A1SJ71AT21B



4.4.3 System devices

									Арр	olicabl	e sys	tem							
								Ν	1ELSE	ECNE	T/B d	ata lir	ık						
						Seco	nd tie	r						Third	d tier				
Module	Мо	del	MEI	_SEC mode	NET	MEL NE mo	SEC TII ode	MEI II C	SECI	NET site	MEL	_SEC mode	NET	MEL NE mc	SEC TII ode	MEL II C	.SECI ompo mode	NET	Remarks
			М	L	R	М	L	М	L	R	L/m	l	r	L/m	l	L/m	l	r	
	A1SJHCPU																		
	A1SHCPU	A1SJ71T21B																	
	A2SHCPU		-	_					-			_					-		
			0	0					0			0					0		
		AJ71AT21B																	
	A3NCPU																		
	A2ASCPU																		
	A2ASCPU-S1																		
	A2USHCPU-S1																		Use the station
	Q2ASCPU	A1SJ71AT21B																	number setting
	Q2ASCPU-S1																		switch to set the
	Q2ASHCPU																		selection of
	Q2ASHCPU-S1																		master or local
	A2ACPU																		station.
	A2ACPU-S1																		
CPU	A3ACPU																		 Installed in an I/O
module	A2UCPU		0	0		0	0	0	0			0			0		0		slot in a base
	A2UCPU-S1																		unit.
	A3UCPU	AJ71AT21B																	
	Q02HCPU-A	A1SJ71AT21B																	
	Q06HCPU-A																		
	Q02CPU																		Used only for
	Q02HCPU																		local stations.
	Q06HCPU	A 4 C 174 AT 22 D C		~					~			~			•		~		Installed in an I/O
	Q12HCPU	ATSJ/TAT23BQ		0			0		0			0			0		0		slot in an
	Q25HCPU																		extension base unit.
Data link	A1SJ71T25B									~									
module	AJ72T25B				0					0			0					0	

Table 4.4 Link modules available for the three-tier system

REMARK

- (1) The definitions of L/m station, ℓ station, and r station in Table 4.4 are as follows.
 - a) L/m station Local station in the second tier/master station in the third tier
 - b) ℓ station Local station in the third tier
 - c) r station Remote I/O station in the third tier



MELSEC-A

O: Available

MELSEC-A

Table 4.4 Link modules available for the three-tier system (Continued)

O: Available

									Арр	olicab	le sys	tem							
								Ν	1ELSE	ECNE	T/B d	ata lin	ık						
						Seco	nd tier	r 						Thir	d tier				
Module	Mode		ME	LSEC	NET		SEC τπ	MEI		NE I Neite	MEL	SEC	NET		SEC τπ	MEI		NET	Remarks
				mode	•	mo	ode	пС	mode			mode	1	m	ode	ш.	mode)	
			М	L	R	М	L	М	L	R	L/m	P	r	L/m	P	L/m	P	r	
	A2ASCPU +A1SJ71AT21B A2ASCPU-S1 +A1SJ71AT21B A2USHCPU-S1 +A1SJ71AT21B Q2ASCPU +A1SJ71AT21B Q2ACPU-S1 +A1SJ71AT21B Q2ASHCPU +A1SJ71AT21B	A1SJ71 AT21B									*2 O			*1*2		*1*2 O			
CPU module with link function + link modulos	Q2ASHCPU-S1 +A1SJ71AT21B Q02CPU-A +A1SJ71AT21B Q02HCPU-A +A1SJ71AT21B Q06HCPU-A +A1SJ71AT21B																		 Use the station number setting switch to set the selection of master or local station. Installed in an
(two)	A2UCPU +AJ71AT21B A2UCPU-S1 +AJ71AT21B A3UCPU +AJ71AT21B A4UCPU +AJ71AT21B Q2ACPU +AJ71AT21B Q2ACPU-S1 +AJ71AT21B Q3ACPU +AJ71A+21B Q4ACPU +AJ71AT21B	AJ71AT21B									*2 0			*2 O		*2 0			I/O slot in a base unit.

*1: Applicable when the A1SJ71AT21B is used.

*2: MELSECNET, MELSECNET II or MELSECNET II composite modes can be used for the second tier.

MELSECNET mode		MELSECNET			MELSECNET/E	5
Operating Mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode
Applicability	0	0	0	0	0	0

MELSEC-A

5 SPECIFICATIONS

This chapter describes the general specifications of the data link system and the performance specifications and functions of the link modules.

5.1 General Specifications

For general specifications, refer to the user's manual for the CPU module used.

5.2 Performance Specifications

The following shows the performance specifications of the data link system. Table 5.1 Performance specifications

		M	ELSECNET data link syste	em							
Itom			Optical data link system								
nem		MELSECNET	MELSECNET II	MELSECNET II							
		mode	mode	composite mode							
Max, available of light points wood you	Input (X)	Up to the maximum I/O	points allowed for master	station CPU module *2							
Max. number of link points used per	[(/	(Total n	umber of slave station link	c points)							
station	Output (Y)	= (Avail	able link points for master	station)							
	В	1024 points (128 bytes)	4096 points	(512 bytes)							
Max. number of link points per system	W	1024 points (2048 bytes)	4096 points	(8192 bytes)							
	Master station	1004 hites	1024 bytes (link pa	rameters; first half)							
	Local station	1024 bytes	1024 bytes (link para	ameters; second half)							
Max. number of link points per station	Remote I/O station	512 bytes Number of I/O points : 512 points	-	512 bytes Number of I/O points : 512 points							
Communication speed			1.25Mbps								
Communications method			Half-duplex bit serial								
Synchronization method			Frame synchronization								
Transmission path		Duplex loop									
		For SI	or H-PCF type optical fibe	er cable							
Overall cable distance *1		: Up to 10km (3	2810ft.) (Between stations	s: 1km (3281ft.))							
		F	or GI type optical fiber cab	ble							
		: Up to 10km (3	2810ft.) (Between stations	s: 2km (6562ft.))							
Number of connectable stations		Up to 65 stations/loop	o (1 master station, 64 loca	al/remote I/O stations)							
Modulation method			CMI								
Transmission format		Conf	forms to HDLC (frame me	thod)							
Error control system		Retry by CRC (generating polynomial $X^{16} + X^{12} + X^5 + 1$) and timeout									
		The loopback function activated by detection of an error or cable									
RAS functions		disconnection									
		The diagnostic funct	tions such as link line che	ck of the host station							
Cable used			Optical fiber cable								
Transmission loss		SI: N	lax. 12dB/km GI: Max. 3d	B/km							
Sending level		SI: -17 to -11 dBm	(peak value) GI: -17 to -10) dBm (peak value)							
Receiving level		SI: -32 to -11 dBm (peak value) GI: -29 to -10 dBm (peak value)									

REMARK

*1: Overall distances

MELSECNET The overall cable distance means a distance from OUT of the master station to IN of the master station via a slave stations.
 MELSECNET/B...... Distance between the farthest link modules. Overall distance Overall distance (12)

(L1)

(12)

[L3]

MELSECNET

MELSECNET mode		MELSECNET			MELSECNET/B	5
Operating Mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode
Applicability	0	0	0	0	0	0

MELSEC-A

Table 5.1 Performance specifications (Continued)

MELS	ECNET data link syst	em	MELSE	ECNET/B data link sys	stem		
C	oaxial cable data link		Shielded twi	sted pair cable data lir	nk system		
MELSECNET	MELSECNET I	MELSECNET I	MELSECNET	MELSECNET I	MELSECNET I		
mode	mode	composite mode	mode	mode	composite mode		
	Up to the maxi (Total number of sla	mum I/O points allowe ve station link points)	ed for master station CPL = (Available link points fo	J module *2 or master station)			
1024 points (128 bytes)	4096 points	(512 bytes)	1024 points (128 bytes)	4096 points (512 bytes)			
1024 points (2048 bytes)	4096 points	(8192 bytes)	1024 points (2048 bytes)	4096 points	(8192 bytes)		
1024 bytes	1024 bytes (link pa 1024 bytes (link para	rameters; first half) ameters; second half)	1024 bytes	1024 bytes (link pa 1024 bytes (link para	rameters; first half) meters; second half)		
512 bytes Number of I/O points : 512 points	-	512 bytes Number of I/O points : 512 points	512 bytes Number of I/O points : 512 points	-	512 bytes Number of I/O points : 512 points		
	1.25Mbps		125kbps	s/250kbps/500kbps/1	Mbps		
ł	Half-duplex bit serial						
		Frame sync	hronization				
-	Duplex loop		Bus system				
U (Betweer	p to 10km (32810ft.) n stations: 500m (1640	D.5ft.))	Varies depending on the communication speed (125kbps: 1200m, 250kbps: 600m, 500kbps: 400m, 1Mbps: 200m)				
U	p to 65 stations/loop		U	p to 32 stations/loop			
(1 master stat	ion, 64 local/remote I/	O stations)	(1 master stat	ion. 31 local/remote l/	O stations)		
	CMI			NRZI method			
		Conforms to HDL	C (frame method)				
	Retry by CRC	C (generating polynom	ial X ¹⁶ + X ¹² + X ⁵ + 1) ai	nd timeout			
The loopback function	activated by detection	of an error or cable					
	disconnection		The diagnostic functions	s such as link line che	ck of the host station		
The diagnostic functions	s such as link line che	ck of the host station					
	Coaxial cable		Shie	elded twisted pair cabl	e		
	-			-			
	-			-			
	-			-			

*2: The A1SHCPU, A1SJHCPU, A2SHCPU, A2ASCPU(S1), A2USHCPU-S1,

Q2ASCPU(S1),Q2ASHCPU(S1), A2UCPU(S1), A3UCPU, A4UCPU, Q2ACPU(S1),

Q3ACPU, Q4ACPU, Q4ARCPU, Q02CPU-A, Q02HCPU-A, Q06HCPU-A can use up to 2048 points.

5.3 Functions

The following shows the functions of the data link system.

Table 5.2 List of data link system functions

Itom	Description	
nem	MELSECNET data link system	
Cyclic transmission function	 (1) Function to periodically communicate data between the master station and the slave stations (local station and remote I/O station). (2) The cyclic transmission function includes the following two communications. One-to-one communication between the master station and a slave station Communication between the master station and all local stations (a) One-to-one communication between the master station and a slave station and a remote I/O station and outputs (Y) are used for communication between the master station and a local station. (b) Communication between the master station and all local stations (c) Communication between the master station and all local stations (c) Communication between the master station and all local stations 	
Transient transmission function	 (1) Function to communicate only when it is requested. (2) The transient transmission function includes the following three types. (a) The master station executes an LRDP or LWTP instruction to read/write devices (T,C,D,W) of the local station. (b) The master station executes an RFRP or RTOP instruction to read/write buffer memory of a special function module mounted to a remote I/O station. (c) Peripheral devices installed to the programmable controller CPU accesses to other stations. 	
Automatic return function	Even if a local station or a remote I/O station is disconnected from the data link system due to a fault, this function automatically returns the disconnected station to the system when the normal operation state is restored.	
Loopback function	(1) Because of double configuration of optical fiber cable and coaxial cable, when a cable breaks or a local or remote I/O station is disconnected from the system, the faulty part is removed so that the normal operation can be maintained by normally operating stations.	
Error detection	(1) Faulty parts can be detected by reading data of the special relays and special registers.	
Self-diagnostics function	(1) Checks link module hardware and optical fiber cable/coaxial cable.	
Extensive use of link relays (B) and link registers (W) in three-tier system	 (1) The link relays (B) and link registers (W) that can be used in a three-tier system are as follows: MELSECNET modeB/W0 to 3FF (1024 points) MELSECNET II modeB/W0 to FFF (4096 points) MELSECNET IIB/W0 to FFF (4096 points) (2) Function to redundantly use the same device number in the three-tier system if the available number of points of link relays (B) and link registers (W) in the three-tier system is not sufficient. 	
MELSECNET mode/ MELSECNET II mode/ MELSECNET II composite mode	 (1) The data link system includes three different operation modes: MELSECNET mode, MELSECNET II mode, and MELSECNET II composite mode. (2) The above-mentioned operation mode can be selected and used depending on the data link module to be used. The data link for the entire MELSECNET II device range (B/W0 to FFF) while maintaining compatibility with the conventional MELSECNET mode. 	

MELSECNET mode		MELSECNET		MELSECNET/B				
Operating Mode	MELSECNET MELSECNET		MELSECNET II composite mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode		
Applicability	0	0	0	0	0	0		

MELSEC-A

Table 5.2 List of data link system functions (Continued)

Description	Reference section
MELSECNET/B data link system	Reference section
	Section 5.3.1
	Section 5.3.2
	Section 5.3.3
-	Section 5.3.4
	Section 5.3.5
(1) Checks link module hardware and shielded twisted pair cables.	Section 5.3.6
	Section 5.3.7
	Section 5.3.8

MELSECNET mode		MELSECNET		MELSECNET/B				
Operating Mode	MELSECNET MELSECNET I mode		MELSECNET II composite mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode		
Applicability	0	0	0	0	0	0		

MELSEC-A

5.3.1 Cyclic transmission function

The cyclic transmission function is a function to periodically communicate data between the master station and the slave stations (local station or remote I/O station). The cyclic transmission function includes the following two communications.

- One-to-one communication between the master station and slave stations
- Communication between the master station and all local stations.
- (1) One-to-one communication

Inputs (X) and outputs (Y) are used for communication between the master station and a remote I/O station and communication between the master station and a local station.

- (a) Communication between the master station and a remote I/O station
 - A master station receives ON/OFF data (inputs (X)) from an input module installed in a remote I/O station, and the master station outputs the operation results (outputs (Y)) (obtained by the master station sequence program) to the output module installed in the remote I/O station.



Receives ON/OFF data from an input module installed in the remote I/O station

- For communication between the master and remote I/O stations, the number of I/O points of the remote I/O station must be assigned to the I/O number used on the master station by link parameters of the master station beforehand.
 - a) When the input module in the remote I/O station is turned ON, the input (X) of the master station assigned by the link parameter is also turned ON.
 - b) When the output (Y) assigned with the link parameter of the master station is turned ON, the output module on the remote I/O station turns ON.



- (b) Communication between the master station and a local station
 - Communicates data between the master station and a local station using some of I/O points in the master station and local stations for the data link. (Communication between local stations or communication between a local station and a remote I/O station is not possible.)
 - 2) As for communication between the master station and a local station, the sender uses outputs (Y), and receiver uses inputs (X).



3) When communicating between the master station and a local station, assign the inputs and outputs used for the data link by a link parameter. The link parameter defines the correspondence between the master station inputs (X) and the local station outputs (Y) and between the master station outputs (Y) and the local station inputs (X).



(2) Communication between the master station and all local stations
 Link relays (B) and link registers (W) are used for communication between the master
 station and a local station or communication between local stations.
 Link relays (B) and link registers (W) are used by the master station and all local
 stations in common.

Link relays (B) and link registers (W) send data to other station, using the range assigned to the host station with a link parameter.

- Link relays (B) are internal relays for the data link and are used for sending ON/ OFF data.
- Link registers (W) are data registers for the data link and are used for sending 16bit data.
- Link relays (B) and link registers (W) handle different types of data.
 However, the range of communicating with other stations in the data link system is the same.

The following explains the range in which communication is possible with link relays (B) and link registers (W) when using the MELSECNET mode.

In the MELSECNET II mode and MELSECNET II composite mode, although the function is basically the same, stations with which communication is possible partly differ between the range assigned to the first half of the link parameter and the range assigned to the second half of the link parameter. (For details, refer to Section 5.3.8.)

(a) Two-tier system configuration

example.

 The master station and local stations can read the entire range of link relays (B) and link registers (W) assigned by the link parameter. The read/write range of each station is explained by showing Fig 5.1 as an



Fig 5.1 Two-tier system configuration

[Link parameter setting range]





When B0 is turned ON in the M station, for example, B0 in the L1, L2, and L3 stations is also turned ON.

Local station No. 1 (L1)

B/W0	100	2	200	300	380	<u>3F</u> F
	M (R)	L1 (W)	L2 (R)	L (F	3 7)	**

When B100 is turned ON in the L1 station, for example, B100 in the M, L2, and L3 stations is also turned ON.

)	Local station No. 2 (L2)
---	--------------------------



When B200 is turned ON in the L2 station, for example, B200 in the M, L1, and L3 stations is also turned ON.



When B300 is turned ON in the L3 station, for example, B300 in the M, L1, and L2 stations is also turned ON.

REMARK

To simplify the example, the same number is assigned to link relays (B) and link registers (W). In actual use, the number can be separately assigned to link relays (B) and link registers (W).

- (b) Three-tier system configuration
 - 1) In the configuration of the three-tier system, the link parameters are set in the master station for the second tier and the master station for the third tier.
 - 2) The master station and local stations for the second tier (including the master station for the third tier) can read the range assigned by the link parameters of the master station for the second tier.
 - 3) The local stations for the third tier can read the entire range assigned by the link parameters of the master station for the third tier and the range assigned to host station (the master station for the second tier) by the link parameters of the master station for the second tier.

The range assigned to the local stations for the second tier and the range assigned by link parameters of the master station for other third tier cannot be read.

The read/write range of each station is explained by showing Fig 5.2 as an example.



Fig 5.2 Three-Tier system configuration



 The M station writes data to the devices in the B/W0 to FF range and sends it to the other stations.

When B0 is turned ON in the M station, for example, B0 in other stations are also turned ON.

- The M station can receive data written to the devices in the B/W100 to 37F range by other stations.
- Devices in the B/W380 to 3FF range can be used instead of internal relays (M) and data registers (D).

B/	N0	100	180 1	C0 2	00 28	0 20	20 30	00 3	80	_3FF	-
		L1/r (R)	n ℓ1) (R)	ℓ2 (R)	L2/m (R)	ℓ'1 (R)	ℓ'2 (R)	L3 (R)		**	

[Range where data can be received, range where data can be sent]



[Range where M can receive data]*



[Range where M can send data]*

Local station No. 1 (Master station (L1/m) for third tier-1)

- The L1/m station writes data to the devices in the B/W100 to 17F range and sends it to the other stations.
 When B100 is turned ON in the L1/m station, for example, B100 in other stations are also turned ON. (Except for ℓ '1 and ℓ '2 stations)
- 2) The L1/m station can receive data written to the devices in the B/W0 to B/ WFF range and the B/W180 to 37F range by other stations.
- Devices in the B/W380 to 3FF range can be used instead of internal relays (M) and data registers (D).

B/V	V01	00 18	0 1	<u>C0 2</u>	00 28	0 20	0 3	00	380	<u>3F</u> F
	M (R)	L1/m (W)	ℓ1 (R)	ℓ2 (R)	L2/m (R)	ℓ'1 (R)	ℓ'2 (R)	L3 (R)		**

[Range where data can be received, range where data can be sent]





[Range where L1/m can receive data]*

[Range where L1/m can send data]*

REMARK

- To simplify the example, the same number is assigned to link relays (B) and link registers (W). In actual use, the number can be separately assigned to link relays (B) and link registers (W).
- 2) *: \bigcirc indicates the range where data can be received.

Local station No. 2 (Master station (L2/m) for third tier-2)

 The L2/m station writes data to the devices in the B/W200 to 27F range and sends to other stations.

When B200 is turned ON in the L2/m station, for example, B200 in other stations are also turned ON. (Except for ℓ 1 and ℓ 2 stations)

- The L2/m station can receive data written to the devices in the range of B/W0 to 1FF and B/W280 to 37F by other station.
- Devices in the range of B/W380 to 3FF can be used instead of internal relays (M) and data registers (D).

B/W0	100 18	30_1	C0 2	00 28	80 20	20 3	00	380	3FF
M (R)	L1/m (R)	ℓ1 (R)	ℓ2 (R)	L2/m (W)	ℓ'1 (R)	ℓ'2 (R)	L3 (R)		**

[Range where data can be received, range where data can be sent]





[Range where L2/m can receive data]*

Local station No. 3 (L3)

1) The L3 station writes data to the devices in the range of B/W300 to 37F and sends it to other stations.

When B300 is turned ON in the L3 station, for example, B300 in other stations are also turned ON. (Except for ℓ 1, ℓ 2, ℓ '1 and ℓ '2 stations.)

- The L3 station can receive data written to the devices in the range of B/W0 to 2FF by other station.
- Devices in the range of B/W380 to 3FF can be used instead of internal relays (M) and data registers (D).

B/ <u>W0</u>	100	180	1C0 2	00 28	0 20	0 3	оо з	80 3FF
M (R)	L1/m ℓ1 (R) (R) (R)	L2/m (R)	ℓ'1 (R)	ℓ'2 (R)	L3 (W)	**

[Range where data can be received, range where data can be sent]





[Range where L3 can receive data]*

[Range where L3 can send data $]^*$

[[]Range where L2/m can send data]*

Local station No. 1 (ℓ 1) in the third tier-1

1) The ℓ 1 station writes data to the devices in the range of B/W180 to 1BF and sends it to the other stations.

When B180 is turned ON in the ℓ 1 station, for example, B180 in other stations are also turned ON. (Except for ℓ '1 and ℓ '2 stations)

- 2) The ℓ 1 station can receive data written to the devices in the range of B/W0 to 17F and B/W1C0 to 1FF by other stations.
- Devices in the range of B/W200 to 3FF can be used instead of internal relays (M) and data registers (D).

B/	NO 10	00 18	80 1	C0 2	00 28	80 20	CO 3	00	380	3FI	F
	M (R)	L1/m (R)	ℓ1 (W)	ℓ2 (R)	L2/m **	ℓ'1 **	ℓ'2 **	L3 **		**	

[Range where data can be received, range where data can be sent]





[Range where ℓ 1 can receive data]*

[Range where l1 can send data]*

Local station No. 2 (ℓ 2) in third tier-1

1) The ℓ 2 station writes data to the devices in the range of B/W1C0 to 1FF and sends it to the other stations.

When B1C0 is turned ON in the ℓ 2 station, for example, B1C0 in other stations are also turned ON. (Except for ℓ '1 and ℓ '2 stations)

- 2) The ℓ 2 station can receive data written to the devices in the range of B/W0 to 1BF by other stations.
- Devices in the range of B/W200 to 3FF can be used instead of internal relays (M) and data registers (D).



[Range where data can be received, range where data can be sent]





[Range where ℓ^2 can receive data]*

[Range where $\ell 2$ can send data]*

Local station No. 1 (ℓ '1) in third tier-2

1) The $\,\ell$ '1 station writes data to the devices in the range of B/W280 to 2BF and sends it to other stations.

When B280 is turned ON in the ℓ '1 station, for example, B280 in other stations are also turned ON. (Except for ℓ 1 and ℓ 2 stations)

- 2) The ℓ '1 station can receive data written to the devices in the range of B/W0 to FF, B/W200 to 27F, and B/W2C0 to 2FF by other station.
- 3) Devices in the range of B/W100 to 1FF and B/W300 to 3FF can be used instead of internal relays (M) and data registers (D).

B/ <u>W0</u>	10	0 180	0 1	<u>C0 2</u>	00 28	0 20	0_3	00 3	880	3FF
	M (R)	L1/m	ℓ1 **	ℓ2 **	L2/m (R)	ℓ'1 (W)	ℓ'2 (R)	L3 **		**

[Range where data can be received, range where data can be sent]





[Range where ℓ '1 can receive data]*

[Range where l'1 can send data]*

Local station No. 2 (ℓ '2) in third tier-2

1) The ℓ '2 station writes data to the devices in the range of B/W2C0 to 2FF and sends it to other stations.

When B2C0 is turned ON in the ℓ '2 station, for example, B2C0 in other stations are also turned ON. (Except for ℓ 1 and ℓ 2 stations)

- 2) The ℓ '2 station can receive data written to the devices in the range of B/W0 to FF and B/W200 to 2BF by other station.
- 3) Devices in the range of B/W100 to 1FF and B/W300 to 3FF can be used instead of internal relays (M) and data registers (D).

B/	wo	100	180	100	200	280 2	C0_3	00 3	80	3FF
1	M (R)	L1	l/m ℓ	1 &2 * *	2 L2/i	m <i>l</i> '1) (R)	ℓ2 (W)	L3 **		**

[Range where data can be received, range where data can be sent]





[Range where l'2 can receive data]*

[Range where $\ell\,{}^{\prime}2$ can send data $\,\,{}_{}]^*$

REMARK

1) *:
indicates the range where data can be received.

MELSECNET mode		MELSECNET			MELSECNET/B	
Operating Mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode
Applicability	0	0	0	0	0	0

MELSEC-A

5.3.2 Transient transmission function

The transient transmission is a function to:

- Read/write from/to devices (T, C, D, and W) in a local station from the master station;
- Read/write the buffer memory of a special function module in a remote I/O station from the master station;
- Enable communications from a peripheral device connected to the programmable controller CPU to the other programmable controllers.
- (1) Read/write of devices (T, C, D, and W) from the master station to a local station The transient transmission function reads/writes the link registers (W) whose range is not set by link parameters or the devices (T, C, D) where cyclic transmission is not possible.

An LRDP or LWTP instruction is executed in the master station sequence program to read/write devices (T, C, D, and W) of a local station. (Refer to Section 9.7 for details on the LRDP and LWTP instructions.)



(2) Read/write of buffer memory in a special function module installed to a remote I/O station

Use RFRP or RTOP instruction to read/write buffer memory of a special function module installed to a remote I/O station from the master station.

Data communication can be executed between the master station and a remote I/O station only when RFRP or RTOP instruction is executed.

To use RFRP or RTOP instruction, link registers (W) are assigned to the remote I/O station with the link parameter.

The buffer memory data is read or data is written to the buffer memory in the special function module loaded to the remote I/O station with the link registers. (Refer to Section 9.8 for details on RFRP and RTOP instructions.)

Example: The following results when reading data of the buf 10 to 17) from a special function module (X/Y280 to remote I/O station No.1 to W200 to 207.	fer memory (address o X/Y29F) loaded in
	Special function module loaded to remote I/O station
Master station	Buffer memory
W200 to Execution of RFRP instruction	Address
w207	
Program	
RFRP H280 K10 W200 K8	
Head device number of special function	Data number to be read device number of devices storing lata er of devices storing data to be read n module

(3) Accessing other station with a peripheral device connected to programmable controller CPU

When accessing to other station from the peripheral device, the station which can be accessed depends on types of access source station (master station, local station, or remote I/O station).

The basic concept is illustrated in Fig 5.3. For details on the executable functions, refer to the manuals for the peripheral device and module to be used.

(1) Accessible station when connecting peripheral devices

- When peripheral device is connected to the master station, host station, a local station, and a remote I/O station are accessible.
- When peripheral device is connected to a local station, host station and the master station are accessible.
- When peripheral device is connected to a remote I/O station, the master station is only accessible.



Access	Access target station								
source station	М	L1	L2	R3	R4				
М	0	0	0	O*1	O*1				
L1	0	0	×	×	×				
L2	0	×	0	×	×				
R3	0	×	×	×	×				
R4	0	×	×	×	×				
*1: Not accessib	le when using GX	Developer.		•					

Fig 5.3 Accessibility between link modules

In the system configuration in Fig 5.3, an A6GPP connected to local station No. 1 (L1) can execute the following operations for the master station.

- · Program read/write
- Monitor
- Test
- Remote RUN/STOP/PAUSE

MELSECNET mode		MELSECNET			MELSECNET/B	
Operating Mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode
Applicability	0	0	0	0	0	0

MELSEC-A

5.3.3 Automatic return function

(1) Automatic return function

If an error occurs to a local station or a remote I/O station, it is disconnected from the data link system to maintain the data link with the normally operating stations. The disconnected local station or remote I/O station is automatically returned to the system when the normal operation state is restored.

The function that makes this possible is the automatic return function.

 Setting with/without automatic return function
 The link module connected to the data link system can select whether or not to use the automatic return function for each station.
 Refer to the user's manual for each link module for details on whether or not to use the automatic return function.

(3) Method for reconnecting a disconnected station with/without automatic return function

The method for reconnecting a disconnected station varies depending on whether the automatic return function is enabled or not.

- (a) When data link stops due to an error in the master station
 - Automatic return function is set for the master station: After resetting the master station, reset all of the local and remote I/O stations for which the automatic return function is set.
 - Automatic return function is not set for the master station: After resetting all of the local and remote I/O stations for which the automatic return function is set, reset the master station.

(b)	When a local	or remote I/O	station is	disconnected	due to an error
-----	--------------	---------------	------------	--------------	-----------------

Master station	Local/Remote I/O station	Conditions for return
	Enabled	Returns automatically after the error is removed.
Enabled	Disabled	Reset the disconnected station and the local and remote I/O stations that do not have the automatic return function.
Disabled	Enabled	Reset the disconnected station and the local and remote I/O stations
Disabled	Disabled	master station.

Enabled: Automatic return function selected Disabled: Automatic return function not selected

MELSECNET mode		MELSECNET			MELSECNET/B	
Operating Mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode
Applicability	0	0	0			

MELSEC-A

5.3.4 Loopback function

(1) Loopback function

The MELSECNET data link system has a double configuration of link cables (optical fiber or coaxial).

The loopback function uses the double configuration to isolate the faulty part and maintain the data link with the normally operating stations when a cable is broken or a local station or remote I/O station is disconnected.

POINT

- (1) In the MELSECNET data link system, the station number of slave stations and sub-slave stations should be set in order (from the station No.1 to the station No. n in the forward loop direction) for the loopback function. (For details, refer to Section 8.2.)
- (2) The loopback function may not work depending on the fault of the data link module.

Identify the faulty data link module in the following method.

1) Check the LED indications ("RUN LED" off, "ERROR LED" on) of all data link modules for the faulty station.

Refer to Section 10.4 for the ERROR LED indications.

2) Turn off all stations power supply, and turn them on in order, starting with the master station.

At this time, check that which station stops the normal data link.

Replace the fault-detected data link module and then make sure that a data link returns to normal.

(2) Normal data link

Under normal operating status, the data link uses the forward loop. Loop data is sent/ received in the following order: master station, station No. 1, station No. 2, etc.



Fig 5.4 Normal data link (When one master station and eight slave stations are used)

(3) Data link when the forward loop is faulty

If data link using the forward loop is disabled due to a broken cable or a problem with a forward loop cable connector, the loop is automatically switched from "forward" to "reverse" to maintain the data link.

In the reverse loop, link data is sent in the following order:

master station \rightarrow station No. n \rightarrow station No. (n-1),

Data link operation when a forward loop cable is broken or disconnected is illustrated in Fig 5.5.



Fig 5.5 Reverse loop data link (When one master station and eight slave stations are used)

(4) Data link when both the forward and reverse loops are faulty When a fault occurs to the cables of forward and reverse loops and the cable connecting connector, the link loops back toward the master station in front of faulty station, and data link is maintained by the normally operating. (Faulty stations are all disconnected.)

When the faulty part is returned to normal, the data link in the forward loop is recovered. Whether the station will remain disconnected or return to the data link depending on the setting of the automatic return function. (Refer to Section 5.3.3.)



Fig 5.6 Data link when forward/reverse loop is faulty (When one master station and eight slave stations are used)

(5) Data link when a power failure occurs to local station or remote I/O station If the data link is disabled due to power failure at a local station or a remote I/O station, the link loops back toward the master station, and the data link is maintained by the normally operating stations. (The station where the power failure occurred is disconnected from the data link.)

When the power supply to the disconnected station is turned on, the data link in the forward loop recovers. Whether the station will remain disconnected or return to the data link is depending on the setting of the automatic return function. (Refer to Section 5.3.3.)



Fig 5.7 Local station or remote I/O station power failure data link (When one master station and eight slave stations are used)

MELSECNET mode		MELSECNET			MELSECNET/B	
Operating Mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode
Applicability	0	0	0	0	0	0

MELSEC-A

5.3.5 Fault detection function

In the data link system, the data link operation status is stored in the special relays (M) and special registers (D) so that it can be easily checked with a sequence program and peripheral device.

Some special relays (M) and special registers (D) can be read by the master station, while some can be read by a local station.

For details of special relays and special registers, refer to Section 9.2.

The following describes the major events that are detected by the fault detection function.

When MELSECNET data link system is used

- (1) Events checked by the master station
 - (a) Data communication status with a slave station (local station and remote I/O station)
 - Stores stations at which a communication error occurs. (D9228 to D9231)
 - Stores stations at which initial communication, necessary for starting communication, has not been completed. (D9224 to D9227)
 - Stores the operation status (RUN/STEP-RUN or STOP/PAUSE) of a local station. (D9212 to D9215)
 - (b) Link parameter error
 - Checks whether the link parameter is set or the data is correct. (M9206)
 - Checks consistency (whether the B/W assignment range overlaps) between the link parameters set on a slave station (master station for the third tier) and those set for the host station.(M9207)
 - (c) Number of communication error occurrences
 - The accumulated number of retries due to transmission errors. (D9210)
 - The accumulated number of receive error occurrences. (D9240)
 - (d) Link card hardware error (M9210)
 - (e) Mode setting switch setting status in the link card
 - Whether the mode setting switch is set for online (0 or 1) or offline test mode (2 to 7). (M9224)
 - (f) Data link status
 - Forward loop error (M9225)
 - Reverse loop error (M9226)
 - Stores information whether data is sent via the forward loop, reverse loop, or forward/reverse loop. (D9204)
 - Stores the stations where loopback occurs. (D9205, D9206)
 - Stores information at which points errors occur in the forward loop line and reverse loop line. (D9232 to D9239)

- (2) Events checked by local stations
 - (a) Data communication status with the master station
 - Checks if cyclic communication is executed normally.(M9246)
 - Checks if cyclic communication is executed normally from the master station for the second tier when the host station is a local station for the third tier. (M9247)
 - Checks if the link parameters are received from the master station. (M9250)
 - Checks if the host station can communicate data. (M9251)
 - (b) Data communication status with other stations
 - Check a faulty local station other than the host station. (D9252 to D9255)
 - The operation status of local stations other than the host station. (RUN/STEP-RUN or STOP/PAUSE)
 - (c) Number of receive errors
 - The accumulated number of receive error occurrences. (D9245)
 - (d) Link card hardware error (M9211)
 - (e) Mode setting switch setting status in the link card (M9240)
 - (f) Data link status
 - Forward loop error (M9241)
 - Reverse loop error (M9242)
 - Loopback in the host station (M9243)

When MELSECNET/B data link system is used

- (1) Events checked by the master station
 - (a) Data communication status with a slave station (local station and remote I/O station)
 - Stores stations in which a communication error occurs. (D9228, D9229)
 - Stores stations at which initial communication, necessary for starting communication, has not been completed. (D9224 and D9225)
 - Stores the operating state (RUN/STEP-RUN or STOP/PAUSE) of a local station. (D9212, D9213)
 - (b) Link parameter error
 - Checks if the link parameters for the host station is not set, or the data has an error. (M9206)
 - Checks consistency (whether the B/W assignment range overlaps) between link parameters set for a slave station (master station for the third tier) and those set for the host station. (M9207)
 - (c) Number of communication errors
 - · Stores the accumulated number of retries due to transmission errors. (D9210)
 - Stores the accumulated number of receive errors. (D9240)
 - (d) Link card hardware fault (M9210)
 - (e) Mode setting switch setting status of link card
 - Checks if the mode switch is set to online (0 or 1) or offline (2 to 7). (M9224)
 - (f) Data link statusStores the data link status. (D9204)

- (2) Events checked by a local station
 - (a) Data communications status with the master station
 - Checks if cyclic data communications is normally executed. (M9246)
 - Checks if cyclic communication is executed normally from the master station for the second tier when the host station is a local station for the third tier. (M9247)
 - · Checks if link parameters are received from the master station. (M9250)
 - Checks if the host station can communicate data. (M9251)
 - (b) Data communications with another station
 - Check a faulty local station other than the host station. (D9252, D9253)
 - The operation status of local stations other than the host station. (RUN/STEP-RUN or STOP/PAUSE)
 - (c) Number of receive errors
 - Checks the number of receive error occurrences (D9245).
 - (d) Link card hardware error (M9211)
 - (e) Mode setting switch setting status in the link card (M9240)

MELSECNET mode		MELSECNET			MELSECNET/B		
Operating Mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	
Applicability	0	0	0	0	0	0	

MELSEC-A

5.3.6 Self-diagnostics function

The self-diagnostics function is a function to check the link module hardware and detect errors like breaking of the cables for data link.

The self-diagnostics function includes the following three tests.

- Self loopback test Checks the link module hardware independently, including the send/receive circuit for transmission system.
- (2) Station-to-station test Checks the wiring of cable and the link module hardware for the two adjacent stations (master station between local station, local station between local station, etc.)
- (3) Forward loop/reverse loop test (Applicable only when MELSECNET data link system is used)

Checks the data link line for the forward loop, reverse loop, and loopback mode when wiring of the all MELSECNET data link system has been connected

REMARK

- 1) *: The cables for the data link are as follows:
 - · MELSECNET data link system Optical fiber cable Coaxial cable
 - · MELSECNET/B data link system Shielded twisted pair cable
- 2) Refer to Section 8.7 for the self-diagnostics test method
| MELSECNET mode | | MELSECNET | | | MELSECNET/B | | |
|----------------|-------------------|----------------------|--------------------------------|-------------------|----------------------|--------------------------------|--|
| Operating Mode | MELSECNET
mode | MELSECNET II
mode | MELSECNET II
composite mode | MELSECNET
mode | MELSECNET II
mode | MELSECNET II
composite mode | |
| Applicability | 0 | 0 | 0 | 0 | 0 | 0 | |

MELSEC-A

5.3.7 Extensive use of link relays (B) and link registers (W) in a three-tier system

Number of link relay (B) and link register (W) points that can be used in a system is as follows:

- MELSECNET mode B/W0 to B/W3FF (1024 points)
- MELSECNET II mode B/W0 to B/WFFF (4096 points)
- MELSECNET II composite modeB/W0 to B/WFFF (4096 points)

In a system with two or more master stations for the third tier as illustrated in Fig 5.8, when the link relay and link register exceeds the above-mentioned points, they can be expanded. In this case, the communication range that can be used is different from the communication range that can be used without expanding the range.

This section describes the link relay (B) and link register (W) range that can be expanded and the device range that can be used for data communication.

- (1) Using expanded link relays (B) and link registers (W)
 - (a) Use the master station for the third tier to turn ON special relays M9208 and M9209 when expanding to use link relays and link registers.
 (It is not allowed to turn ON either one of M9208 or M9209.)

M9208	 Set whether to send B/W data controlled by the second-tier master station to local stations (sub-slave stations) in the third tier or not. ONNot send OFFSend
M9209	 This turns ON when B/W data consistency is not checked between the second and third tiers. ONLink parameters for the second and third tiers are not checked. OFFLink parameters for the second and third tiers are checked.

(b) Use the SET instruction to turn ON special relays M9208 and M9209 as illustrated below. Once turned ON, do not turn these special relays ON or OFF during control.

]						
P	M9036	[SET	M9208	<u>}</u> {	
			SET	M9209]]	
I					۱ ۱	

- (2) Expanded link relay (B) and link register (W) range
 - The device range that is not assigned to the link parameters for the second tier can be assigned to the link parameters for the third tier (third tier-1, third tier-2, ... third tier-n). For this assignment, the same range can be assigned to different data link. For example, when 512 points of B/W0 to 1FF are used for the second tier as illustrated in Fig 5.8, the device range of B/W200 to 3FF can be redundantly assigned to the third-tier-1 link and also to the third-tier-2 link.



Fig 5.8 Expanded use example of link relay (B) and link register (W)

POINT

Since the L3/m station has not been set for expanded use of B/W, its communication range is as described in Section 5.3.1.

- (3) Data communication range with expanded link relays (B) and link registers (W)
 - (a) Link relays (B) and link registers (W) used by the two-tier link cannot be used for sending data to the local stations connected to any of the third-tier links.
 They can be used to send data to the master stations (L1/m, L2/m,, Ln/m) for the third tiers. This is because these master stations execute communication with the master station for the second tier as local stations for the second tier.
 - (b) The devices in the range set by the link parameters for the master station for the third tier cannot be used for sending data to the master station for the second tier and the local stations for the two-tier link.

The master station for the third tier can only receive data from the local stations connected to the host station.

For example, L1/m can only receive data from ℓ 1 and ℓ 2.

The expanded link relay (B) and link register (W) device range that can be used for communication in the system configuration in Fig 5.8 is shown in Table 5.3.

Link parameter	Sending						Rec	eive (o	data rea	ad) rai	nge	
setting station	(data write) station	Device range	М	L1/m	L2/m	L3/m	ℓ 1	ℓ2	ℓ'1	ℓ'2	Remarks	
	М	B/W0 to 7F	0	0	0	0					Only the stations connected to the second tion $(M + 1/m + 2/m + 3/m)$	
Master station	L1/m	B/W80 to FF	0	0	0	0					can be read. Reading of data from	
(M) for the second tier	L2/m	B/W100 to 17F	0	0	0	0					the local stations (ℓ 1, ℓ 2, ℓ '1,	
	L3/m	B/W180 to 1FF	0	0	0	0					ℓ '2) in the third-tier link is not possible.	
	L1/m	B/W200 to 2FF		0			0	0			Only the stations connected to the	
Master station (L1/m) for the	l 1	B/W300 to 37F		0			0	0			third tier - 1 (L1/m, ℓ 1, ℓ 2) can be read. Reading of the other stations	
third-tier-1	ℓ 2	B/W380 to 3FF		0			0	0			(M, L2/m, L3/m, ℓ '1, ℓ '2) is not possible.	
	L2/m	B/W200 to 27F			0				0	0	Only the stations connected to the	
Master station (L2/m) for the	ℓ 1	B/W280 to 37F			0				0	0	third tier - 2 (L2/m, ℓ '1, ℓ '2) can be read. Reading of the other	
third-tier-2	<i>ℓ</i> 2	B/W380 to 3FF			0				0	0	stations (M, L1/m, L3/m, ℓ 1, ℓ 2) is not possible.	

Table 5.3 Communication range when the link relay (B) and link register (W) ranges are expanded

POINT

Distinguish between the device range to communicate with the stations in the twotier link and the device range to communicate with the stations in the three-tier link and write a program.

As for L1/m station in Table 5.3, for example, use the device range B/W80 to FF for communication with the stations (M, L2/m, L3/m) connected to the two-tier link. Use device range B/W200 to 2FF for communication with the stations (ℓ 1, ℓ 2) connected to the three-tier-1 link.

MELSECNET mode		MELSECNET			MELSECNET/B		
Operating Mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	
Applicability		0	0		0	0	

MELSEC-A

5.3.8 MELSECNET II mode and MELSECNET II composite mode

When a MELSECNET II mode-compatible link module is used as the master station, the operation mode for the MELSECNET data link system can be selected in the MELSECNET II mode or the MELSECNET II composite mode depending on the system configuration.

(1) When the MELSECNET II mode is used, if a MELSECNET mode-compatible link module is used as a slave station, communication with this station will stop.

(1) MELSECNET II mode

Select the MELSECNET II mode only when modules for data link are all MELSECNET II mode-compatible link modules. Remote I/O stations cannot be connected. When connecting a remote I/O station, use the MELSECNET II composite mode.

(a) As for link relay (B) and link register (W), data link is possible up to 4096 points in the range of B/W0 to FFF.

- (b) The maximum number of link points per master station or local station is 2048 bytes. The MELSECNET II mode has the first half link parameter and the second half link parameter. Since up to 1024 bytes can be assigned, up to 2048 bytes can be used for data link. (Up to 1024 bytes can be used for the MELSECNET mode.)
- (c) Data link is possible, using only a first half link parameter.
 In this case, data sent from other stations can be received at the same time because data send/receive processing is executed at a time as when using the first half link parameter.
 If the maximum number of link points per station is 1024 bytes or less, perform data link with the first half link parameter only. This will reduce the handshake processing of sequence programs. (Refer to Section 9.1.)
- (d) The station for which only the first or second half link parameter is assigned can read all of the data assigned to the first and second half link parameters. In the link parameter setting illustrated in Fig 5.9, although only the first half link parameter is assigned for L3 station, it can read devices that are assigned to the second half link parameters used by other station.



Fig 5.9 System using MELSECNET ${\rm I\!I}$ mode

(2) MELSECNET II composite mode

In the MELSECNET II composite mode, MELSECNET mode-compatible local stations and remote I/O stations can be connected to the master station (MELSECNET II mode-compatible data link module).

By selecting the MELSECNET II composite mode, data link modules used in the MELSECNET mode (previous MELSECNET data link system) and those compatible with the MELSECNET II mode can be used within the same data link.

- (a) As for link relays (B) and link registers (W), data link is possible up to 4096 points in the range of B/W0 to FFF.
- (b) The maximum number of link points per master station or MELSECNET II mode-compatible local station is 2048 bytes. The MELSECNET II mode has the first half link parameter and the second half link parameter. Since up to 1024 bytes can be assigned, up to 2048 bytes can be used for data link.

In this regard, however, the MELSECNET mode-compatible local station is up to 1024 bytes which is the same as that of MELSECNET mode and the MELSECNET mode-compatible remote I/O station is up to 512 bytes.

POINT

A MELSECNET mode-compatible local station can only read the range of link relays (B) and link registers (W) assigned to the first half link parameter. Devices assigned to the second half link parameter cannot be read.

L1 station in Fig 5.10, for example, cannot read the devices in the range of B/ W300 to 4FF because it is used for a MELSECNET mode-compatible data link module.

MELSECNET II mode-compatible master stations and local stations can read the range assigned to both the first and second half link parameters.



Fig 5.10 System using MELSECNET ${\rm I\!I}$ composite mode

REMARK

- (1) When a MELSECNET mode-compatible link module is used for the master station, the system operates in the MELSECNET mode even if a MELSECNET II mode-compatible link module is connected to a local station.
- (2) When the MELSECNET mode link parameter is set using a MELSECNET II mode-compatible link module, the system operates in the MELSECNET mode.
- (3) When only the first half link parameter is set using the MELSECNET II composite mode, the system operates in the MELSECNET mode.

	1						
MELSECNET mode		MELSECNET			MELSECNET/B		
Operating Mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	
Applicability	Ó	0	0				

MELSEC-A

5.4 Optical Fiber Cable Specifications

The following shows the optical fiber cable specifications used for optical data link of the MELSECNET.

Special skills and tools are required for connecting optical fiber cables to connectors. In addition, dedicated connector plugs are also needed. When purchasing any of the connector plugs, please consult your local Mitsubishi Electric System & Service Co., Ltd.

5.4.1 SI/GI-type optical fiber cable

The Table 5.4 shows specifications of SI/GI-type optical fiber cable.

Item	SI (Multi particulate glass)	H-PCF (Plastic clad)	GI (Quartz glass)	
Distance between stations	1km	1km	2km	
Transmission loss	12dB/km	6dB/km	3dB/km	
Core diameter	200 <i>µ</i> m	200μ m	50 <i>µ</i> m	
Clad diameter	220 <i>µ</i> m	250 <i>µ</i> m	125 <i>µ</i> m	
Primary membrane	250 <i>µ</i> m	-	-	
Applicable connector	F06/F08 or equivalent (Compliant with JIS C5975/5977)			

Table 5.4 SI/GI-type optical fiber cable

REMARK

- (1) The optical cables include the following types.
 - A type: Cable for inside control panel connection.
 - B type: Cable for connection between outside control panels.

C type: Cable for outdoor connections.

D type: Cable for reinforced outdoor connections.

There are special cables available for moveable applications and resistance to heat.

Contact your Mitsubishi Electric System & Service Co., Ltd. for details.

MELSECNET mode	MELSECNET			MELSECNET/B		
Operating Mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode
Applicability	0	0	0			

5.5 Coaxial Cable Specifications

This section describes the specifications of coaxial cable used for the coaxial data link. The high-frequency coaxial cables "3C-2V" and "5C-2V" (conforming to JIS C 3501) are used as coaxial cables.

5.5.1 Coaxial cable specifications

The specifications of the coaxial cable are shown in Table 5.5.

Select coaxial cables that meet the operating ambient temperature (0 to $55^{\circ}C$) shown in the general specifications of the programmable controller.

Item	3C-2V	5C-2V		
Structure	Internal condcuctive materisl material	External conductive External material sheath		
Cable diameter	5.4mm (0.21inch)	7.4mm (0.29inch)		
Allowable bend radius	22mm (0.87inch) or more	30mm (1.18inch) or more		
Internal conductor diameter	0.5mm (0.02inch) (Annealed copper wire)	0.8mm (0.03inch) (Annealed copper wire)		
Insulator diameter	3.1mm (0.12inch) (Polyethylene)	4.9mm (0.19inch) (Polyethylene)		
External conductor diameter	3.8mm (0.15inch) (Single annealed copper wire mesh)	5.6mm (0.22inch) (Single annealed copper wire mesh)		
Applicable connector plug	Connector plug for 3C-2V (BNC-P-3-Ni is recommended.)	Connector plug for 5C-2V (BNC-P-5 or BNC-P-5DV-SA(01) is recommended.)		

Table 5.5 Coaxial cable

REMARK

Consult nearest Mitsubishi representative with connector plug.

	1			r			
MELSECNET mode		MELSECNET			MELSECNET/B		
Operating Mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	
Applicability	0	0	0				

MELSEC-A

5.5.2 Connector for the coaxial cable

The following explains the structure and connecting procedures of the BNC connector (connector plug for the coaxial cable).

Structure of the BNC connector and the coaxial cable
 Fig 5.11 shows the structure of the BNC connector and the coaxial cable.



Fig 5.11 Structure of the BNC connector and the coaxial cable

- (2) Procedure for connecting the BNC connector and the coaxial cable The following describes the procedure for connecting the BNC connector and the coaxial cable.
 - (a) Remove the outer sheath of the coaxial cable to specified dimensions as shown below. Use caution not to damage the external conductor.



Applicable cable	А
3C-2V	15mm
5C-2V	10mm

(b) Slip a nut, a washer, a gasket, and a clamp on the coaxial cable, and loosen the external conductor as shown below.



(From the previous page)

(c) Cut the external conductor, insulator, and internal conductor to specified dimensions as shown right.

As for the external conductor, cut it as the same dimension as that of taper part and smooth down it on the clamp.



Applicable cable	В	С
3C-2V	6mm	3mm
5C-2V	7mm	5mm

(d) Solder the contact to the tip of the internal conductor.



(e) Insert the contact assembly to the plug shell, and engage the plug shell with the nut.



POINT

- (1) Use caution as follows when soldering the contact to the internal conductor.(a) Solder must not be pumped up at soldering part.
 - (b) The tail end of the contact and the cut end of the insulator must contact close to each other. Also, the contact must not cut in the insulator.
 - (c) Apply solder quickly so that the insulator may not deform.
- (2) Before connecting or disconnecting the coaxial cable connector, be sure to touch a grounded metal object to discharge the static electricity from the human body.

Failure to do so may harm the module.

5.6 Twisted Pair Cable Specifications

This section explains the twisted pair cable for the MELSECNET/B data link system.

Model name	KNPEV-SB 0.5SQ × 1P
Cable type	Shielded twisted pair cable
Number of cores	2
Conductive resistance (20 $^{\circ}$ C)	39.4 Ω /km or lower
Insulation resistance (20 $^{\circ}$ C)	10 M Ω - km or higher
Dielectric withstand voltage (V-min)	1000 V AC for one minute
Electrostatic capacity (1kHz)	70 nF/km or less (on average)
Characteristic impedance (100kHz)	$110\pm10\Omega$
Cross section	Blue White

Table 5.6 Shielded twisted pair cable

MEMO

MELSECNET mode	MELSECNET			MELSECNET/B		
Operating Mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode
Applicability	0	0	0	0	0	0

MELSEC-A

6 LINK DATA SEND/RECEIVE PROCESSING AND PROCESSING TIME

This chapter describes how link data are sent or received in the data link system and it's processing time.

- 6.1 Link Data Communication Processing
- 6.1.1 Communication processing outline

In the data link system, the link data set at the link parameters of the master station is communicated repeatedly.

- (1) Link module configuration
 - (a) A link module used in a master station or local station has the link data storage area which communicates the link data to other stations and the data memory storage area used for processing of the host station.
 - (b) A link module used in a remote I/O station has the link data storage area which stores the link data to be communicated to the master station.
- (2) Link data communication

Link data communication includes a link scan and a link refresh.

- (a) A link scan is communications of link data between link modules (between link data storage areas).
- (b) A link refresh is communications of data link within a link module.
 - Link refresh for a master station or local station is executed when link data is communicated between the link data storage area and the data memory storage area.
 - Link refresh for a remote I/O station is executed when link data is communicated between the link data storage area and the I/O module or special function module.



Fig 6.1 Link data communications

MELSECNET mode	MELSECNET			MELSECNET/B		
Operating Mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode
Applicability	0	0	0	0	0	0

MELSEC-A

6.1.2 Link refresh execution timing

The link refresh executing timing can be divided into the following three types.

- Immediately after the completion of a link scan A link refresh will be executed when a link scan is completed. The sequence program will be interrupted.
- (2) After the execution of an END instruction in sequence program A link refresh will be executed only after an END instruction has been executed in a sequence program.
- (3) At setting timeA link refresh will be executed at the time defined by the user beforehand.

For (1) or (3), data containing both old and new data fragments may be transferred because a link refresh can be executed during sequence program operation.

To prevent the above, use a handshake between the sending station and the receiving station. (Refer to Section 9.1.)

The Table 6.1 shows classification of link modules by link refresh timing.

		Link refresh timing	
Link module name	After the completion	After the execution of	At preset intervals
	of a link scan	an END instruction	At preset litter vals
A1SHCPU			
+ A1SJ71AP21/R21			
A2SHCPU			
+ A1SJ71AP21/R21	0	O*1	-
A0J2HCPUP21/R21			
AnNCPUP21(S3)/R21			
A2CCPUP21/R21			
A2USCPU(S1)			
+ A1SJ71AP21/R21			
A2USHCPU-S1			
+ A1SJ71AP21/R21			
Q2ASCPU(S1)			
+ A1SJ71AP21/R21			
Q2ASHCPU(S1)			
+ A1SJ71AP21/R21			
AnACPUP21(S3)/R21			
AnUCPU	-	0	-
+ AJ71AP21(S3)/R21			
Q2ACPU(S1)			
+ AJ71AP21(S3)/R21			
Q3ACPU			
+ AJ71AP21(S3)/R21			
Q4ACPU			
+ AJ71AP21(S3)/R21			
Q02CPU-A			
+A1SJ71AP21/R21			

 Table 6.1 Link module classification by link refresh timing

	Link refresh timing				
Link module name	After the completion of a link scan	After the execution of an END instruction	At preset intervals		
Q02HCPU-A					
+A1SJ71AP21/R21		0	-		
Q06HCPU-A	-				
+A1SJ71AP21/R21					
A0J2P25(S3)/R25	0				
AJ72P25(S3)/R25	0	-	-		

Table 6.1 Link module classification by link refresh timing (Continued)

O: Applicable

P	
POINT	
(1) *1	The link refresh will be executed only after the execution of an END instruction in a sequence program when the programs steps between step 0 and the END instruction are set by the sequence program as a link-refresh inhibited zone. The link-refresh inhibited zone can be set by EI/DI instruction. For details of EI/DI instruction, refer to the Type ACPU/QCPU-A (A Mode) Programming Manual (Common Instructions).
(2) The link ref	resh timing of the following modules is the same as that of the CPU
module.	
• A1SJ71AT	21B • AJ71AT21B
• A1SJ71AF	'21/R21 • AJ71AP21/R21
• AJ71AP21	-53 • AJ71P22/R22
(3) A link scan	does not influence the processing time (for example, scan time for the
ACPU) of a	master station or local station, since link scan is executed in parallel
with master	and local station processing.
(4) The process the link refre	sing time (scan time) of the master and local stations is increased by esh time.
The followir	ng chart shows an example of executing link refresh processing after
END proces	ssing.
	Link refresh time
	Sequence program
Soquence	END 0 END 0
execution	
	Scan time
(5) Link data co	ommunication is possible in any of the following ACPU states: RUN,
STOP, PAU	SE, STEP-RUN.

MELSECNET mode	MELSECNET			MELSECNET/B		
Operating Mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode
Applicability	0	0	0	0	0	0

MELSEC-A

6.1.3 Link data during a communication error

The following describes how the link data used for data link is processed when a communication error occurs. (The same applies to a communication error of the station connected to the bypass switch.)

- If a master or local station has the communication error, link data just before the communication error will be held in the station.
- If the communication error occurs on a remote I/O station, all of the output modules mounted on the station turns OFF.
- Normally operating station will hold link data received before occurrence of the communication error.

The following describes how the link data of normally operating station and abnormally operating station is handled, dividing into the master station, local station, and remote I/O station.

- (1) When a communication error occurs at a master station
 - (a) Communications with all slave stations are stopped.
 - (b) The master station either turns ON M9210 or stores "5" in D9204. The data received from a slave station is maintained in a state just before the occurrence of the communication error. The link special relays M9224 to M9239 and link special registers D9202 to D9242 of the master station, the data is maintained in a state just before the occurrence of the communication error.
 - (c) The local stations turn ON M9250 and M9251.At devices in the range of the data link, data is maintained in a state just before the occurrence of the communication error.
 - (d) At remote I/O stations, all of the points of the output modules and special function modules installed to the host station are turned OFF.
- (2) When a communication error occurs at a local station
 - (a) The local station at which the communication error occurred, either M9211 or M9250 and M9251 are turned ON.
 Communication with other normally operating stations continues.
 The link special relays (M) M9240 to M9255 (except M9250 and M9251) and link special registers (D) D9243 to D9255 of the local station hold data in the state just before the communication error.
 - (b) The master station can recognize the station number at which the communication error has occurred by reading the data in M9237 and D9228 to D9231.

At the link devices which receive data from the local station at which the communication error has occurred, the data is maintained in a state just before the occurrence of communication error.

(c) Normally operating local stations can recognize the station number at which the communication error has occurred by reading the data in M9255 and D9252 to D9255.

At the link devices which receive data from the station at which the communication error has occurred, the data is maintained in a state just before the occurrence of communication error.

- (d) All remote I/O stations operate normally.
- (3) When a communication error occurs at a remote I/O station
 - (a) At the remote I/O station, all of the output points of the output modules and special function modules installed to the host station are turned OFF.
 - (b) The master station can recognize the station number at which the communication error has occurred by reading the data in M9231 and D9228 to D9231.

At the link devices which receive data from the remote I/O station at which the communication error has occurred, the data is maintained in a state just before the occurrence of communication error.

(c) All local stations operate normally.

6.2 Transmission Delay Time in Two-Tier System

Calculate the transmission delay time in MELSECNET and MELSECNET/B data link system using the following times for the formulas given in Section 6.3.

- Link refresh time for the master, local, and remote I/O stations
- Link scan time
- · Scan time for the master and local stations
- (1) Link refresh time for the master, local, and remote I/O stations (α 1 to α 3) Link refresh time is time required for link refresh. (Refer to Section 6.1.1.) Calculate the link refresh time using the formulas given in Section 6.2.2.
- (2) Link scan time (LS) Link scan time is time required for link scan. (Refer to Section 6.1.1.) Calculate the link scan time using the formulas given in Section 6.2.2. When data link is being executed, link scan time can be checked by monitoring link or data link special registers (D9207 to D9209) with a peripheral device.
- (3) Scan time for the master and local stations (M, L)
 Scan time is defined as the time required for operating a sequence program from step 0 to the next step 0.
 It can be checked by monitoring ladder or special registers (D9017 to D9019) with a peripheral device.

REMARK

1) Refer to Section 9.3.1 for details of the data link special registers.

2) Refer to the Type ACPU/QCPU-A (A Mode) Programming Manual (Common Instructions) for details of the special registers (D9017 to D9019)

MELSECNET mode	MELSECNET			MELSECNET/B		
Operating Mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode
Applicability	0	0	0	0	0	0

MELSEC-A

6.2.1 Transmission delay time in a two-tier system

This section shows the maximum transmission delay times of a data link in a link data system.

(1) The following table shows the transmission delay times when the CPU of the type that immediately makes a link refresh after link scan is used.

Table 6.2 Maximum transmission delay time

		L < LS < M , LS < L < M	LS < M < L
Master station	Link relay (B) Link register(W) Output (Y)	$M + \alpha_1 + LS + \alpha_2 + L$ $M = e^{1}$ LS LS LS LS LS LS LS LS	$M + \alpha_1 + (L + \alpha_2) \times 2$ $M + \alpha_1 + (L + \alpha_2) \times 2$ $M = \underline{\alpha_1} + \underline{\alpha_2} + \underline{\alpha_1} + \underline{\alpha_2} + \underline{\alpha_1} + \underline{\alpha_2} + \underline{\alpha_2} + \underline{\alpha_1} + \underline{\alpha_2} +$
to local station	LRDP or LWTP instruction	M x 4 + α_1 x 4 LRDP executed LRDP completed M $\underbrace{c_{1}}_{c_{1}} \underbrace{c_{1}}_{c_{1}} \underbrace{c_{1}}_{c_{2}} \underbrace{c_{1}} \underbrace{c_{1}} \underbrace{c_{1}} \underbrace{c_{1}} \underbrace{c_{1}} \underbrace{c_{1}} \underbrace{c_{1}} \underbrace{c_{1}}$	$(M + \alpha_1) \times 2 + (M + \alpha_1) \times \left(\frac{(L + \alpha_2) \times 2}{M + \alpha_1}\right)$ Integer (decimal fraction rounded up) CRDP executed $M \text{Stel to Extore to E to $
Local station to master station	Link relay (B) Link register (W) Input (X)	$M \times 4 + \alpha_1 \times 3 - LS - \alpha_2$ $M \xrightarrow{Signature{}}_{LS} \underbrace{I_{S}}_{LS} \underbrace{I_{S}} \underbrace{I_{S}}_{LS} \underbrace{I_{S}}_{LS} \underbrace{I_{S}} \underbrace{I_{S}}_{LS} \underbrace$	$M \times 2 + \alpha_1 + (L + \alpha_2) \times 2 - LS$ $M \xrightarrow{\alpha_1} \qquad \qquad$
Local station to local station	Link relay (B) Link register (W)	$M \times 2 + \alpha_1 \times 2 - \alpha_2 + \alpha_2' + L2$ $M = 0 e_1 e_2 e_1 e_2 e_1 e_2 e_1 e_2 e_1 e_2 e_1 e_2 e_1 e_2 e_1 e_2 e_2 e_1 e_2 $	$(L2 + \alpha_2) \times 2 + (L1 + \alpha_2') \times 2$ $(L2 + \alpha_2) \times 2 + (L1 + \alpha_2') \times 2$ $(L1 + \alpha_2') \times 2$ $(L2 + \alpha_2) \times 2 + (L1 + \alpha_2') \times 2$ $(L2 + \alpha_2) \times 2 + (L1 + \alpha_2') \times 2$ $(L2 + \alpha_2) \times 2 + (L1 + \alpha_2') \times 2$ $(L2 + \alpha_2) \times 2 + (L1 + \alpha_2') \times 2$ $(L2 + \alpha_2) \times 2 + (L1 + \alpha_2') \times 2$ $(L2 + \alpha_2) \times 2 + (L1 + \alpha_2') \times 2$ $(L2 + \alpha_2) \times 2 + (L1 + \alpha_2') \times 2$ $(L2 + \alpha_2) \times 2 + (L1 + \alpha_2') \times 2$ $(L2 + \alpha_2) \times 2 + (L1 + \alpha_2') \times 2$ $(L2 + \alpha_2) \times 2 + (L1 + \alpha_2') \times 2$ $(L2 + \alpha_2) \times 2 + (L1 + \alpha_2') \times 2$ $(L2 + \alpha_2) \times 2 + (L1 + \alpha_2') \times 2$ $(L2 + \alpha_2) \times 2 + (L1 + \alpha_2') \times 2$ $(L2 + \alpha_2) \times 2 + (L1 + \alpha_2') \times 2$ $(L2 + \alpha_2) \times 2 + (L1 + \alpha_2') \times 2$ $(L2 + \alpha_2) \times 2 + (L1 + \alpha_2') \times 2$ $(L2 + \alpha_2) \times 2 + (L1 + \alpha_2') \times 2$ $(L2 + \alpha_2) \times 2 + (L1 + \alpha_2') \times 2$ $(L2 + \alpha_2) \times 2 + (L1 + \alpha_2') \times 2$ $(L2 + \alpha_2) \times $
Master station to remote	Output (Y)	M + α ₁ + LS + α ₃	
I/O station	RFRP or RTOP instruction	Μ × 6 + α ₁ × 6	
Remote I/O station to master station	Input (X)	M x 4 + α ₁ x 3 - α ₃ - LS	

M : Sequence program scan time for the master station

 α 1: Link refresh time for the master station α 2: Link refresh time for a local station

L : Sequence program scan time for a local station LS : Data communication (send/receive) time

 α 3: I/O refresh time for a remote I/O station

M < L < LS , L < M < LS	M < LS < L
$LS \times 2 + \alpha_1 + \alpha_2 + L$ $M = \frac{\alpha_1}{1 + \alpha_2 + L}$ $LS = \frac{\alpha_1}{1 + \alpha_2 + L}$	$LS + \alpha_1 + L + \alpha_2 + \left(\frac{L + \alpha_2 - \alpha_1}{LS + \alpha_1}\right) \times (LS + \alpha_1)$ $M \xrightarrow{\text{ore}} \xrightarrow{\text{ore}} \underbrace{\text{tore}}_{\text{tore}} \underbrace{\text{tore}} \text{tor$
LS x 4 + α_1 x 4 LRDP executed LRDP completed M o c	$(LS + \alpha_1) \times 2 + \left(\frac{L + \alpha_2}{LS + \alpha_1}\right) \times (LS + \alpha_1) + M$ $LRDP \text{ executed } \text{ integer (decimal fraction rounded up)} Completed$ $M Step - to to - to to - to to - to to - to -$
$M + \alpha_1 \times 3 + LS \times 2 - \alpha_2$ $M = \alpha_1 + \alpha_2 + \alpha_2$ $M = \alpha_1 + \alpha_2 + \alpha_2 + \alpha_2$ $M = \alpha_1 + \alpha_2 +$	$(LS + \alpha_1) \times 2 + M + \left(\frac{L + \alpha_2 - \alpha_1}{LS + \alpha_1}\right) \times (LS + \alpha_1) + \alpha_1$ $(LS + \alpha_1) + \alpha_1$ $M = \frac{\alpha_1}{1 + 1} + \frac{\alpha_1}{1 + 1$
$LS \times 2 + \alpha_1 \times 2 + LS + \alpha'_2 - \alpha_2$ (1) M (2) LS (2)	$LSx2+\alpha_1 + \left(\frac{LS+\alpha_2 - \alpha_1}{LS+\alpha_1}\right) \times (LS+\alpha_1) + \left(\frac{L1+\alpha'_2 - \alpha_1}{LS+\alpha_1}\right) \times (LS+\alpha_1) + \alpha_2 + L1$ $\xrightarrow{\alpha_1} \qquad \qquad$
$LS \times 2 + \alpha_1 + \alpha_3$ $M = \frac{\alpha_1}{c_1} + \frac{\alpha_2}{c_1}$ $R = \frac{\alpha_1}{c_2} + \frac{\alpha_1}{c_1} + \frac{\alpha_2}{c_2}$	
M + LS x 5 + α_1 x 6 M = LS x 5 + α_1 x 6 R = α_1 R = α_2 R = α_1 R = α_2 R = α_1 R = α_2 R = α_1 R = α_2 R = α_2 R = α_1 R = α_2 R = α_2 R = α_1 R = α_2 R = α_2 R = α_2 R = α_2 R = α_2 R = α_3 R = α	
$M + LS \times 2 + \alpha_1 \times 3 - \alpha_3$ $M = \begin{bmatrix} \alpha_1 & \alpha_2 & \alpha_3 \\ 0 & \epsilon & \alpha_1 & \epsilon \\ 0 & \epsilon & \alpha_2 & \alpha_3 \\ 0 & \epsilon & \alpha_3 & \alpha_4 \\ 0 & \epsilon & \alpha_4 & \alpha_5 & \alpha_6 \\ 0 & \epsilon & \alpha_4 & \alpha_5 & \alpha_6 \\ 0 & \epsilon & \alpha_4 & \alpha_5 & \alpha_6 & \alpha_6 \\ 0 & \epsilon & \alpha_4 & \alpha_5 & \alpha_6 & \alpha_6 & \alpha_6 \\ 0 & \epsilon & \alpha_4 & \alpha_5 & \alpha_6 & \alpha_6 & \alpha_6 & \alpha_6 \\ 0 & \epsilon & \alpha_4 & \alpha_5 & \alpha_6 & \alpha_6 & \alpha_6 & \alpha_6 & \alpha_6 \\ 0 & \epsilon & \alpha_5 & \alpha_6 & \alpha$	

Operating Mode MELSECNET MELSECNET I MELSECNET I MELSECNET I MELSECNET I mode mode mode mode mode mode mode mode	MELSECNET mode		MELSECNET		MELSECNET/B		
Applicability	Operating Mode	MELSECNET MELSECNET I mode		MELSECNET II composite mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode
	Applicability	0	0	0	0	0	0

MELSEC-A

(2) The following shows the transmission delay times when the CPU of the type that makes a link refresh after END processing is used.

Table 6.3 Maximum transmission delay time

		L < LS < M, LS < L < M	LS < M< L	
Master station	Link relay (B) Link register (W) Output (Y)	M+ α 1+LS+L×2+ α 2 M LS L	$M + \alpha 1 + L \times 3 + \alpha 2 \times 2$ $M = H^{\alpha} + H^{\alpha$	
station	LRDP or LWTP instruction	$M \times 5 + \alpha 1 \times 5$ $M \qquad \qquad$	$(M + \alpha 1) \times 3 + (L + \alpha 2) \times 3$	
Local station to master station	Link relay (B) Link register (W) Input (Y)	$M \times 4 + \alpha 1 \times 3 + L$ $M = 1 \times 3 + L$ $LS = 1 \times 1 \times 3 + L$	$M \times 3 + \alpha 1 \times 2 + L + \alpha 2$ $M \downarrow I \downarrow $	
Local station to local station	Link relay (B) Link register (W)	$M \times 2 + \alpha 1 \times 2 + L1 + LS + L2 \times 2 + \alpha 2'$	$M+\alpha 1+L1+\alpha 2+L2\times 3+\alpha 2'\times 2$ M LS $L1$ $L1$ $L2$	
Master station to remote	Output (Y)	M + α ₁ + LS + α ₃ Μ <u>σ</u> LS R		
I/O station	RFRP or RTOP instruction	M × 6 + α ₁ × 6 M σ LS R		
Remote I/O station to master station	Input (X)	M × 4 + α ₁ × 3 - α ₃ - LS M <u>σ</u> LS R		

M : Sequence program scan time for the master station

 $\alpha\,$ 2: Link refresh time for a local station

L : Sequence program scan time for a local station LS : Data communication (send/receive) time

 $\alpha\,$ 3: I/O refresh time for a remote I/O station

M < L < LS, L < M < LS	M < LS < L
$M + \alpha 1 + LS \times 2 + L \times 2 + \alpha 2$ $M \qquad \qquad$	$M+\alpha 1+LS+L\times 3+\alpha 2\times 2$ $M \xrightarrow{0 \in 0 \in F} \underbrace{0 \in 0 \in F}_{0 \in 0 \in F} \underbrace{0 \in 0 \in F}_{0 \in 0 \in F} \underbrace{0 \in 0 \in F}_{0 \in 0 \in F} \underbrace{0 \in 0 \in F}_{0 \in 0 \in F} \underbrace{0 \in F}_{0 \in F} \underbrace{0 \in F} \underbrace{0 \in F}_{0$
$(M + \alpha 1) \times 5 + LS \times 5$	$(M + \alpha 1) \times 3 + LS \times 3 + (L + \alpha 2) \times 3$ $M \xrightarrow{0 \text{ EDD } 3 \text{ or ED }$
$M \times 4 + \alpha 1 \times 3 + LS \times 3 + L$ $M \xrightarrow{0} E 0 E + 0 E 0 E + 0 E 0 E + 0 E 0 E + 0 E 0 E$	$M \times 3 + \alpha 1 \times 2 + LS \times 2 + L + \alpha 2$ $M 0 \in 0 \in I \cap E \cap$
$(M+\alpha 1) \times 2 + LS \times 3 + L1 + L2 \times 2 + \alpha 2'$ $M c elo elo elo elo elo elo elo elo elo elo$	$M+\alpha 1+LS+L1+\alpha 2+L2\times 3+\alpha 2'\times 2$ M LS $L1$ $L2$ M $M+\alpha 1+LS+L1+\alpha 2+L2\times 3+\alpha 2'\times 2$
$M + \alpha 1 + LS \times 2 + \alpha 3$ $M = 4 + \alpha 1 + LS \times 2 + \alpha 3$ $LS = 4 + \alpha 1 + LS \times 2 + \alpha 3$ $R = 4 + \alpha 1 + LS \times 2 + \alpha 3$ $R = 4 + \alpha 1 + LS \times 2 + \alpha 3$	
$(M + \alpha 1) \times 6 + LS \times 6$ M $0 = E 0 = E + 0 $	
$M \times 3 + \alpha 1 \times 2 + LS \times 2$ $M = \begin{bmatrix} 0 & E & 0 & E & 0 & E & 0 & E \\ LS & & & & \\ R & & & & & \\ R & & & & & \\ R & & & &$	

MELSECNET mode		MELSECNET		MELSECNET/B		
Operating Mode	MELSECNET MELSECNET II mode		MELSECNET II composite mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode
Applicability	0	0	0			

MELSEC-A

6.2.2 Link refresh time

The following describes the method used for calculating the link refresh processing time.

The symbols used for calculating the link refresh processing time are listed below:

B :Total number of link relays (B) used in all stations

W :Total number of link registers (W) used in all stations

X0:Total number of link inputs (X) assigned to the master station

Yo:Total number of link outputs (Y) assigned to the master station

X1:Number of link inputs (X) used in the corresponding station

 $\ensuremath{\mathsf{Y1}}$:Number of link outputs (Y) used in the corresponding station

 α 1 to $\ \alpha$ 3: Link refresh time

KM1, KL1 , KR1: Constants

KM2, KL2, KR2: Bit device constants

KM3, KL3: Word device constants

REMARK

The refresh range of the AnUCPU, QnACPU, A2US(H)CPU(S1), Q2AS(H)CPU(S1), or QCPU-A is determined by the network refresh parameter and the link parameter. (Refer to Section 7.3.)

When a MELSECNET data link system is used

(1) Master station

Use the following formula to calculate the link refresh time α 1 required for the master station.

In the MELSECNET II mode or MELSECNET II composite mode, the number of B/W points is all of the points that are set with the first and second half link parameters.

 $\alpha_1 = k_{M1} + k_{M2} \times \frac{B + X_0 + Y_0}{2048} + k_{M3} \times \frac{W}{1024} \text{ [ms]}$

MELSEC-A

		Model	Km1	Км2	Кмз
	A0J2HCPUP21/R21	-	1.59	1.1	5.02
	AnNCPUP21/R21	-	0.8	1.0	4.1
	A2ACPUP21(S1)/R21	-	0.26	0.18	1.45
	A3ACPUP21/R21	-	0.20	0.14	1.09
	-	AnNCPUP21/R21 + AJ71AP21(S3)/R21	0.8	1.2	6.4
	-	A2ACPU(S1) + AJ71AP21(S3)/R21	0.54	0.54	4.32
	-	A3ACPU + AJ71AP21(S3)/R21	0.48	0.52	4.16
	-	A2UCPU(S1) + AJ71AP21(S3)/R21	0.54	0.54	4.32
	-	A3UCPU + AJ71AP21(S3)/R21	0.48	0.52	4.16
	-	A4UCPU + AJ71AP21(S3)/R21	0.48	0.51	4.16
Master	-	A1SCPU + A1SJ71AP21/R21	1.52	1.53	6.57
station for	-	A1SJCPU-S3 + A1SJ71AP21/R21	1.14	1.53	6.68
the second	-	A2SCPU + A1SJ71AP21/R21	1.06	1.49	6.60
tier	-	A1SHCPU + A1SJ71AP21/R21	0.55	0.51	3.82
	-	A1SJHCPU + A1SJ71AP21/R21	0.55	0.51	3.82
	-	A2SHCPU + A1SJ71AP21/R21	0.56	0.58	4.20
	-	A2CCPUP21/R21	1.59	1.11	5.02
	-	A2ASCPU(S1) + A1SJ71AP21/R21	0.66	0.64	4.38
	-	A2USHCPU-S1 + A1SJ71AP21/R21	0.45	0.51	4.31
	-	Q2ASCPU(S1) + A1SJ71AP21/R21	0.95	0.51	4.53
	-	Q2ASHCPU(S1) + A1SJ71AP21/R21	0.51	0.53	4.18
	-	Q2ACPU(S1) + AJ71AP21(S3)/R21	0.97	0.51	4.56
	-	Q3ACPU + AJ71AP21(S3)/R21	0.81	0.61	4.37
	-	Q4ACPU + AJ71AP21(S3)/R21	0.43	0.82	4.54
	AnNCPUP21/R21	-	0.8	1.2	6.4
	A2ACPUP21(S1)/R21	-	0.54	0.54	4.32
	A3ACPUP21/R21	-	0.48	0.52	4.16
Master station for the third tier	-	AnNCPUP21/R21 + AJ71AP21(S3)/R21	0.8	1.2	6.4
	-	A2ACPU(S1) + AJ71AP21(S3)/R21	0.54	0.54	4.32
	-	A3ACPU + AJ71AP21(S3)/R21	0.48	0.52	4.16
	-	A2UCPU(S1) + AJ71AP21(S3)/R21	0.54	0.54	4.32
	-	A3UCPU + AJ71AP21(S3)/R21	0.48	0.52	4.16
	-	A4UCPU + AJ71AP21(S3)/R21	0.48	0.51	4.16
	-	A2ASCPU(S1) + A1SJ71AP21/R21	0.34	1.02	4.42
	-	A2USHCPU-S1 + A1SJ71AP21/R21	0.28	0.41	4.02
	-	Q2ASCPU(S1) + A1SJ71AP21/R21	0.43	1.28	5.30
	-	Q2ASHCPU(S1) + A1SJ71AP21/R21	0.28	0.83	4.17
	-	Q2ACPU(S1) + AJ71AP21(S3)/R21	0.50	1.12	5.26
	-	Q3ACPU + AJ71AP21(S3)/R21	0.39	1.00	5.83
	-	Q4ACPU + AJ71AP21(S3)/R21	0.45	0.82	3.89

POINT

Two types of link refresh (link refresh of a local station for the second tier and link refresh of a master station for the third tier) are executed for the master station for the third tier.

(2) Local stations

Use the following formula to calculate the link refresh time α_2 required for a local station.

When the link operates in the MELSECNET II mode or the MELSECNET II composite mode, the number of B/W points is all points set with the first half and second half link parameters.

$$\alpha_2 = K_{L1} + K_{L2} \times \frac{B + X_1 + Y_1}{2048} + K_{L3} \times \frac{W}{1024} [ms]$$

	Model	KL1	KL2	KL3
AnNCPUP21(S3)/R21	-	0.4	1.0	4.1
A2ACPUP21(S3)/R21(S1)	-	0.16	0.18	1.45
A3ACPUP21(S3)/R21	-	0.13	0.14	1.09
-	AnNCPUP21/R21 + AJ71AP21(S3)/R21	0.8	1.2	6.4
-	A2ACPU(S1) + AJ71AP21(S3)/R21	0.54	0.54	4.32
-	A3ACPU + AJ71AP21(S3)/R21	0.48	0.52	4.16
-	A2UCPU(S1) + AJ71AP21(S3)/R21	0.54	0.54	4.32
-	A3UCPU + AJ71AP21(S3)/R21	0.48	0.52	4.16
-	A4UCPU + AJ71AP21(S3)/R21	0.48	0.51	4.16
- A0J2HCPUP21/R21		1.00	1.20	5.05
- A1SCPU + A1SJ71AP21/R21		1.09	1.36	6.53
- A1SJCPU-S3 + A1SJ71AP21/R21		0.77	1.30	6.58
-	- A2SCPU + A1SJ71AP21/R21		1.44	6.54
-	- A1SHCPU + A1SJ71AP21/R21		0.59	3.86
-	A1SJHCPU + A1SJ71AP21/R21	0.30	0.59	3.86
-	A2SHCPU + A1SJ71AP21/R21	0.32	0.64	4.23
-	A2CCPUP21/R21	1.00	1.20	5.05
-	A2ASCPU(S1) + A1SJ71AP21/R21	0.47	0.64	4.41
-	A2USHCPU-S1 + A1SJ71AP21/R21	0.29	0.51	4.05
- Q2ASCPU(S1) + A1SJ71AP21/R21		0.78	0.59	4.51
-	Q2ASHCPU(S1) + A1SJ71AP21/R21	0.34	0.54	4.20
-	Q2ACPU(S1) + AJ71AP21(S3)/R21	0.80	0.55	4.53
-	Q3ACPU + AJ71AP21(S3)/R21	0.74	0.50	4.28
-	Q4ACPU + AJ71AP21(S3)/R21	0.31	0.68	4.36

(3) Remote I/O stations

Use the following formula to calculate the I/O refresh time α ₃ required for a remote I/ O station.

$$\alpha_3 = K_{R1} + K_{R2} \times \frac{X_1 + Y_1}{256}$$
[ms]

Model	KR1	Kr2
AJ72P25(S3)/R25	0.6	0.9

When a MELSECNET/B data link system is used

(1) Master station

Use the following formula to calculate the link refresh time α_1 required for a master station.

When the link operates in the MELSECNET II mode or the MELSECNET II composite mode, the number of B/W points is all of the points set with the first half and second half link parameters.

$$\alpha_1 = K_{M1} + K_{M2} \times \frac{B + X_0 + Y_0}{2048} + K_{M3} \times \frac{W}{1024} [ms]$$

	Мо	Kuu	Kuo	Kua	
	CPU module	Link module	rxm1	NM2	IXM3
	A1SCPU		1.34	1.15	6.47
	A1SJCPU-S3		1.34	1.15	6.47
	A2SCPU		1.22	1.20	6.60
	A1SHCPU		0.62	0.48	3.79
	A1SJHCPU	A18 171 AT21 P	0.62	0.48	3.79
	A2SHCPU	A1337 IAI218	0.51	0.64	4.24
	A2ASCPU(S1)		0.69	0.47	4.38
Mastar	A2USHCPU-S1		0.44	0.51	4.05
station for	Q2ASCPU(S1)		0.95	0.68	4.60
the second	Q2ASHCPU(S1)		0.49	0.60	4.21
tier	AnNCPU		0.8	1.2	6.4
	A2ACPU(S1)		0.54	0.54	4.32
	A3ACPU		0.48	0.52	4.16
	A2UCPU(S1)		0.54	0.54	4.32
	A3UCPU	AJ71AT21B	0.48	0.52	4.16
	A4UCPU		0.48	0.51	4.16
	Q2ACPU(S1)		0.94	0.70	4.61
	Q3ACPU		0.91	0.52	4.24
	Q4ACPU		0.54	0.56	4.17
	A2ASCPU(S1)		0.38	0.93	4.29
	A2USHCPU-S1	A18 171 AT21 P	0.26	0.81	4.00
	Q2ASCPU(S1)	A1337 IAI216	0.48	1.50	4.98
	Q2ASHCPU(S1)		0.31	0.98	4.39
	AnNCPU		0.8	1.2	6.4
Master	A2ACPU(S1)		0.54	0.54	4.32
station for	A3ACPU		0.48	0.52	4.16
the third tier	A2UCPU(S1)		0.54	0.54	4.32
	A3UCPU	AJ71AT21B	0.48	0.52	4.16
	A4UCPU		0.48	0.52	4.16
	Q2ACPU(S1)		0.48	1.50	4.98
	Q3ACPU		0.40	1.20	4.65
	Q4ACPU		0.31	0.98	4.39

POINT

At the master station for the third tier, link refresh processings for local stations for the second tier and master station for the third tier are executed. (2) Local stations

Use the following formula to calculate the link refresh time α_2 required for a local station.

When the link operates in the MELSECNET II mode or the MELSECNET II composite mode, the number of B/W points is all points set with the first half and second half link parameters.

$$\alpha_2 = K_{L1} + K_{L2} \times \frac{B + X_1 + Y_1}{2048} + K_{L3} \times \frac{W}{1024} [ms]$$

Мс	odel	KL1	K 12	KL3
CPU module	Link module		, till	1420
A1SCPU		0.82	1.18	6.52
A1SJCPU-S3		0.82	1.18	6.52
A2SCPU		0.86	0.98	6.43
A1SHCPU	A1SJ71AT21B	0.33	0.64	3.86
A1SJHCPU		0.33	0.64	3.86
A2SHCPU		0.32	0.67	4.22
A2ASCPU(S1)		0.47	0.54	4.22
A2USHCPU-S1		0.29	0.54	4.04
Q2ASCPU(S1)		0.76	0.69	4.50
Q2ASHCPU(S1)		0.41	0.51	4.13
AnNCPU		0.8	1.2	6.4
A2ACPU(S1)		0.54	0.54	4.32
A3ACPU		0.48	0.52	4.16
A2UCPU(S1)		0.54	0.54	4.32
A3UCPU	AJ71AT21B	0.48	0.52	4.16
A4UCPU		0.48	0.51	4.16
Q2ACPU(S1)		0.80	0.59	4.56
Q3ACPU	1	0.71	0.57	4.31
Q4ACPU	1	0.38	0.56	4.19

(3) Remote I/O stations

Use the following formula to calculate the I/O refresh time α ₃ required for a remote I/ O station.

$\alpha 3 = K_{R1} + K_{R2} \times \frac{X_1 + Y_1}{256}$ [ms]		
Model	KR1	Kr2
A1SJ72T25B	0.04	0.8
AJ72T25B	0.04	0.0

MELSECNET mode	MELSECNET			MELSECNET/B		
Operating Mode	MELSECNET MELSECNET I mode		MELSECNET II composite mode	MELSECNET MELSECNET I MEL mode Melsecnet I com		MELSECNET II composite mode
Applicability	0	0	0			

MELSEC-A

6.2.3 Link data communication time (link scan)

The following describes the method used for calculating the link data communication time.

When a MELSECNET data link system is used

(1) In the MELSECNET mode

LS = K + K_B × (Total number of remote I/O stations) + K_L × (Total number of local stations) + K_B [ms]

(2) In the MELSECNET II mode

LS = K + K_L x [Total number of local stations + Number of local stations allocated to second half link parameters] + K_B [ms]

(3) In the MELSECNET II composite mode

 $LS=K+K_R\times [Total number of remote I/O stations]+K_L \ [Total number of local stations + Number of local stations allocated to second half link parameters]+K_B [ms]$

(4) Find K, KL and KR in the calculation expression from the following.

Total number of slave stations	1 to 8	9 to 16	17 to 24	25 to 32	33 to 40	41 to 48	49 to 56	57 to 64
к	3.5	4.0	4.5	5.0	5.5	6.0	6.5	7.0
Kr	1.3	1.3	1.4	1.4	1.5	1.5	1.6	1.6
K∟	2.0	2.0	2.1	2.1	2.2	2.2	2.3	2.3

(5) Calculate the total number of link points (bytes) and use the following graph to obtain the KB value in the calculation expression.

Total number of link points = $\frac{B+X_0 + Y_0 + (W \times 16)}{8192}$

- B : Total points for link relays (B) that are used on all stations
- W : Total points for link registers (W) that are used on all stations
- Xo: Total points for link inputs (X) that are assigned to master station
- Yo : Total points for link outputs (Y) that are assigned to master station



6 - 17

MELSECNET mode	MELSECNET			MELSECNET/B			
Operating Mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	
Applicability				0	0	0	

When a MELSECNET/B data link system is used

(1) In the MELSECNET mode

LS = K + KR x (Total number of remote I/O stations) + KL x (Total number of local stations) + KB [ms]

(2) In the MELSECNET II mode

 $\label{eq:LS} LS = K + K_L \; x \; [Total number of local stations + Number of local stations allocated to second half link parameters] + K_B [ms]$

(3) In the MELSECNET II composite mode

LS = K + KRx [Total number of remote I/O stations] + KL x [Total number of local stations + Number of local stations allocated to second half link parameters] + KS [ms]

> (4) K, KL and KR values in the calculation expression vary depending on the communication speed of MELSECNET/B data link system. Find them from the following.

Communication	Constant	Total number of slave station				
speed setting (BPS)		1 to 8	9 to 16	17 to 24	25 to 31	
125K	к	6.7	7.2	7.7	8.2	
	KL	3.8	3.8	3.9	3.9	
	KR	3.9	3.9	4.0	4.0	
250K	К	5.8	6.3	6.8	7.3	
	KL	3.1	3.1	3.2	3.2	
	KR	3.1	3.2	3.3	3.3	
500K	К	5.8	6.3	6.8	7.3	
	KL	2.7	2.7	2.8	2.8	
	KR	2.9	2.9	3.0	3.0	
1M	к	5.8	6.3	6.8	7.3	
	KL	2.6	2.6	2.7	2.7	
	KR	2.8	2.8	2.9	2.9	

(5) By calculating the total number of link points (bytes) and from one of the graphs 1) to4) on the next page, obtain a KB value used for the calculation expression.

(Total number of link points) = $\frac{B + X_0 + Y_0 + (W + 16)}{8192}$

MELSEC-A



6 - 19

MELSECNET mode	MELSECNET			MELSECNET/B			
Operating Mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	
Applicability	0	0	0	0	0	0	

MELSEC-A

6.3 Transmission Delay Time in Three-Tier System

Calculate the transmission delay time for a three-tier system by adding the following delay factors to the transmission delay time.

- The transmission delay time from the master station/local station for the second tier to the master station for the third tier.
 Use the formulas in Section 6.2.3 to obtain this delay time.
- (2) The transmission delay time from the master station for the third tier to the local station for the third tier.Use the formulas in Section 6.2.3 to obtain this delay time.
- (3) The time required for sending the data received from the second tier to the third tier. Add either the scan time for the master station for the third tier or the link scan time for the third tier, whichever is longer. However, if the master station for the third tier has selected the mode in which the link refresh is executed only after the execution of an END instruction of a sequence program and the link scan time for the third tier is longer, add the following factor:

(Three-tier link scan time) + (Scan time for the master station for the third tier)



Fig 6.2 Three-tier system

Examples:

- (1) To transmit B/W data from M to $\ell 1$
 - (a) If (L1/m scan time) > (third tier-1 link scan time) (Transmission delay time from M to L1/m) + (Transmission delay time from L1/m to ℓ1) + (L1/m scan time)
 - (b) If (L1/m scan time) < (third tier-1 link scan time) (Transmission delay time from M to L1/m) + (Transmission delay time from L1/m to ℓ1) + (third tier-1 link scan time)
- (2) To transmit B/W data from $\ell 1$ to M
 - (a) If (L1/m scan time) > (third tier-1 link scan time)
 (Transmission delay time from ℓ1 to L1/m) + (Transmission delay time from L1/m to M) + (L1/m scan time)
 - (b) If (L1/m scan time) < (third tier-1 link scan time) (Transmission delay time from ℓ1 to L1/m) + (Transmission delay time from L1/m to M) + (third tier-1 link scan time)
- (3) To transmit B/W data from ℓ '1 to L3
 - (a) If (L2/m scan time) > (third tier-2 link scan time)
 (Transmission delay time from ℓ'1 to L2/m) + (Transmission delay time from L2/m to L3) + (L2/m scan time)
 - (b) If (L2/m scan time) < (third tier-2 link scan time) (Transmission delay time from ℓ'1 to L2/m) + (Transmission delay time from L2/ m to L3) + (third tier-2 link scan time)

When the master station for the third tier executes link refresh after the execution of an END instruction of a sequence program, add the L1/m scan time to the values obtained with calculation in (1)(b) or (2)(b). If (3)(b) is used for the calculation, add the L2/m scan time.

6.4 Time to Access Another Station from a Peripheral Device

This section describes the processing time when accessing to other station from a peripheral device.

(For the processing time of LRDP/LWTP and RFRP/RTOP, refer to Section 6.2.3.) When a MELSECNET/B data link system is used, the transmission processing time varies depending on the set communication speed and the total number of stations.

To transfer a sequence program (6K steps) to other station from a peripheral device using a MELSECNET/B data link system, the processing time requires.

- 4 minutes and 7 seconds (to write to other station) *1
- 2 minutes and 1 seconds (to read from other station) *2
- 1 minutes and 56 seconds (to verify with other station) *2

The above data can be applied when the communication speed is set to 1MBPS and the total number of stations is 32.

If either a larger sequence program is sent or the communication speed is set slower, the processing time will become longer than above-mentioned values.

If the total number of stations is less than 32, the processing time becomes shorter than the above.

REMARK

*1...... When the CPU module to which a sequence program is written is set to STOP.
 *2....... When the CPU module is set to RUN.
MELSECNET mode		MELSECNET		MELSECNET/B		
Operating Mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode
Applicability	0	0	0	0	0	0

MELSEC-A

7 DATA LINK SETTINGS

This chapter describes the setting of the number of modules and the setting of the network refresh parameters and link parameters, which are required for communications with other stations in the data link system.

7.1 Data Link Settings Overview

Set the number of modules and the network refresh parameters only when the AnUCPU, QnACPU, A2ASCPU(S1), A2USHCPU-S1, Q2AS(H)CPU(S1) or QCPU-A is used. The data link system can use any of the three operation modes according to the combination of link modules to be connected.

The operation mode can be selected by link parameter settings.

- MELSECNET mode
- + MELSECNET ${\rm I\!I}$ mode
- MELSECNET II composite mode

Set the link parameters by using a peripheral device, and store them in the programmable controller CPU of a master station.

Use the following peripheral device or software which is compatible with the AnACPU or later to set link parameters for the MELSECNET II mode or the MELSECNET II composite mode.

- A6GPP + SW5GP-GPPAE system FD
- A6PHP + SW5GP-GPPAE system FD
- IBM PC/AT + SW0IX-GPPAE system FD
- GX Developer

The following describes link parameters required to use each operation mode.

POINT

- (1) For the AnUCPU, A2ASCPU(S1) or A2USHCPU-S1 use the following peripheral device or software compatible with the AnUCPU or later.
 - A6GPP + SW0GP-GPPAUE system FD
 - A6PHP + SW0GP-GPPAUE system FD
 - GX Developer
- (2) For the QnACPU or Q2AS(H)CPU(S1), use the following software compatible with the QnACPU.
 - IBM PC/AT + SW0IX-GPPAUE system FD
 - GX Developer
- (3) For the software compatible with the QCPU-A, refer to "Software package" in QCPU-A (A Mode) User's Manual SH-080065.
- (4) When using MELSECNET and MELSECNET/10 in combination, refer to the AnUCPU or QnACPU corresponding MELSECNET/10 Reference Manual.

MELSECNET mode		MELSECNET		MELSECNET/B		
Operating Mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode
Applicability	0	0	0	0	0	0

MELSEC-A

7.2 Setting of the Number of Modules

Set the number of the data link modules installed to the AnUCPU, QnACPU, A2ASCPU(S1), A2USHCPU-S1, Q2AS(H)CPU(S1) or QCPU-A and head I/O numbers etc.

Settings must be made to the master and local stations.

- (1) Setting items
 - (a) Number of network modules
 Set the total number of the MELSECNET (II) data link modules and
 MELSECNET/10 network modules that are installed to the AnUCPU, QnACPU,
 A2US(H)CPU(S1), Q2AS(H)CPU(S1) or QCPU-A.
 The setting range is 1 to 4.
 Note that the number of mountable data link modules is limited to two.
 - (b) Valid module number for other station access Set which module will be the target of other station access from the peripheral device (SW4GP-GPPA, SW0SRXV-GPPA, etc.) or special function module (AJ71C24-S8, AD51H-S3, etc.) incompatible with the AnUCPU, QnACPU, A2ASCPU(S1), A2USHCPU-S1, Q2AS(H)CPU(S1) or QCPU-A. The setting range is 1 to 4.
 - (c) Head I/O Number
 Set the head I/O number (first 3 digits of 4-digit hexadecimal) of the data link module installed.
 For example, set "5" if the I/O numbers are X/Y50 to 6F.
 *: In GX Developer, set it in 4 digits. In the above example, enter "0050".
 - (d) Network module type name Select the type of the data link module installed from the followings.

5: MELSECNET II (master station)6: MELSECNET II (local station)

(e) Network No.

This setting is not required for the MELSECNET ${\mathbb I}.$

(2) Setting example

An example of setting the number of modules is shown below.



(a) System configuration example

Fig 7.1 System configuration example

(b) Setting screen

Fig 7.2 shows the setting screen for the above system configuration example.

[1]			
D BY GPP []			
[01]			[]
MELSECNETI			
MASTER		-	
STATION			
	[]	1	r 1
	D BY GPP [] MODULE 1 [01] MELSECNET II MASTER STATION	D BY GPP [] MODULE 1 MODULE 2 [01] [] MELSECNET II MASTER STATION	D BY GPP [] MODULE 1 MODULE 2 MODULE 3 [01] [] [] MELSECNET II MASTER STATION

Fig 7.2 Screen for setting the number of modules

MELSECNET mode		MELSECNET		MELSECNET/B		
Operating Mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode
Applicability	0	0	0	0	0	0

MELSEC-A

7.3 Network Refresh Parameters

The network refresh parameters are set to transfer the link device (LB, LW, LX, LY) which are stored in the data link module to the devices that can be used in a sequence program. In the network refresh parameters, set the head link device number of the data link module, the head device number of the AnUCPU, QnACPU, A2ASCPU(S1), A2USHCPU-S1, Q2AS(H)CPU(S1) or QCPU-A, and the transfer size. The devices refreshed by this setting are those in the transfer size that begins with the head link device number and in the range assigned using the link parameter.

For example, LB800 to FFF of the data link module are refreshed by B800 to FFF of the AnUCPU, QnACPU, A2ASCPU(S1), A2USHCPU-S1, Q2AS(H)CPU(S1) or QCPU-A in the following settings.

: B800

: LB800

<Network refresh parameters>

Head device No.

Head link device No. Transfer size

<Link parameter>

Assignment range



: 4096 points (1000H)



POINT

Between data link modules, do not set the parameter for transfer between data links.

Use the parameter for transfer between data links when transferring link device data between the data link module (MELSECNET) and network module (MELSECNET/10).

(1) Default values

The default values of the network refresh parameters are assigned according to the setting of the number of modules.

No settings are required unless specific changes must be made to the default values. Table 7.1 shows the default values. The module Nos. correspond to the numbers in the setting of the number of modules.

When a MELSECNET II data link module is mounted, it is always assigned to the area of module No. 1. If two MELSECNET II data link modules are mounted, they are assigned to the area for one module.

Number of mounted modules	Module				
Number of mounted mountes	Module No. 1	Module No. 2			
1	LB/LW0 to FFF \rightarrow B/W0 to FFF	_			
I I	LX/LY0 to 7FF \rightarrow X/Y0 to 7FF	_			
2	-	-			

Table 7.1 Default values of network refresh parameters

When a MELSECNET II data link module is mounted, the default values of the network refresh parameters are assigned as follows:

- 1) Always assigned to the area of module No. 1.
- 2) If two MELSECNET II data link modules are mounted, they are assigned to the area for one module.
- 3) The MELSECNET II data link module is not refreshed since it does not have SB/SW.

The data link information of the MELSECNET II is stored into M/D9200 to 9255.

4 LX/LY000 to 7FF of the MELSECNET II is assigned to the area of X/Y000 to 7FF.

For example, when the MELSECNET II master station is set as the network module No. 1 in the setting of the number of modules as shown in the example in Fig 7.3, the default values of the network refresh parameters are assigned as follows.

1) The MELSECNET II master station set to module No. 1 in the setting of the number of modules is assigned to the area of module No. 1 shown in Table 7.1.

	Module No. 1	Module No. 2	Module No. 3
Head I/O No.	[01]	[]	[]
Network module type	MELSECNET II master station	-	-
Network No.	-	-	-
Refresh range	LB/LW0 to FFF \rightarrow B/W0 to FFF		
(Default values)	LX/LY0 to 7FF \rightarrow X/Y0 to 7FF	-	-

Fig 7.3 Default ranges of network refresh parameters

(2) Setting items

The setting items of the network refresh parameters include the refresh ranges of LB, LW, LX, LY, SB and SW and the setting of the error history area. The following describes each setting item.

Table 7.2 shows the network refresh parameter setting items of the MELSECNET II.

Itom	MELSEC	CNET/10	MELSECNET II		
lien	Setting 1	Setting 2	Setting 1	Setting 2	
LB ⇔ B transfer	0	0	0	×	
LW ⇔ W transfer	0	0	0	×	
LX ⇔ X transfer	0	0	0	×	
LY ⇔ Y transfer	0	0	0	×	
SB transfer device	0	×	×	×	
SW transfer device	0	×	×	×	
$LB \to \text{ extension file register transfer}$	0	×	×	×	
$LW \rightarrow \text{ extension file register transfer}$	0	×	×	×	
Error history area setting	0	×	×	×	

Table 7.2 Network refresh parameter setting items

○ :Available × : N/A

(a) Refresh range setting for LB

1) The refresh destinations of LB are the link relay (B).



2) The transfer size is in units of 16 points.

- (b) Refresh range setting for LW
 - 1) The refresh destinations of LW are the link register (W).



- 2) The transfer size is in units of 1 point.
- (c) Refresh range setting for LX/LY
 - The refresh destination of LX is the input (X), and that of LY is the output (Y) The actual I/O and later can be set as the refresh destination. Since this area is also used for MELSECNET/10, MELSECNET/MINI, CC-Link, etc., assign it without overlapping.
 - 2) The transfer size is in units of 16 points.

(3) Setting example

In the system configuration exemplified in Fig 7.4, an example of setting the network refresh parameters to assign the refresh ranges in Table 7.3 is shown below.



Table 7.3 Refresh ranges

Module type	MELSECNETI master station Module No. 1	Module No. 2	Module No. 3
Refresh range	LB/LW0 to FFF \rightarrow B/W0 to FFF LX/LY0 to 7FF \rightarrow	-	-
	X/Y0 to 7FF		

(a) Setting screens

Setting of network module No. 1

	1:0000-0 2:)7FF		
	3: 4:			
	٦.			
[PAGE 1] NETWOR	K MODULE	1	·····	
MELSECNET II (MA	STER STN.	.) I/O No.	01	NETWORK NO
LB <> B	LB [0000] <> B	[0000]	SIZE[0800]HPT.
LW <> W	LB [0000] <> W	[0000]	SIZE[0800]HPT.
LX/LY <> X/Y	LX/LY C)000 <> X/Y	[0000]	SIZE[0800]HPT.

MELSECNET mode	MELSECNET			MELSECNET/B		
Operating Mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode
Applicability	0			0		

MELSEC-A

7.4 Link Parameters

7.4.1 Link parameters to be set for the MELSECNET mode

When the MELSECNET mode is used, only one type of link parameter is used. On the link parameter setting screen, set the data for the following items.

- Total link slave stations The total number of slave stations (local stations, remote I/O stations) to be connected.
- (2) Monitoring time (Refer to Section 7.5.) The monitoring time is used by the system to determine whether communication between the master station and all slave stations (local stations and remote I/O stations) is being executed normally.
- (3) Master station assignment Assign the device range (B/W0 to 3FF) to be used by the master station for writing data to the link relays (B) and link registers (W).
- (4) Slave station type
 Set the type of slave station (local station or remote I/O station) to be connected per station number.
- (5) Local station assignment
 - (a) Assign the device range (B/W0 to 3FF) to be used by a local station for writing data to the link relays (B) and link registers (W).
 - (b) Assign the link range using the outputs (Y) of the master station and inputs (X) of a local station, and the inputs (X) of the master station and outputs (Y) of a local station.
- (6) Remote I/O station assignment
 - (a) Assign the range of I/O module installed in the remote I/O station. Set it in units of 16 points.
 - (b) Assign the link registers (W) to read/write the special function modules installed to the remote I/O station.

REMARK

The GPP link parameter setting screen is shown below.

		* LIN	K *			M : B ++	ALL L:	B 000-27F
	SLAVE	M -+ /	ALL L	W.D.T.	INTER-	M : W →	ALL R: V	V 300-341
WASIEN	STATIONS	В	w	10ms	10ms	M:Y =	ALL L: X	K 680-77F
М	4	000-00F	000-0FF	200	XXXX	M : X +	ALL L:	600-77F 200-4BF
	· M +	- L	M → R	M ← R	M -	→ L/R	M ←	L/R
NO.	В	w	w	W	Y	X/Y	x	Y/X
R 1 L 2 L 3 R 4	100-17F 200-27F	100-17F 200-2BF	300-310 320-341 -	360-36F 	230-30F 680-6FF 700-77F 480-59F - -	030-10F 200-27F 200-27F 080-19F - -	200-28F 600-67F 700-77F 400-4BF - - -	000-08F 280-2FF 200-27F 000-0BF - - -
t LOCAL	Ŭ. TE			·	M : MA	STER L:	LOCAL	R : REMO

MELSECNET mode	MELSECNET			MELSECNET/B		
Operating Mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode
Applicability		0			0	

MELSEC-A

7.4.2 Link parameters to be set for the MELSECNET II mode

When the MELSECNET II mode is used, two types of link parameter (first half and second half) are provided.

The system can be operated in the MELSECNET II mode with only the first half link parameters set.

The settings of the second half link parameters are only effective when more than 1024 bytes of link data are to be used per a station (master station or local station).

- (1) First half link parameters
 - (a) Total slave stations
 The number of local stations to be connected. (Remote I/O stations cannot be connected.)
 - (b) Monitoring time The time used by the system to determine whether communication between the master station and all slave stations (local stations) is being executed normally.
 - (c) Master station assignment Assign the device range (B/W0 to FFF) to be used by the master station for writing data from the master station to the link relays (B) and link registers (W).
 - (d) Local station assignment
 - 1) Assign the device range (B/W0 to B/WFFF) to be used by a local station for writing data to the link relays (B) and link registers (W).
 - 2) Assign the link range using the outputs (Y) of the master station and inputs (X) of a local station, and the inputs (X) of the master station and outputs (Y) of a local station.
- (2) Second half link parameters
 - (a) Master station assignment
 - Assign the device range to be used by the master station for writing data to the link relays (B) and link registers (W).
 - (b) Local station assignment Assign the device range to be used by the local station for writing data to the link relays (B) and link registers (W).

REMARK

The device range that can be assigned to the second half link parameters is the range assigned to the first half link parameters: "(final device number) + 1 " or later.

(Example: If the range of B0 to FF is assigned to the first link parameters,

"B100" or later can be assigned to the second half parameters.)

If 0 point is assigned to the first half link parameters, assignment of the second half link parameters can be started with B/W0.

POINT

- (1) The following can be set in the MELSECNET II mode:
 - 1024 bytes in the first half link parameters
 - 1024 bytes in the second half link parameters

(3) Link parameter setting screensThe following shows the link parameter setting screens.(a) First half link parameters

	*	MELSECNET	II MODE LINI	< *		M : B1 ↔ M : W1 ↔	ALL L : B1 ALL L : W	000-8FF 000-4FF
MASTER	SLAVE	FIRST	M → ALL L	W.D.T.	INTER-	M : B2 ↔ M : W2 ↔ M : W →	ALL L: B2 ALL L: W2 ALL B: W	2 C00-CFF 2 800-8FF
WASIEN	STATIONS	В	w	10ms	10ms	M:W ← M:Y →	ALL R:W	280-37F
м	3	000-0FF	000-0FF	200	XXXX	$\begin{array}{ccc} M & \vdots & Y & \rightarrow \\ M & \vdots & X & \leftarrow \end{array}$	ALL R:Y	200-37F
<u></u>				······	·····	<u>M:X ←</u>	ALL R: X	
L/R	FIRST M ← L				M → L		M ← L	
NO.	В	w			Y	X	x	Y
L 1	200-2FF	200-2FF			280-2FF	100-17F	200-27F	180-1FF
	500-4FF	400-4FF			300-37F	100-17F	300-37F	100-17F
	-	-		-	:		-	
	-	-	-	-	-	-	-	-
· ·		-	-	-	<u> </u>		L	
1					M : MAS	TER L:	LOCAL R	: REMOTE
L : LOCAL R : REMO	ŤE			PRESS <ssi< td=""><td>N> TO SELEC</td><td>T 1ST/2ND REN</td><td>IGE OF B/W</td><td></td></ssi<>	N> TO SELEC	T 1ST/2ND REN	IGE OF B/W	

		*	MELSECNET		K +		M : B1 ↔ M : W1 ↔	ALL L : B1 ALL L : W	1000-8FF 000-4FF
	MARTER	SLAVE	SECOND	M → ALL L	W.D.T.	INTER-	M : B2 ↔ M : W2 ↔	ALL L: B2 ALL L: W2	2 C00-CFF 2 800-8FF
	MASIEN	STATIONS	В	w	10ms	10ms		ALL R:W	280-37F
	M	3	C00-CFF	800-8FF	200	XXXX	M Y → M X ←	ALL R: Y ALL L: Y	200-37F
İ.							<u>M:X +</u>	<u>ALL R:X</u>	•
	L/R	SECOND	M ← L			N	l → L	M	- L
	NO.	В	W			Y	x	×	Y
	L 1	-	-						
	L 211	-	-					• • • • • • • •	
	L 311	-	-						
		-	-	- 1	-	-	•	-	-
		-	•	-	-	-	-	-	
		•	-	•	•				
		-	-	-	-		-	-	-
	t	<u> </u>				M : MAS	STER L:	LOCAL R	: REMOTE
	L : LOCAL R : REMO	TE			PRESS <ss< td=""><td>N> TO SELE</td><td>CT 1ST/2ND RE</td><td>NGE OF B/W</td><td></td></ss<>	N> TO SELE	CT 1ST/2ND RE	NGE OF B/W	

(b) Second half link parameters

MELSECNET mode	MELSECNET			MELSECNET/B		
Operating Mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode
Applicability			0			0

7.4.3 Link parameters to be set for the MELSECNET II composite mode

When the MELSECNET II composite mode is used, two types of link parameters (first half and second half) are provided.

The system can be operated in the MELSECNET II composite mode with only the first half link parameters set. In this case, however, the device range used for data link is B/W0 to 3FF, and the system will operate on the same level as the MELSECNET mode system. Assign the range of B/W400 to FFF of the link relays (B) and link registers (W) to the second half link parameters.

First half link parameters

(The same data as that set for the MELSECNET mode is set for the first half link parameters.)

- Link total slave stations
 The number of slave stations (local stations, remote I/O stations) to be connected.
- (2) Monitoring time (Refer to Section 7.5.)

The time used by the system to determine whether communication between the master station and all slave station (local stations and remote I/O stations) is being executed normally.

(3) Master station assignment Assign the device range (B/W0 to 3FF) to be used by the master station for writing data to the link relays (B) and link registers (W).

(4) Slave station type

Set the type of slave station (the local station for the MELSECNET mode or the local station/remote I/O station for the MELSECNET II mode) per station number.

- (5) Local station assignment
 - (a) Assign the device range (B/W0 to 3FF) to be used by a local station for writing data to the link relays (B) and link registers (W).
 - (b) Assign the link range using the outputs (Y) of a master station and inputs (X) of a local station, and the inputs (X) of the master station and outputs (Y) of a local station.
- (6) Remote I/O station assignment
 - (a) Assign the number of points of the I/O module or special function module to be installed to the remote I/O station.
 - (b) Assign the link registers (W) to be used for reading/writing the special function modules installed to the remote I/O station.

Second half link parameters

(The same data as that set for the MELSECNET ${\rm I\!I}$ mode should be set for the second half link parameters.)

 Master station assignment Assign the device range to be used by the master station for writing data to the link relays (B) and link registers (W).

(2) Local station assignment

Assign the device range to be used by the local station for writing data to the link relays (B) and link registers (W).

REMARK

The device range that can be assigned to the second half link parameters is the range assigned to the first half link parameters: "(final device number) + 1". (Example: If the range of B0 to FF is assigned to the first link parameters, "B100" or later can be assigned to the second half parameters.) If 0 point is assigned to the first half link parameters, assignment of the second half link parameters can be started with B/W0. Link parameter setting screen

The following shows link parameter setting screens.

(a) First half link parameters

	MELSECNET II MULTI MODE LINK M : B1 ↔ ALL L : B1 000-37F M : W1 ↔ ALL L : W1 000-2BF										
	SLAVE FIRST		M → ALL L	W.D.T.	INTER-	M : B2 ++ M : W2 ++	ALL R: B2 ALL R: W2	800-8FF 800-AFF			
MASTER	PC STATIONS	В	w	LINK 10ms	MITTENT 10ms	$\begin{array}{c} M : W & \leftarrow \\ M : W & \leftarrow \\ M : Y & \leftarrow \\ M : Y & \leftarrow \end{array}$	ALL R: W ALL L: X	360-341 360-39F 680-77F 230-59F			
м	4	000-0FF	000-0FF	200	XXXX	M:X + M:X +	ALL L:Y ALL R:X	600-77F 200-4BF			
L/R	FIRST M ← L		M → R	M + R	м	M → L/R		L/R			
NO.	В	w	w	w	Y	X/Y	x	Y/X			
R 1 L 2 L 3II R 4	100-1FF 280-37F	100-1FF 240-28F -	300-310 320-341	360-36F 380-39F	230-30F 680-6FF 700-77F 480-59F	030-10F 200-27F 200-27F 080-19F	200-28F 600-67F 700-77F 400-48F - -	000-08F 280-2FF 200-27F 000-0BF			
t L : LOCA R : REMO * : MELS	L ITE ECNET II (LOO	CAL)		PRESS <ssi< td=""><td>M : MAS</td><td>STER L : CT 1ST/2ND REM</td><td>LOCAL R NGE OF B/W</td><td>: REMOTE</td></ssi<>	M : MAS	STER L : CT 1ST/2ND REM	LOCAL R NGE OF B/W	: REMOTE			

(b)	Second half	link parameters
-----	-------------	-----------------

		* ME	LSECNET II I		M : B1 ↔ M : W1 ↔	ALL L: B1 ALL L: W1	•		
		SLAVE	SECOND	M → ALL L	W.D.T.	INTER-	M : B2 ↔ M : W2 ↔	ALL L : B2 ALL L : W2	800-AFF 800-AFF
1	MASTER	STATIONS	В	w	LINK 10ms	MITTENT 10ms	$\begin{array}{cccc} M & W & \rightarrow \\ M & W & \leftarrow \\ M & Y & \rightarrow \\ M & Y & \rightarrow \end{array}$	ALL R:W ALL R:W ALL L:X	•
	M	4	800-8FF	800-8FF	200	XXXX	M : X ← M : X ←	ALL L:Y ALL R:X	-
		SECOND		M→R M←		M → L/R		M ← L/R	
	NO.	В	w	w	w	Y	X/Y	×	Y/X
	R 1 L 2 L 3 II R 4	A00-AFF	A00-AFF	······ ·······························	 - -		· · · · · · · · · · · · · · · · · · ·	 	
	t L : LOCAI R : REMO * : MELSI	TE ECNET II (LOO	CAL)		PRESS <ssm< td=""><td>M : MA N> TO SELE</td><td>STER L : CT 1ST/2ND REM</td><td>LOCAL R NGE OF B/W</td><td>: REMOTE</td></ssm<>	M : MA N> TO SELE	STER L : CT 1ST/2ND REM	LOCAL R NGE OF B/W	: REMOTE

MELSECNET mode		MELSECNET		MELSECNET/B			
Operating Mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	
Applicability	0	0	0	0	0	0	

7.5 Monitoring Time Setting

The monitoring time is the reference time used by the system to determine whether communication between the master station and all slave stations (local stations and remote I/O stations) is being executed normally.

- If the link scan is repeatedly executed within the monitoring time (set time), the local stations and remote I/O stations determine that the master station is operating normal (data link normal).
- (2) If the link scan is not repeated within the monitoring time, the local and remote I/O stations determine that the master station is faulty (data link error) and the following processing is executed.
 - (a) Master station
 - 1) Communication with all slave stations stops.
 - If the automatic return function is selected, communication is retried.
 - 2) Receive data is maintained in a status just before the occurrence of the communication error.
 - 3) The value "5" is stored to data link special register D9204.
 - (b) Local stations
 - 1) The ERROR LED "TIME" of the link module lights.
 - 2) Receive data is maintained in a status just before the occurrence of the communication error.
 - 3) Data link special relay M9251 (link stop) is turned ON.
 - (c) Remote I/O stations
 - 1) The ERROR LED "TIME" of the link module lights.
 - 2) All output modules installed to the remote stations are turned OFF.



REMARK

(1) For details of data link special register, refer to Section 9.3.

(2) For details of data link special relay, refer to Section 9.2.

- (3) Setting the monitoring time
 The monitoring time can be set between 10 to 2000ms in units of 10ms. (The usual setting is 2000ms.)
 Setting the minimum value to monitoring time is valid for the case to immediately turn off the output of remote I/O stations where communication error occurred.
 Use the following procedure to set the minimum value to monitoring time.
 - (a) Set "200" (2000ms) to monitoring time and write the setting to the master station.
 - (b) Perform data link in the actual system and monitor the link scan by link monitor of GPP connected to the master station. (Refer to Section 10.1.1 for the link monitor.)
 - (c) Turn off the power supply to one of the slave stations so that the system operates in the loopback mode.
 - (d) Read the maximum link scan time value.
 - (e) Add 50ms to the maximum link scan time value read out and set it as monitoring time. When using the A70BD-J71P21/R21 or A7BD-J71AP21/R21 as a master station, add 250ms to the maximum link scan time value and set it as monitoring time.

MELSECNET mode		MELSECNET		MELSECNET/B			
Operating Mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	
Applicability	0	0	0	0	0	0	

7.6 Common Element

The following describes the common information to know before assigning link parameters.

7.6.1 Maximum number of link points per station

The maximum number of link points (number of points assigned to a station by link parameters) of "B, W, X, Y" that can be used for data link of a station (master, local or <u>remote I/O station</u>) in the data link system.

(1) Maximum number of link points per station (master station and local stations)(a) MELSECNET mode

Up to 1024 bytes of link points can be used by a station (master or local station). Use the following formula to calculate the maximum number of link points.

$$\frac{(B \text{ points}) + (Y \text{ points})}{8} + \{2 \times (W \text{ points})\} \le 1024 \text{ bytes}$$

(b) MELSECNET I mode or MELSECNET I composite mode Up to 1024 bytes of link points can be assigned to the first half link parameters and up to 1024 bytes can be assigned to the second half link parameters used by a station (master or local station).

Use the following formula to calculate the maximum number of link points.



MELSECNET mode		MELSECNET		MELSECNET/B		
Operating Mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode
Applicability	0	0	0	0	0	0

(2) Maximum number of link points per remote I/O station

The maximum number of link points which can be assigned to a remote I/O station is 512 I/O points (X/Y0 to 1FF) or less, and the total link data size must be set to 512 bytes.

Use the following formula to calculate the maximum number of link points.

$\frac{(X \text{ points}) + (Y \text{ points})}{8} + \{2 \times (W \text{ points})\} = 8$	512 bytes
(X points) ≤ 512 bytes	
(Y points) ≤ 512 bytes	

Note that the maximum number of I/O points that can be used in a remote I/O station is 512 points for the total of input, output, and special function modules.

7.6.2 Determining the link relay (B) assignment range

Assign link relays (B) as follows:

- (1) Assign link relays (B) in units of 16 points ($B \Box \Box 0$ to $B \Box \Box F$).
- (2) The device range that can be assigned to the second half link parameters is the range assigned to the first half link parameters: "(final device number allocated with the first half link parameters) + 1 " or later.

(Example: If the range of B0 to FF is assigned to the first link parameters, "B100" or later can be assigned to the second half parameters.)

If 0 point is assigned to the first half link parameters, assignment to the second half link parameters can be started with B000.



MELSECNET mode		MELSECNET		MELSECNET/B			
Operating Mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	
Applicability	0	0	0	0	0	0	

MELSEC-A

7.6.3 Determining the link register (W) assignment range

Link registers (W) are used for communications between the master station and a local station and between the master station and a remote I/O station. Assign the link register in units of 1 point.

Assign the link register to the area for communication

Assign the link registers to the area for communications between the master and a local station, and the area for communications between the master and a remote I/O station separately, as shown below.

- (1) When assigning the link parameters of the MELSECNET mode and the first half link parameters of the MELSECNET II composite mode
 - (a) The area used by the master or local station to write data to host station (hereinafter referred to as the "M/L area".)
 - (b) The area used to read/write from the special function module installed to a remote I/O station (RFRP/RTOP instruction) (hereinafter referred to as the "M/R area".)

1) Further divide M/R area into a read area and a write area and assign them.

- a) The area used to read data from a remote I/O station to the master station (hereinafter referred to as the "M ← R area")
- b) The area used to write data from the master station to a remote I/O station (hereinafter referred to as the "M \rightarrow R area")



Fig 7.6 Link register (W) assignment range

2) The system uses the M → R area to execute RFRP/RTOP instructions. Consider the range used by the system when assigning the M → R area. [Number of points used by the system] The link register (W) 1 point is used for one special function module installed to a remote I/O station. [Range used by the system] The range used by the system begins at the M → R area head device

number assigned to each remote I/O station to "number of use points -1 ".

Example When three special function modules are installed to remote I/O station No.2 with link parameters set as shown below, W360 to 362 in the M \rightarrow R area (W360 to 3AF) are used by the system. [Setting of link parameters] WO 200 260 2B0 300 360 3B0 3FF R۱ R₂ R₃ Rı R2 R3 M ← R area LM → R area M/R area [Assignment of remote I/O station No. 2] Link registers W000 W360 For system W361 For system For system W362 $M \rightarrow R$ area for R2 W363 Can be used with S RTOP instruction **W3AF**

(2) The device range that can be assigned to the second half link parameters is the range assigned to the first half link parameters: "(final device number) + 1 " or later. (Example: If the range of B0 to FF is assigned to the first link parameters, "B100" or later can be assigned to the second half parameters.)

If 0 point is assigned to the first half link parameters, assignment of the second half link parameters can be started with W000.

POINT

- When only the RFRP instruction is used, set the M → R area used by the system in the link parameter.
- (2) If the M → R area used by the system is used incorrectly in a user program, data cannot be read/written correctly, when the RFRP/RTOP instruction is executed.

MELSECNET mode		MELSECNET		MELSECNET/B			
Operating Mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	
Applicability	0	0	0	0	0	0	

MELSEC-A

7.6.4 Determining the input (X) and output (Y) assignment range

Assignment of input (X) and output (Y) can be executed to the link parameter of MELSECNET mode and the first half link parameters of MELSECNET II composite mode. Input (X) and output (Y) which can be used in the data link is the range installing the I/O module and special function module by the master station or local station or later.

- (1) Divide inputs and outputs as shown below and assign them.
 - (a) The area used for communication between the master station and a local station (hereinafter referred to as the "M/L area").
 - (b) The area used for communication between the master station and a remote I/O station (hereinafter referred to as the "M/R area").



Fig 7.7 Input (X) and output (Y) assignment range

(2) Inputs (X) and outputs (Y) are assigned to each station in units of 16 points (X□□0 to X□□F and Y□□0 to Y□□F).



- Inputs (X) and outputs (Y) are assigned to the first half link parameters in the MELSECNET II mode.
- Since remote I/O stations cannot be connected in a MELSECNET II mode, the area does not need to be divided for M/L and M/R.

MELSECNET mode		MELSECNET			MELSECNET/B	1
Operating Mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode
Applicability	0			0		

MELSEC-A

7.7 Link Parameters in the MELSECNET Mode

The following describes the link parameter setting when the MELSECNET mode is used. (When the MELSECNET composite mode is used, the first half link parameters can be set in a similar way.)

The system configuration in the MELSECNET mode includes the following three types.

- System consisting of the master station and local stations (hereinafter Referred to as a "local system").
- System consisting of a master station and remote I/O stations (hereinafter referred to as a "remote I/O system").
- System consisting of a master station, local stations, and remote I/O stations (hereinafter referred to as a "local/remote I/O system").

Concept and precautions at link parameter setting are described per each system configuration.

7.7.1 Local system assignment and link parameter setting example

The following describes the assignment of link relays, link registers, inputs, and outputs and the link parameter setting.

Assignment for a local system

When setting link parameters of the local system, the following points must be considered.

- (1) Determine the link relay (B) and link register (W) assignment range for each master station and local station. (Refer to Section 7.6.2 and Section 7.6.3.)
- (2) If the number of link relay (B) points is insufficient, examine to substitute inputs (X) and outputs (Y) for the information communicated between the master station and a local station (one to one). (Refer to Section 7.6.4.)
- (3) Make sure that the number of link points per station is as follows. (Refer to Section 7.6.1.)
 Master station

Master station	1024 bytes or less
Local station	1024 bytes or less

Link parameter setting example

Fig 7.8 shows a link parameter setting of a local system when the MELSECNET mode is used.



Fig 7.8 System configuration example

- (1) Assignment of link relay (B)
 - (a) Fig 7.9 shows the case when assigning 256 points to the master station, 128 points to local station No.1, 128 points to local station No.2, and 128 points empty area between local station No.1 and local station No.2.

- 80			Coil		Used as			Lised as			
				tact	relay (M) by the host station	Coil	Con- tact	internal relay (M) by the host station	Coil	Con- tact	Used as internal relay (M) by the host station
ВО	м		o	o			o			o	
B100 B180	<u>L1</u>			0		0	0			0	
B200 B280	L2			0			0		0	0	
					ο			o			o
B3FF											L
				Сс Сс О	oil : ON/ ontact : Rea : Usa	OFF co ids ON/0 ible rang	ntrol of li OFF dati je	ink relays (B a using link r) relay (B)	contact	ł
	B100 B180 B200 B280 B3FF	M B100 L1 B180 *1 B200 L2 B280 B3FF	M B100 L1 B180 *1 B200 L2 B280 B3FF	M o B100 L1 B180 *1 B200 L2 B280	M o o B100 L1 o B180 *1 o B200 L2 o B280 o o B3FF o o	M o o B100 L1 o B180 *1 o B200 L2 o B280 o o B3FF o o	M o o o B100 L1 o o B180 *1 o o B200 L2 o o B280 o o o B3FF o o o Coil : ON/OFF co Contact : Reads ON/O O : Usable range o : Usable range	M o o o o B100 L1 o o o o B180 *1 o o o o B200 L2 o o o o B280 o o o o o B3FF O cold o o o Coil : ON/OFF control of I Contact : Reads ON/OFF data O : Usable range O : Usable range	M o o o o B100 L1 o o o o B180 *1 o o o o B200 L2 o o o o B280 o o o o o B3FF O O o o o Coil : ON/OFF control of link relays (B Contact : Reads ON/OFF data using link r o : Usable range	M o o o o o B100 L1 o o o o o B180 *1 o o o o o B200 L2 o o o o o B280 o o o o o o B3FF o o o o o o Coil : ON/OFF control of link relays (B) Contact : Reads ON/OFF data using link relay (B) O : Usable range	M o

Fig 7.9 Link relay (B) assignment example

(b) The empty area marked with *1 in the M/L area in Fig 7.9 cannot be used as a substitute for internal relay (M) in the master station and all local stations. The M/L area is the range (B0 to 27F) from the minimum device number to the maximum device number assigned by the link parameters.

- (2) Assignment of link register (W)
 - (a) Assignment for the master station and local stations (M/L area)
 Fig 7.10 shows the case when assigning 256 points to the master station, 128 points to local station No.1, 196 points to local station No.2, and 128 points empty area between local station No.1 and local station No.2.



Fig 7.10 Link register (W) assignment example

(b) The empty area marked with *1 in the M/L area in Fig 7.10 cannot be used as a substitute for data register (D) in the master station and all local stations. The M/L area is the range of the smallest to the largest device number (W0 to 2BF) assigned by the link parameters.

- (3) Assignment of inputs (X) and outputs (Y)
 - (a) The range used for data link by the master station The master station uses the X/Y0 to X/Y14F range as inputs and outputs of host station.

The X/Y150 to X/Y7FF range can be used for the data link.

(b) Assignment for local stations

In this example, inputs (X) and outputs (Y) are not required because there is an empty area in the link relay (B) assignment. However, to explain the assignment example of local station, 128 input (X) points and 128 output (Y) points have been assigned.

1) Assignment of local station No.1

Local station No.1 uses the X/Y0 to 17F range as inputs and outputs for host station.

The X/Y180 to X/Y7FF range can be used for the data link.

 Assignment of local station No.2 Local station No.2 uses the X/Y0 to X/Y14F range as inputs and outputs for host station.



The X/Y150 to X/Y7FF range can be used for the data link.

Fig 7.11 Inputs (X) and outputs (Y) assignment example

(4) Link parameter setting

When the assignment of (1) to (3) is executed, set the link parameters as shown below.

$ \begin{array}{cccc} M : W & \rightarrow \\ M : W & \leftarrow \\ M : Y & \rightarrow \\ M : Y & \rightarrow \\ M : X & \leftarrow \\ M : X & \leftarrow \\ M : X & \leftarrow \\ \end{array} $	ALL R:W ALL R:W ALL L:X ALL R:Y ALL L:Y ALL R:X	680-77F 600-77F
$ \begin{array}{cccc} M : Y & \rightarrow \\ M : Y & \rightarrow \\ M : X & \leftarrow \\ M : X & \leftarrow \\ \end{array} $	ALL L:X ALL R:Y ALL L:Y ALL R:X	680-77F 600-77F
M:X ← M:X ←	ALL L:Y ALL R:X	600-77F
	Y	
- 4/11	M ←	L/R
X/Y	x	Y/X
200-27F 200-27F	600-67F 700-77F	280-2FF 200-27F
	-	-
L : LOCA	L R : RE	- MOTE
	L : LOCA	L:LOCAL R:REI

MELSECNET mode		MELSECNET			MELSECNET/B	
Operating Mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode
Applicability	0			0		

MELSEC-A

7.7.2 Remote I/O system assignment and link parameter setting example

The following describes the assignment of link relays, link registers, inputs, and outputs and link parameters used for the remote I/O system.

Assignment for a remote I/O system

For the link parameter setting of a remote I/O station, consider the following points for the assignment.

- Determine what number of master station the device number of I/O module installed to the remote I/O station is assigned.
 The I/O numbers that can be assigned to a remote I/O station is the I/O numbers used for master station as I/O of host station. (Refer to Section 7.6.4.)
- (2) When a special function module is installed to a remote I/O station, determine the link register (W) assignment range (M/R area) to be used for reading/writing buffer memory (RFRP/RTOP instruction). (Refer to Section 7.6.4.)
- (3) Make sure that the number of link points per station is as follows. (Refer to Section 7.6.1.)
 - Remote I/O station...... 512 bytes

(Inputs and outputs are less than 512 points of X/ Y0 to 1FF)

POINT

- If the M → R area used by the system is used incorrectly in a user program, data cannot be read/written correctly, when the RFRP/RTOP instruction is executed.
- (2) The number of link points can be reduced by mounting modules to a remote I/ O station in groups of input modules, special function modules, and output modules as shown below. If the installation of I/O modules is changed from (a) to (b), a reduction of 16 input assignment points and 32 output assignment points can be achieved.



Link parameter setting example

Fig 7.12 shows a link parameter setting of a remote I/O system when the MELSECNET mode is used.



Fig 7.12 System configuration example

- Assignment of link relay (B)
 Data link cannot be executed by using a link relay in a remote I/O system. (Setting at link parameter is unnecessary.)
- (2) Assignment of link register (W)
 - (a) Assignment for remote I/O stations (M \rightarrow R area, M \leftarrow R area)
 - Assign 16 points for the RTOP instruction and 16 points for the RFRP instruction in the user program of the master station as remote I/O station No.1.

No.1 requires 17 points (16 points (for RTOP) + 1 point (for OS)) for the M \rightarrow R area because one special function module is installed.

 Assign 32 points for the RTOP instruction and 32 points for the RFRP instruction in the user program of the master station as remote I/O station No.2.

No.2 requires 34 points (32 points (for RTOP) + 2 points (for OS)) for the M

- Remote I/O station No. 1 Remote I/O station No. 2 Master station (M) (R1) (R2) Used as a data register (D) by the Read from Write by Read from Write by Read Write the master the master the master the master station station station station host station w000 0 R area W300 R1 o W310 0 t W320 ≥ R2 o W341 0 M/R W360 R1 ο R area area W36F *1 W380 ţ R2 0 Σ W39F 0 **W3FF** Read : Read word data Write : Write word data о : Usable range
- \rightarrow R area because two special function modules are installed.

Fig 7.13 Link register (W) assignment example

(b) In Fig 7.13, the empty area marked with *1 in the M ← R area cannot be used as a data register (D) by the master station.

- (3) Assignment of inputs (X) and outputs (Y)
 - (a) Input and output range used for data link by master station
 The master station uses the X/Y0 to X/Y14F range as I/O of host station. The X/
 Y150 to 7FF range can be used for the data link.
 - (b) Assignment of remote I/O stations
 - Assignment of remote I/O station No.1 Inputs (X) : X0 to 8F Outputs (Y) : Y30 to 10F
 - Assignment of remote I/O station No.2
 Inputs (X) : X0 to BF
 Outputs (Y) : Y80 to 19F



Fig 7.14 Inputs (X) and outputs (Y) assignment example

(4) Link parameter setting

When the assignment of (1) to (3) is executed, set the link parameters as shown in the figure below.

 			+ LINK	*		M:B ↔	ALL L:B	*
	SLAVE	M →		W.D.T.	INTER-	$M: W \leftrightarrow I$ $M: W \rightarrow I$	ALL L:W	300-341
MASTER	PC STATIONS	в	W	LINK 10ms	MITTENT 10ms	$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	ALL R:W ALL L:X ALL R:Y	230-59F
М	2	-	-	200	XXXX	M:X ← M:X ←	ALL L:Y ALL R:X	- 200-4BF
L/R	M +	- L	M → R	M ← R	M	→ L/R1	M	L/R
NO.	В	w	w	w	Y	X/Y	×	Y/X
R 1 R 2			300-310 320-341	360-36F 380-39F	230-30F 480-59F	030-10F 080-19F	200-28F 400-4BF	000-08F 000-0BF
	-	-	-		-	•	-	-
	-	-	-	-	-	-	-	-
	-	-	-	-		-		-
Î L : LOCAL R : REMO	TE	<u>. </u>		Ň	I : MASTER	L:LOCA	L R:REI	MOTE

MELSECNET mode		MELSECNET			MELSECNET/B	
Operating Mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode
Applicability	0		0			

MELSEC-A

7.7.3 Local/remote I/O system assignment and link parameter setting example

The following describes the assignment of link relays, link registers, inputs, and outputs and the link parameters.

Assignment for a local/remote I/O system

When setting link parameters of local/remote I/O system, consider the following points for assignment.

- Determine the link relay (B)/link register (W) assignment range for each master station and local station. (Refer to Section 7.6.2 and Section 7.6.3.)
 Determine the assignment range, dividing M/L area into the one for the master station and the one for the local station.
- When a special function module is installed to a remote I/O station, link registers (W) are required for reading/writing (RFRP/RTOP instruction) buffer memory.
 M/R area is divided into the area for M ← R and the area for M → R to assign. (Refer to Section 7.6.3.)
 Divide link registers (W) into the M/L area (for communication between the master

station and local stations) and the M/R area (for communication between the master station and remote I/O stations) and assign them. (Refer to Section 7.6.3.)

wo			200	, 	300 34	0 30		
	М	L1	L4	Empty	R ₂	R₃	R ₂	R ₃
L			.		M + are	- R a	M → area	► R
						M/R	T area	

Fig 7.15 Link register (W) assignment example

- (3) Assign the input (X) and output (Y) range used for data link to the I/O number used in the I/O module and special function module by the master station I/O. (Refer to Section 7.6.4)
 - (a) Divide the area used for data link into an M/R area (for communication between the master station and a remote I/O station) and an M/L area (for communication between the master station and a local station) to assign the range.
 - (b) The M/L area is used when the number of link relay (B) points is insufficient. Therefore, it is not necessary to assign the M/L area when there is sufficient number of link relay (B) points.
- (4) Make sure that the number of link points per station is as follows. (Refer to Section 7.6.1.)
 - Master station..... 1024 bytes or less
 - Local station..... 1024 bytes or less
 - Remote I/O station...... 512 bytes or less

(Inputs and outputs is 512 points or less of X/Y0 to 1FF)

POINT

- (1) If the M → R area used by the system is used incorrectly in a user program, data cannot be read/written correctly, when the RFRP/RTOP instruction is executed.
- (2) The number of link points can be reduced by mounting modules to a remote I/O station in groups of input modules, special function modules, and output modules as shown below. If the installation of I/O modules is changed from (a) to (b), a reduction of 16 input assignment points and 32 output assignment points can be achieved.


MELSECNET mode		MELSECNET		MELSECNET/B				
Operating Mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode		
Applicability	0			0				

MELSEC-A

Link parameter setting example

The following explains the procedure for setting link parameters when a local/remote I/O system is used in the MELSECNET mode.



Fig 7.16 System configuration example

- (1) Assignment of link relays (B)
 - (a) Fig 7.17 shows the case when assigning 256 points to the master station, 128 points to local station No.2, 128 points to local station No.3, and 128 points empty area between local station No.2 and local station No.3.

		<u></u>										
					Master st	ation	Lo	cal statior	n No. 2	Lo	cal statior	No. 3
				Coil	Contact	Used as internal relays (M) by the host station	Coil	Contact	Used as internal relays (M) by the host station	Coil	Contact	Used as internal relays (M) by the host station
	BO		[
		м		0	o			0			0	
	B100	L2			0		o	o			0	
M/I	B180	*1										
area	B200	L3			o			0		o	0	
	D20U					o			ο			o
	B3FF	L	J	L	I	Coi Cor O	I : (ntact : F : I	L ON/OFF o Reads Of Jsable ra	control of lin N/OFF data nge	k relays using lin	l (B) k relay (l	3) contact

Fig 7.17 Link relay (B) assignment example

(b) The empty area marked with *1 in the M/L area in Fig 7.17 cannot be used as a substitute for internal relays (M) in the master station and all local stations. The M/L area is the range (B0 to 27F) from the minimum device number to the maximum device number assigned by the link parameters.

- (2) Assignment of link registers (W)
 - (a) Assignment for the master station and local stations (M/L area)
 256 points to the master station, 128 points to local station No.2, and 196 points to local station No.3 are assigned
 - (b) Assignment for the remote I/O stations (M → R area, M ← R area) Assign 16 points for the RTOP instruction and 16 points for the RFRP instruction in the user program of the master station as remote I/O station No.1. No.1 requires 17 points (16 points (for RTOP) + 1 point (for OS)) for the M → R area because one special function module is installed. Assign 32 points for the RTOP instruction and 32 points for the RFRP instruction in the user program of the master station for the remote I/O station No.4. For No.4, 34 points (32 points for RTOP + 2 points for OS) are required for the M → R area, because two special function modules are installed.

										r							
					Ма	ster st	ation (M)	Remote N	I/O station	Lo	cal I/O No.	station 2	Lo	cal I/C No.	station 3	Remote I	/O station 0. 4
					Read	Write	Used as a data register (D) by the host station	Read from master station	Write by the master station	Read	Write	Used as a data register (D) by the host station	Read	Write	Used as a data register (D) by the host station	Read from master station	Write by the master station
	Î	- wo	м		•	0				0			o				
area		W100	L2		0					0	0		0				
M/L 4		W180	*1														
		w200	L3		•					0			0	0			
		- 2001					0					0			0		
	Ì	-W300 to	R1						0			0			٥		
	M→R area	W310					o					0			o		
e a		to W341	R4									0			0		0
/R ar							0					0			0		
Ň		to	R1					0				0			0		
	M⊷R area	Wann	*2									0		<u> </u>	0		
		to W39F	R4									0			0	•	
							0					0			•		
				J	• • • • • • • • • • • • • • • • • • •	•	<u> </u>	•	<u> </u>				Re: Wri O	ad ite	: Reac : Writi : Usab	ling wor ng word le range	d data data e

Fig 7.18 Link register (W) assignment example

(c) The empty area marked with *1 in the M/L area in Fig 7.18 cannot be used as a substitute for data register (D) in the master station and all local stations.
 In addition, empty area of M ← R area marked with *2 cannot be used as a data register (D) by the master station.

- (3) Assignment of inputs (X) and outputs (Y)
 - (a) Input and output range used for data link by the master station The master station uses the X/Y0 to X/Y14F range as I/O of host station. For the data link, the X/Y150 to X/Y7FF range can be used.
 - (b) Assignment of remote I/O stations
 - 1) Assignment of remote I/O station No.1 Inputs (X): X0 to 8F Outputs (Y): Y30 to 10F
 - 2) Assignment of remote I/O station No.4 Inputs (X): X0 to BF Outputs (Y): Y80 to 19F
 - (c) Assignment of local stations
 In this example, inputs (X) and outputs (Y) are not required because there is empty area in the link relay (B) assignment. However, to simplify the explanation, 128 input (X) points and 128 output (Y) points are assigned.
 - Assignment of local station No.2 Local station No.2 uses the X/Y0 to X/Y17F range as I/O of host station. For the data link, the X/Y180 to X/Y7FF range can be used.
 - Assignment of local station No.3 Local station No.3 uses the X/Y0 to X/Y14F range as I/O of host station. For the data link, the X/Y150 to X/Y7FF range can be used.



Fig 7.19 Input (X) and output (Y) assignment example

(4) Link parameter setting exampleWhen the assignment of (1) to (3) is executed, set the link parameters as shown in the figure below.

				+ LINK +	k		M : B ↔ M : W ↔	ALL L: ALL L:	B 000-27F W 000-2BF
[MASTER	SLAVE	M → /	ALL L	W.D.T.	INTER-	M:W → M:W ←	ALL R: ALL R:	W 300-341 W 360-39F
ļ	MASIEN	STATIONS	В	w	10ms	10ms	M:Y → M:Y →	ALL L: ALL R:	X 680-77F Y 230-59F
	м	4	000-0FF	000-0FF	200	XXXX	M : X +	ALL L: ALL R:	X 200-4BF
ſ		M +		M → B	M + B	M -	• L/B	M	L/B
	L/H NO.	в	 w	w	w	Y	X/Y	x	Y/X
	R 1 L 2 L 3 R 4	100-17F 200-27F	100-17F 200-2BF	300-310 320-341	360-36F	230-30F 680-6FF 700-77F 480-59F	030-10F 200-27F 200-27F 080-19F	200-28F 600-67F 700-77F 400-48F	000-08F 280-2FF 200-27F 000-0BF
				-	380-39F		-	•	
		-	-	•		-	-	-	-
 	t L : LOCAI R : REMO	те Те		L		M : MASTI	ER L:L	OCAL R	: REMOTE

MELSECNET mode		MELSECNET		MELSECNET/B				
Operating Mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode		
Applicability		0			0			

MELSEC-A

7.8 Link Parameters in the MELSECNET I Mode

The following describes link parameter setting when the MELSECNET I mode is used. When setting the link parameters, consider the following points for assignment.

- (1) Determine whether to assign both the first and second half link parameters.
 - (a) If both the first and second half link parameters are assigned, up to 2048 bytes can be used for each station.

If the link range is less than 1024 bytes per station, setting of only the first half link parameters is required.

- (b) Using only the first half link parameters makes handshakes easy.
 Handshake processing is required when both first and second half link parameters are set. (For handshake processing, refer to Section 9.1(4).)
- (c) The setting range of the first half link parameters is B0 to FFF and W0 to FFF.
- (d) The range that available for the second half link parameters is the range assigned to the first half link parameters: (final device number) + 1.
 If 0 point is set for the first half link parameters, assignment of the second half link parameters can be started with B/W0.



- (2) Determine the link relay (B) and link register (W) assignment range for each master station and local station. (Refer to Section 7.6.2 and Section 7.6.3.)
- (3) If the number of link relay (B) points is insufficient, examine to substitute inputs (X) and outputs (Y) for (one-to-one) data communicated between the master station and a local.
- (4) Make sure that the number of link points per station is as follows. (Refer to Section 7.6.1.)
 - Master station's first half link parameters...... 1024 bytes or less

 - Local station's second half link parameters...... 1024 bytes or less

Link parameter setting example

The link parameter setting of the system configuration shown in Fig 7.20 is explained when the MELSECNET II mode is used.



Fig 7.20 Local system configuration example

- (1) Number of assigned points
 - (a) Link relays (B) 512 points and link registers (W) 512 points for the master station
 - (b) Link relays (B) 256 points and link registers (W) 256 points for the local station No.1
 - (c) Link relays (B) 512 points and link registers (W) 256 points for the local station No.2
 - (d) Link relays (B) 1024 points and link registers (W) 256 points for local station No.3
- (2) Examine the ranges to be assigned with the first and second half link parameters
 - (a) Because the number of master station link points is 1088 bytes

 $(512/8 + (512 \times 2) = 1088)$, both the first and second half link parameters are required.

To simplify this explanation, the number of link relays (B) and link registers (W) are halved and assigned with the first and second half link parameters each. Although it is possible to assign the link relays (B) only with the first half link parameters and the link registers (W) only with the second half link parameters, assign the link relays (B) used for handshake processing with both the first and second half link parameters.

(b) Only the first half link parameters are required for the local stations (No.1 to 3) because each station uses less than 1024 link points.



(3) Assignment of link relays (B)

Fig 7.21 Link relay (B) assignment example

(a) The empty area marked with *1 in the M/L area for the first half link parameters in Fig 7.21 assigned cannot be used as a substitute for internal relays (M) in the master station and all local stations.
 This is also true for any empty area in the M/L area assigned with the second

This is also true for any empty area in the M/L area assigned with the second half link parameters.

(b) The range that can be assigned with second half link parameters is the range assigned with the first half link parameters: (final device number) + 1 or later. (Example: If the range of B0 to FF is assigned to the first link parameters, "B100" or later can be assigned to the second half parameters)
In Fig 7.21, since the range of B000 to 8FF is assigned to the first half link parameters, B900 or later can be assigned to the second half link parameter.

				Ма	ster statio	on (M)	Ľ	ocal stati No. 1 (L1	on)	Loca	al station (L2)	No. 2	Loc	al station (L3)	1 No. 3
				Read	Write	Used as a data register (D) by the host station	Read	Write	Used as a data register (D) by the host station	Read	Write	Used as a data register (D) by the host station	Read	Write	Used as a data register (D) by the host station
5	w000	М]	0	0		0	[0			0		
f static	W100	*1													
st hal neter	W200	L1		0			0	0		0			0		
area the fir parar	W300	L2		0			0			0	0		0		
M/L for t	W400	L3		0	_		0			0			0	0	
	w500														
half						o			0			0			0
scond teter	- W800														<u> </u>
he se parai	W900	M		0	0		0			0			0		{
M/L for t link										1					
	WFFF	L,	J	L	I	1	I	L	I	l	[Read	: Read	l ling wo	rd data
												Write O	: Writir : Usab	ng word le rang	l datat e
L												· · · · · ·			

(4) Assignment of link registers (W)

Fig 7.22 Link register (W) assignment example

(a) The empty area marked with *1 in the M/L area for the first half link parameters in Fig 7.22 cannot be used as a substitute for data registers (D) in the master station and all local stations.

This is also true for any empty area in the M/L area assigned with the second half link parameters.

(b) The range that can be assigned with the second half link parameters is the range assigned with the first half link parameters: (final device number) + 1 or later. In Fig 7.22, since the range of W000 to 4FF is assigned to the first half link parameters, W500 or later can be assigned to the second half link parameters. (5) Assignment of inputs (X) and outputs (Y) In this system example, inputs (X) and outputs (Y) are not to be used because there is an empty area in the link relay (B) assignment. However, to simplify the explanation, 128 input (X) points and 128 output (Y) points each are assigned to local stations No.1 and No.3.

(Inputs (X) and outputs (Y) are not set for local station No.2.)

- (a) Range used for data link by master station
 The master station uses the X/Y0 to 14F range as I/O of host station.
 For data link, the X/Y150 to 7FF range can be used.
- (b) Input and output range used for data link by local station No.1 The local station No.1 uses the X/Y0 to 7F range as I/O of host station. For the data link, the X/Y80 to 1FF range can be used.
- (c) Input and output range used for data link by local station No.3 The local station No.3 uses the X/Y0 to FF range as I/O of host station. For the data link, the X/Y100 to 7FF range can be used.



Fig 7.23 Input (X) and output (Y) assignment example

(6) Link parameter setting

Set the link parameters assigned as (1) to (5) as follows:

(a) First half link parameters

		* ME	LSECNET II	MODE LINK	•	M : B1 ++ M : W1 ++	ALL L: ALL L:	B1 000-8FF W1 000-4FF
MAOTER	SLAVE	FIRST M	→ ALL L	W.D.T.	INTER-	M : B2 ↔ M : W2 ↔	ALL L:	B2 C00-CFF W2 800-8FF
MASIER	STATIONS	В	w	10ms	10ms	M:W M:W ∓	ALL R: ALL R:	W -
м	3	000-0FF	000-0FF	200	XXXX	$\begin{array}{ccc} M & Y & \rightarrow \\ M & Y & \rightarrow \\ M & Y & \rightarrow \end{array}$	ALL L: ALL R:	X 280-37F Y - Y 200.37F
						M : X ←	ALL R:	X -
L/R	FIRST I	M ← L			м	→ L	M	⊢ L
NO.	В	w			Y	x	×	Y
L 1 II	200-2FF	200-2FF			280-2FF	100-17F	200-27F	180-1FF
	500-4FF	400-4FF			300-37F	100-17F	300-37F	100-17F
	•			-		-	-	
	: [-		-			-	
	-	-	·			-	-	-
					M : MAS	TER L:	LOCAL I	R : REMOTE
		PRE	SS <ssn> T</ssn>	O SELECT 1	ST/2ND RAN	GE OF B/W		

(b) Second half link parameters

	MASTER	SLAVE PC STATIONS	SECOND	<pre>MELSECN M → ALL L W</pre>	W.D.T. FOR LINK 10ms	INK + INTER- MITTENT 10ms	$\begin{array}{c} M : B1 \\ M : W1 \\ H \\ M : B2 \\ H \\ M : W2 \\ H \\ M : W \\ H \\ M \\ H \\ \end{array}$	ALL L: ALL L: ALL L: ALL R: ALL R:	B1 000-8FF W1 000-4FF B2 C00-CFF W2 800-8FF W
	м	3	C00-CFF	800-8FF	200	XXXX	$\begin{array}{c} M : Y \rightarrow \\ M : Y \rightarrow \\ M : X \rightarrow \\ M : X \rightarrow \end{array}$	ALL L: ALL R: ALL L: ALL R:	X 280-37F Y - Y 200-37F X -
[L/B	SECOND	0 M ← L			M -	→ L	м -	-L
	ÑÓ.	В	w			Y	×	×	Y
	L 1 II L 2 II L 3 II	-	-				 		
			PRI	ESS <ssn></ssn>		M : MAST	IGE OF B/W		1 : HEMOTE

MELSECNET mode		MELSECNET			MELSECNET/B	
Operating Mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode
Applicability			0			0

MELSEC-A

7.9 Link Parameters in the MELSECNET I Composite Mode

The following describes the link parameter setting when the MELSECNET II composite mode is used.

The system configuration in the MELSECNET II composite mode includes the following three types.

- System consisting of a master station and local stations (Local system)
- System consisting of a master station and remote I/O stations (Remote I/O system)
- System consisting of a master station, local stations, and remote I/O stations (Local/ remote I/O system)

7.9.1 Local system assignment

When setting link parameters of a local system, consider the following points for assignment.

- Examine the link relay (B) and link register (W) range to be assigned with the first and second half link parameters for each MELSECNET II mode-compatible station.
 As for MELSECNET mode-compatible stations, only first half link parameters is set.
 - (a) The device range assigned with the first half link parameters can be read by the master station and all local stations. However, the range assigned with the second half link parameters can only be read by MELSECNET II mode-compatible stations.
 Examine the assignment range according to the station with which data communication will be executed.
 - (b) The setting range of the first half link parameters is B0 to 3FF and W0 to 3FF.
 - (c) The device range that can be assigned with the second half link parameters is the range assigned with the first half link parameters: "(final device number) + 1 " or later.

(Example: If the range of B0 to FF is assigned to the first link parameters, "B100" or later can be assigned to the second half parameters)

If 0 point is assigned with the first half link parameters, assignment with the second half link parameters can be started with B/W0.



(2) If the number of link relay (B) points is insufficient, examine to substitute inputs (X) and outputs (Y) for data communicated between the master station and a local station (one-to-one). (Refer to Section 7.6.4.)

- (3) Make sure that the number of link points per station is as follows. (Refer to Section 7.6.1.)
 - Master station first half link parameters...... 1024 bytes or less
 - Master station second half link parameters....... 1024 bytes or less

MELSECNET mode		MELSECNET			MELSECNET/B	
Operating Mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode
Applicability			0			0

MELSEC-A

7.9.2 Remote I/O system assignment

When setting link parameters of a remote I/O system, consider the following points for assignment.

- Determine what number of master station the device number of I/O module installed to the remote I/O station is assigned.
 The I/O numbers that can be assigned to a remote I/O station is the I/O numbers used for master station as I/O of host station. (Refer to Section 7.6.4.)
- (2) When a special function module is installed to a remote I/O station, determine the link register (W) assignment range (M/R area) to be used for reading/writing buffer memory (RFRP/RTOP instruction). The W0 to 3FF range can be used for the M/R area. (Refer to Section 7.6.3.)
 - (a) Divide the M/R area into the M ← R area and the M → R area to be assigned.
 When connecting more than one remote I/O station, assign an M ← R area and an M → R area to each remote I/O station.

For example, when connecting three remote I/O stations, divide the M \leftarrow R area and \rightarrow R area into three stations as shown in Fig 7.24.

W0 20	00 2	60 2	B0 3	00 3	60 3E	ю з	FF
	*1 R1	*2 R 2	*3 R 3	*1 R1	*2 R 2	*3 R 3	
	M +	- Rar	ea	M	→Rar	ea	•
*1: R₁ indicates the remote I/O station No. 1 *2: R₂ indicates the remote I/O station No. 2			M/R	area			*
*3: R₃ indicates the remote I/O station No. 3	1						1

Fig 7.24 Link register (W) assignment example

- (b) Consider the range used by the system when assigning the $M \rightarrow R$ area. The system uses the $M \rightarrow R$ area to execute RFRP/RTOP instructions.
 - 1) Number of points used by the system

The link register (W) 1 point is used for one special function module installed to a remote I/O station.

2) Range used by the system

The range used by the system is from the head device number of $M \rightarrow R$ area assigned from each remote I/O station to "number of use points - 1 ".

For example, when three special function modules are installed to remote I/O station No.2, the range W360 to 362 in the $M \rightarrow R$ area W360 to W3AF shown below is used by the system.



- (3) Make sure that the number of link points per station is as follows. (Refer to Section 7.6.1.)
 - Remote I/O station.....

(Inputs and outputs are 512 points of X/Y0 to 1FF)

POINT

 If the M → R area used by the system is used incorrectly in a user program, data cannot be read/written correctly, when the RFRP/RTOP instruction is executed.

512 bytes or less

(2) The number of link points can be reduced by mounting modules to a remote I/O station in groups of input modules, special function modules, and output modules as shown below. If the installation of I/O modules is changed from (a) to (b), a reduction of 16 input assignment points and 32 output assignment points can be achieved.



REMARK

The same concept is used to set the link parameters of a remote I/O system configured in the MELSECNET II composite mode and that of a remote I/O system configured in the MELSECNET mode.

MELSECNET mode		MELSECNET			MELSECNET/B	
Operating Mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode
Applicability			0			0

MELSEC-A

7.9.3 Local/remote I/O system assignment

When setting link parameters of a local/remote I/O system, consider the following points for assignment.

(1) Link relay assignment

For link relay (B), determine an assigned range for each of master and local stations. (Refer to Section 7.4.3.)

(a) Examine the link relay (B) assignment range with the first and second half link parameters for each MELSECNET II mode-compatible station. The device range assigned with the first half link parameters can be read by the master station and all local stations. However, the range assigned with the second half link parameters can only be read by MELSECNET II modecompatible stations. Determine the assignment range according to the station with which data

Determine the assignment range according to the station with which data communication will be executed.

- (b) The setting range of the first half link parameters is B0 to 3FF.
- (c) The device range that can be assigned with the second half link parameters is the range assigned with the first half link parameters: "(final device number) + 1" or later.

(Example: If the range of B0 to FF is assigned to the first link parameters, "B100" or later can be assigned to the second half parameters)

If 0 point is assigned with the first half link parameters, assignment with the second half link parameters can be started with B0.

(2) When a special function module is installed to a remote I/O station Link registers (W) are required for reading/writing buffer memory (RFRP/RTOP instruction).

Divide the link registers (W) in the W0 to 3FF range to be assigned with the first half link parameters into the M/L area (for communication between the master station and a local station) and the M/R area (for communication between the master station and a remote I/O station) and assign them.

(3) Link register assignment

For the M/L area of link registers (W), determine an assigned range for each of master and local stations.

- (a) Examine the link register (W) assignment range of the first and second half link parameters for each MELSECNET II mode-compatible station.
 The device range assigned with the first half link parameters can be read by the master station and all local stations. However, the range assigned with the second half link parameters can only be read by MELSECNET II mode-compatible stations.
 Determine the assignment range according to the station with which data communication will be executed.
- (b) The device range that can be assigned with the first half link parameters is W0 to 3FF.

MELSEC-A

If 0 point is assigned with the first half link parameters, assignment with the second half link parameters can be started with W0.



- (4) Divide the M/R area in link register (W) into the M ← R area and the M → R area and assign them.
 - Only the W0 to 3FF range can be used for the M/R area.
 - (a) When connecting more than one remote I/O station, assign an M \leftarrow R area and an M \rightarrow R area to each remote I/O station.

For example, when connecting two remote I/O stations, divide the M \leftarrow R area and M \rightarrow R area into two stations as illustrated below and assign them.

WO 1	00 18	80 2	00 30	0 340	380	3C0	3FF
м	L1	L4	Empty	*1 R2	*2 R3	*1 R2	*2 R3
L		1	-	M ◄ are	- R a	M → area	R
*1: R2 indicates the *2: R3 indicates the	e remote l e remote l	/O statio /O statio	n No. 2. n No. 3		M/R	area	

Fig 7.25 Link register (W) assignment example

- (b) Consider the range used by the system when assigning the $M \rightarrow R$ area. The system uses the $M \rightarrow R$ area to execute the RFRP/RTOP instructions.
 - Number of points used by the system The link register (W) 1 point is used for one special function module installed to a remote I/O station.
 - 2) Range used by the system

The range used by the system is from the head device number of $M \rightarrow R$ area assigned to each remote I/O station to "number of use points - 1 ". For example, when two special function modules are installed to remote I/O station No.3, the range W380 to 381 in the $M \rightarrow R$ area W380 to 38F is used by the system as shown below.



- (5) Assign the input (X) and output (Y) range used for data link to the I/O number used by the master station as I/O of host station.
 - (a) Divide the area used for data link into an M/R area (for communication between the master station and a remote I/O station) and an M/L area (for communication between the master station and a local station) to assign.
 - (b) The M/L area is used when the number of link relay (B) points is insufficient. Therefore, it is not necessary to assign the M/L area when there is sufficient number of link relay (B) points.
- (6) Make sure that the number of link points per station is as follows. (Refer to Section 7.6.1.)
 - Master station first half link parameters...... 1024 bytes or less
 - Master station second half link parameters... 1024 bytes or less

 - · Local station second half link parameters..... 1024 bytes or less

(Inputs and outputs are 512 points

of X/Y0 to 1FF)

POINT

- If the M → R area used by the system is used incorrectly in a user program, data cannot be read/written correctly, when the RFRP/RTOP instruction is executed.
- (2) The number of link points can be reduced by mounting modules to a remote I/O station in groups of input modules, special function modules, and output modules as shown below. If the installation of I/O modules is changed from (a) to (b), a reduction of 16 input assignment points and 32 output assignment points can be achieved.



REMARK

When all of the local stations are MELSECNET II mode-compatible data link module, the master station and all of the local stations can be assigned with the second half link parameters, which can simplify handshake processing mentioned in Section 9.1 (4).

MELSECNET mode		MELSECNET		MELSECNET/B				
Operating Mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode		
Applicability			0			0		

MELSEC-A

7.9.4 Link parameter setting example

The following describes the link parameter setting system configuration shown in Fig 7.26, using the MELSECNET II composite mode.



Fig 7.26 Local/remote I/O system configuration example

- (1) Number of assigned points
 - (a) Master station is the MELSECNET II mode-compatible station. Link relays (B) 256 points and link registers (W) 256 points are assigned with the first half link parameters to communicate with local station No.2/No.3 (all local stations). In addition, link relays (B) 256 points and link registers (W) 256 points are

assigned with the second half link parameters to communicate with local station No.3 (MELSECNET II compatible station).

- (b) Local station No.2 is MELSECNET mode-compatible station. Link relays (B) 256 points and link registers (W) 256 points are assigned with the first half link parameters.
- (c) Local station No.3 is MELSECNET II mode-compatible station. Link relays (B) 256 points and link registers (W) 128 points are assigned with the first half link parameters to communicate with master station/local station No.2 (master station and all local stations).
 In addition, link relays (R) 256 points and link registers (W) 256 points are

In addition, link relays (B) 256 points and link registers (W) 256 points are assigned with the second half link parameters to communicate with the master station (master MELSECNET II mode-compatible station).

- (2) Assignment of link relays (B)
 - (a) Assign 256 points for the master station, 256 points for local station No.2, and 256 points for local station No.3 with the first half link parameters. Assign 256 points for the master station and 256 points for the local station No.3 with the second half link parameters.



Fig 7.27 Link relay (B) assignment example

- (b) The empty area marked with *1 in the M/L area for the first/second half link parameters in Fig 7.27 cannot be used as a substitute for internal relays (M) in the master station and all local stations.
- (c) The device range that can be assigned with the second half link parameters is the range assigned with the first half link parameters: "(final device number allocated with the first half link parameters) + 1 " or later.
 In Fig 7.27, since the range of B000 to 37F is assigned to the first half link parameters, B380 or later can be assigned to the second half link parameters.
- (d) Since local station No.2 is a MELSECNET mode-compatible station, the B400 to FFF range cannot be used.

- (3) Assignment of link registers (W)
 - (a) Assignment for the master station and local stations (M/L area)
 Assign 256 points for the master station (M), 256 points for local station No.2 (L2), and 128 points for local station No.3 (L3) with the first half link parameters.
 Assign 256 points for the master station (M) and 256 points for local station No.3 (L3) with the second half link parameters.
 - (b) Assignment for remote I/O stations (M → R area, M ← R area) 16 points for the RTOP instruction and 16 points for the RFRP instruction are required for the remote I/O station No.1 (R1). No.1 requires 17 points (16 points (for RTOP instruction) + 1 point (for system)) for the M → R area since one special function module is installed.

32 points for the RTOP instruction and 32 points for the RFRP instruction are required for the remote I/O station No.4. No.4 requires 34 points (32 points (for RTOP instruction) + 2 points (for system)) for the $M \rightarrow R$ area since two special function modules are installed.

					Ma	ister s	tation (M)	Remote I No. 1	/O station (R1)	L	ocal s No. 2	tation (L2)		_ocal s No. 3	station (L3)	Remote I No. 4	/O station (R4)
alf					Read	Write	Used as a data register (D) by the host station	Read from master station	Write by master station	Read	Write	Used as a data register (D) by the host station	Read	Write	Used as a data register (D) by the host station	Read from master station	Write by master station
L It		wo	м		0	0				0			0				
er		W100	12		0		1			0	0		0				
a for amet		W200	*1														
are		W240	L3		0					0			0	0			
Σ Ξ		W2C0					0	L		ļ		0			0		
		W300	R1					ļ	<u> </u>			0		<u> </u>	•		
	M→R area	W311					<u> </u>		ļ			•		├	0		
		W320	R4		-		<u> </u>	0	ļ			<u> </u>			0		
area		W360					• •					0			0		
	M⊷R area	W370	<u>H1</u>	·	·							0			0		
		W380	<u>2</u> R4			<u>†</u>			-		<u> </u>	0			0	0	
	<u>*</u>	W3A0					0					0			0		
half		W400		<u></u>			0								o		
puo:	t	- W800	м		6	0						r	0			1.	
e sec		W900	*1		·		1			1				1		1	
or th leter		WA00	12			┢				+			0	0			1
irea f aram	.	- WB00			· ⊢					-			1	-			
M/L a link p		WEEE					•								• 0		
													R W C	ead /rite)	: Re : Wri : Us	ading wo iting wor able rang	ord data d data ge

Fig 7.28 Link register (W) assignment example

- (c) The empty areas marked with *1 (in the M/L area) and *2 (in the M ← area) in Fig. 7.28 cannot be used as a substitute for data register (D) in the master station and all local stations.
- (d) Since local station No.2 is a MELSECNET-compatible station, the B400 to FFF range cannot be used.

- (4) Assignment of inputs (X) and outputs (Y)
 - (a) Input and output range can be used for data link by master station The master station uses the X/Y0 to X/Y14F range as I/O of host station. For the data link, the X/Y150 to X/Y7FF range can be used.
 - (b) Assignment of remote I/O stations
 - Assignment of remote I/O station No.1 Assignment range of inputs (X): X0 to 8F Assignment range of outputs (Y): Y30 to 10F
 - Assignment of remote I/O station No.4 Assignment range of inputs (X): X0 to BF Assignment range of outputs (Y): Y80 to 19F
 - (c) Assignment of local stations
 In this example, inputs (X) and outputs (Y) are not required because there is empty area in the link relay (B) assignment. However, to simplify the explanation, 128 input (X) points and 128 output (Y) points are assigned.
 - Assignment of local station No.2 Local station No.2 uses the X/Y0 to 17F range as I/O station of host station. For the data link, the X/Y180 to 7FF range can be used.
 - Assignment of local station No.3 Local station No.3 uses the X/Y0 to 14F range as I/O station for host station. For the data link, the X/Y150 to 7FF range can be used.



Fig 7.29 Input (X) and output (Y) assignment example

(5) Link parameter setting

When the assignment of (1) to (4) is executed, set the link parameters as shown in the figure below.

	·····	*	M : B1 ↔ M : W1 ↔ M : B2 ↔ M : W2 ↔	ALL L: ALL L: ALL L:	B 1 000-37F W1 000-2BF B 2 C00-CFF W2 800-8FF						
MASTER	SLAVE	FIRST M	→ ALL L	W.D.T. FOBLINK		M:₩ →	ALL R:	N 300-341			
MASTEN	STATIONS	В	w	10ms	10ms	M:W ← M:Y →	ALL H:	X 680-39F X 680-77F			
М	4	000-0FF	000-0FF	200	XXXX	M:X +	ALL L:	Y 600-77F X 200-4BF			
			·····	· · · · · · · · · · · · · · · · · · ·							
L/R	FIRST M ← L		M → R	M ← R	M →	M → L/R		L/R			
NO.	В	w	W	w	Y	X/Y	x	Y/X			
R 1 L 2 L 3 II R 4	100-1FF 280-37F - - - -	100-1FF 240-2BF	300-310 320-341	360-36F 380-39F 	230-30F 680-6FF 700-77F 480-59F - - - - -	030-10F 200-27F 200-27F 080-19F - - -	200-28F 600-67F 700-77F 400-4BF - -	000-08F 280-2FF 200-27F 000-0BF - - - -			
M : MASTER L : LOCAL R : REMOTE											
R : REMC	TE ECNET-II (LO	CAL)	PRE	SS <ssn> TO</ssn>	SELECT 1S	T/2ND RANGI	E OF B/W				

(a) First half link parameters

		*	MELSECNE		DE LINK *	$\begin{array}{c} M : B1 \\ M : W1 \\ M : B2 \end{array}$	ALL L: E	B1 - V1 - B2 800-AFF
MACTER	SLAVE	SECOND N	M → ALL L	W.D.T.	INTER-	M : W2 ↔ M : W →	ALL L:V	V2 800-AFF
MASIER	STATIONS	В	w	10ms	10ms	$\begin{array}{ccc} M : W & \leftarrow \\ M : Y & \rightarrow \end{array}$	ALL R:V	Ý - K -
м	4	800-8FF	800-8FF	200	xxxx	M:Y → M:X ←	ALL R:	Υ - Υ -
				M:X ←	ALL H: 7	<u>·</u>		
L/R	SECON)M ← L	M → R	R M ← R I		L/R	M ←	L/R
NO.	В	w	w	w	Y	X/Y	x	Y/X
	A00-AFF - - -	A00-AFF		 - - - - - - - - - - - -		 	LOCAL R	
R : REMC * : MELS	TE ECNET-II (LO	CAL)	PRE	SS <ssn> TO</ssn>	SELECT 1S	T/2ND RANG	E OF B/W	

(b) Second half link parameters

MELSECNET mode		MELSECNET		MELSECNET/B				
Operating Mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode		
Applicability	0	0	0	0	0	0		

MELSEC-A

7.10 Three-Tier System Assignment

The following describes the link parameter setting required to use the three-tier system. The link parameter setting of the three-tier system is basically similar to that of the two-tier system described in Section 7.7 to 7.9.

Set the link parameters according to the operation mode used for each tier.

7.10.1 Common element

The following describes the common elements to know before setting link parameters for the third tier.

(1) The link relay (B) and link register (W) range assigned to the master station and local stations for the third tier is the device range assigned to the host station by link parameters for the second tier.

The link register (W) range which can be assigned to the remote I/O station for the third tier is the range assigned to the remote I/O station for the second tier and empty area.

Link relay (B) assigr	nment ran	ige				
Concernd Alex Note	0					<u>3FF</u>
parameters	м	L1/m	L2		Empty	
Link parameters of the master station (L1/m) for the third tier		L1/m l1 l2 Range usable as the M/L are	 ea			
Link register (W) ass	signment	range				
N	<u>vo</u>					<u>3FF</u>
parameters	М	L1/m	L2	M/R area	Empty	
Link parameters of the master station (L1/m) for the third tier		L1/m ℓ1 ℓ2 ← → → Range usabl as the M/L a	e rea	Rang as the	e usable e M/R area	

Fig 7.30 Assignment of link relays (B) and link registers (W)

(2) The input (X) and output (Y) range that can be assigned to the master station for the third tier can use the area starting from the I/O use range of host station. This is the same range as assignment range of the master station for the second tier. If inputs (X) and outputs (Y) are used to establish data link between the master station for the third tier and the master station for the second tier, the assignment range of the master station for the second tier, the assignment range of the master station for the third tier should exclude this range.

Link with inpu	ts (X) and outputs (Y) in only the		Link with inputs	(X) and outputs (Y) in	the second and				
third tier				third tiers						
X.Y000	Range used for I/O by the host station		-	X.Y000	Range used for I/O by the host station					
		Area			Range used for link with master station in 2nd tier					
		usable for third tier				Area usable for third tier				
X.Y7FF	L]	<u> </u>		X.Y7FF	L]					

Fig 7.31 Assignment of inputs (X) and outputs (Y) (When A3NCPU is used)

(3) Table 7.4 lists the nine types of combinations ((1) to (9)) depending on the operation modes set for the second tier and the third tier.

Third tier operation	Second tier operation mode								
mode	MELSECNET mode	MELSECNETT mode	MELSECNETI						
mode	MEEGEONET MODE	MELSECINETITINODE	composite mode						
MELSECNET mode	1)	4)	7)						
MELSECNETI mode	2)	5)	8)						
MELSECNETI	3)	6)	0)						
composite mode	5)	5)	3)						

Table 7.4 Operation mode combinations for second/third tier

- (4) When the MELSECNET I mode or MELSECNET I composite mode is selected, the link relay (B) and link register (W) range that can be assigned with the first and second half link parameters is as shown below.
 - (a) Assign the range assigned with the first half link parameters for the second tier with the first half link parameters for the third tier.

Assign the range assigned with the second half link parameters for the second tier with the second half link parameters for the third tier.

В/	wo											FF	F
Link parameters for the second tier	М	L	1/m		L2	L3	Empty	м	L1/m	۱	L2	L3	
B/	W0	[] }		1					1 1 1	1	 	FF	F
Link parameters for the master sta-	Empty	L1/m	Q1	٤2			Empty		L1/m 21	Q1	Em	ipty	
tion (L1/m) in the third tier	Range t first half	that can be assigned with the f link parameters for the third tier						Range th second h	hat can be as alf link para	ssigned meters	I with the for the th	ird tier	



(b) When the selected mode only provides one type of link parameter (the second tier is used for MELSECNET mode) or when the second half link parameters are not assigned, the device range that can be assigned with the second half link parameters is the range assigned with the first half link parameters: "(final device number) + 1" or later.

(Example: If the range of B0 to FF is assigned to the first half link parameters for second tier, "B100"or later can be assigned to the second link parameters for the third tier.)

B	/W0										FF
Link parameters for the second tier	Empty	м	L	1/m		L2		L3		Empty	
Link parameters for u			1 1 1		1				1		r
the master station			L1/m	Q1	l2				L1/m	l1	l2
(L1/m) in the third tier	Rai	t can be	assig	ned wit	Range that can be assigned with the						
	firs	t half lin	ik param	eters	for the	third tier	seco	ond half	link paramete	ers for th	e third ti

Fig 7.33 When only first half link parameters for the second tier are assigned

(5) In the three-tier system, link relays (B) and link registers (W) can be assigned to the ranges explained in (1) and (4).

In the MELSECNET data link system, range check of link parameters for second tier and link parameters for third tier is performed. This check is called the consistency check.

In the consistency check, the link relay (B) and link register (W) range assigned with the link parameters for the third tier are checked whether or not to be within the range assigned by the host station (master station for the third tier) using link parameters for the second tier. The check results are stored to M9235 and D9220 to 9223 in the master station for the second tier and to M9270 in the master station for the third tier.

(a) The consistency check is executed using the parameters shown in Table 7.5 according to the operation mode combinations of the second and third tiers. For example, if the operation mode of the second tier is the MELSECNET II composite mode and the operation mode of the third tier is the MELSECNET II mode, the first half link parameters for the second tier is compared with the first half link parameters for the third tier and the second half link parameters for the second tier are compared with the second half link parameters for the third tier.

Consistency check is executed
 Consistency check is not executed

		Se	econd tier operation	on mode and link	parameters		
Third tier operation	on mode and link	MELSECNET mode link	MELSECN	ET ∏ mode	MELSECNET I composite mod		
paran	neters		First half link	Second half	First half link	Second half link	
		parameter	parameters	link parameters	parameters	parameters	
MELSECNET mod	de link parameters	0	0	×	0	×	
MELSECNET I	First half link parameters	0	0	×	0	×	
mode	Second half link parameters	×	×	0	×	0	
MELSECNET I	First half link parameters	0	0	×	0	×	
composite mode	Second half link parameters	×	×	0	×	0	

Table 7.5 Consistency check of link parameter settings

- (b) The following consistency check is executed when there is difference between the number of link parameters set for the second tier and the number of link parameters set for the third tier.
 - 1) When two types of link parameters are set for the second tier and one type of link parameters is set for the third tier:

The consistency check is executed on the first half link parameters for the second tier and the first half link parameters for the third tier (including the MELSECNET mode).

The consistency check is not executed for the second half link parameters for the second tier.

2) When one type of link parameters is set for the second tier and two types of link parameters are set for the third tier:

The consistency check is executed on the first half link parameters (including the MELSECNET mode) for the second tier and the first half link parameters for the third tier.

The second half link parameters for the third tier are checked whether or not to begin after the final device number assigned with the first half link parameters for the second tier.

(c) When the range of link relays (B) and link registers (W) is extended (turning M9208 and M9209 ON to use), the consistency check will not be executed. Make sure that the link parameters assigned to the third tier are not also assigned to the second tier.

Вл	NO		FFF							
Link parameters for the second tier	м	L1/m	L2	L3		м	L1/m	L2	L3	
Link parameters for the third tier	Device first ha	e range as	ssigned	with the s s Device half link	range assigned with parameters for the t	bevice second the first hird tier	range as	ssigned parame De se the	with the sters	ge assigned with the f link parameters for er

Fig 7.34 Assignment when link relay (B) and link register (W) range is extended

MELSECNET mode		MELSECNET		MELSECNET/B			
Operating Mode	MELSECNET mode	MELSECNET II MELSECNET II composite mode		MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	
Applicability	0			0			

MELSEC-A

7.10.2 Using the MELSECNET mode in the second tier

The link relay (B) and link register (W) range which can be assigned to the third tier is described per operation mode used in the third tier. Since the assignment of M/L area for link relay (B) is the same as that of M/L area for link register (W), the assignment of the link register (W) is only explained.

The link assignment range of input (X) and output (Y) is the same regardless of combinations of operation modes. Refer to Section 7.10.1 (5).

- (1) MELSECNET mode used in the third tier
 - (a) The range assigned to the master station for the third tier with the link parameters for the second tier is used for the M/L area for the third tier.
 - (b) The M/R area and the empty area of the second tier is used for the M/R area of the third tier.



Fig 7.35 Assignment when the MELSECNET mode is used for the third tier

- (2) MELSECNET I mode used in the third tier
 - (a) The range assigned to the master station for the third tier with the link parameters for the second tier is used for the first half link parameters for the third tier.
 - (b) The empty area that begins after the M/L area assigned with the link parameters for the second tier and the M/R area are used for the second half link parameters for the third tier.



Fig 7.36 Assignment when the MELSECNET II mode is used for the third tier

- (3) MELSECNET II composite mode used in the third tier
 - (a) First half link parameters for the third tier
 - 1) The range assigned to the master station for the third tier with the link parameters for the second tier is used for the M/L area.
 - 2) The M/R area in the second tier and the empty area in the W0 to 3FF range is used for the M/R area.
 - (b) The second half link parameters for the third tier can handle the empty area that begins after the M/L area using the link parameters for the second tier. However, exclude the range used as M/R area in the first half link parameters for the third tier.



Fig 7.37 Assignment when the MELSECNET ${\rm I\!I}$ composite mode is used for the third tier

POINT	

When connecting a remote I/O station to the third tier, consider the range to be assigned to the M/R area for the third tier with the link parameters for the second tier and assign it.

Even when the MELSECNET composite mode is used for the third tier, the range which can be used as M/R area is the W0 to 3FF range. Therefore, the similar attention should be paid.

	MELSECNET		MELSECNET/B			
MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	
	0			0		
	MELSECNET	MELSECNET Melsecnet I mode	MELSECNET I MELSECNET II MELSECNET II mode O MELSECNET II	MELSECNET I MELSECNET I mode MELSECNET I mode O	MELSECNET MELSECNET I MELSECNET I mode MELSECNET I MELSECNET I mode MELSECNET I mode MELSECNET I MELSECNET I	

MELSEC-A

7.10.3 Using the MELSECNET ${\rm I\!I}$ mode in the second tier

The link relay (B) and link register (W) range which can be assigned to the third tier is described per operation mode used in the third tier. Since the assignment of M/L area for link relay (B) is the same as that of M/L area for link register (W), the assignment of the link register (W) is only explained.

The link assignment range of input (X) and output (Y) is the same regardless of combinations of operation modes. Refer to Section 7.10.1 (5).

- (1) MELSECNET mode used in the third tier
 - (a) The range assigned to the master station for the third tier with the first half link parameters for the second tier is used for the M/L area for the third tier. The range assigned to the master station for the third tier with the second half link parameters for the second tier cannot be used.
 - (b) The empty area within the W0 to 3FF range assigned with link parameters for the second tier is used for the M/R area for the third tier.

	M/L area assigned with the first half link parameters				-l le	M/L area assigned with the second half link parameters				
	wo				400					FFF
Link parameters for the second tier	Empty	м	L1/n	n	L2	Empty	М	L1/m	L2	Empty
link naramatara	wo				3FF					
for the master station (L1/m) in the third tier			L1/m 🛛	1 22						
Usa	ble as the M/R a	area	Usable as	the M/L	area					

Fig 7.38 Assignment when the MELSECNET mode is used for the third tier

- (2) MELSECNET II mode used in the third tier
 - (a) The range assigned to the master station for the third tier with the first half link parameters for the second tier is used for the first half link parameters for the third tier.
 - (b) The range assigned to the master station for the second tier with the second half link parameters for the second tier is used for the second half link parameters for the third tier. If no area is set to the second half link parameters, use the area that begins after the M/L area assigned to the first half link parameters for the second tier.



Fig 7.39 Assignment when the MELSECNET ${\rm I\!I}$ mode is used for the third tier

- (3) MELSECNET II composite mode used in the third tier
 - (a) First half link parameters for the third tier
 - 1) The range assigned to the master station for the third tier with the first half link parameters for the second tier is used for the M/L area.
 - 2) The empty area within the W0 to 3FF range assigned with link parameters for the second tier is used for the M/R area.
 - (b) The area assigned to the master station for the third tier with the second half link parameters for the second tier is used for the second half link parameters for the third tier.

	WO					FFF			
Link parameters for the second tier	Empty	М	L1/m	L2	Empty	M	L1/m	L2	Empty
	wo						1 1 1	1 1 1	FFF
Link parameters for the master station (L1/m) in			L1/m 21 22				L1/m 21 22		
Usable as for the sec	the M/R area cond half link para	L meters f	Usable as the M/R area for the first half link parameters for the second half link parameters					meters	

Fig 7.40 Assignment when the MELSECNET ${\rm I\!I}$ composite mode is used for the third tier

POINT

When the MELSECNET mode or the MELSECNET II composite mode is selected as the operation mode for the third tier, consider the following points when setting link parameters for the second tier.

- (1) The device range assigned to the master station for the third tier is within the B0 to 3FF range and the W0 to 3FF range. If the B400 to FFF range or W400 to FFF range is assigned to the master station, the area cannot be used for the first half link parameters (including the MELSECNET mode link parameter) for the third tier.
- (2) When an M/R area is required for the third tier, provide an empty area in the W0 to 3FF range with the link parameters for the second tier.

MELSECNET mode		MELSECNET		MELSECNET/B			
Operating Mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	
Applicability			0			0	

MELSEC-A

7.10.4 Using the MELSECNET II composite mode in the second tier

The link relay (B) and link register (W) range which can be assigned to the third tier is described per operation mode used in the third tier. Since the assignment of M/L area for link relay (B) is the same as that of M/L area for link register (W), the assignment of the link register (W) is only explained.

The link assignment range of input (X) and output (Y) is the same regardless of combinations of operation modes. Refer to Section 7.10.1 (5).

- (1) MELSECNET mode used in the third tier
 - (a) The range assigned to the master station for the third tier with the first half link parameters for the second tier is used for the M/L area for the third tier.
 - (b) The empty area within the W0 to 3FF range assigned with link parameters for the second tier is used for the M/R area for the third tier.





- (2) MELSECNET I mode used in the third tier
 - (a) The range assigned to the master station for the third tier with the first half link parameters for the second tier is used for the first half link parameters for the third tier.
 - (b) The range assigned to the master station for the third tier with the second half link parameters for the second tier is used for the second half link parameters for the third tier.



Fig 7.42 Assignment when the MELSECNET I mode is used for the third tier

- (3) MELSECNET II composite mode used in the third tier
 - (a) First half link parameters for the third tier
 - 1) The range assigned to the master station for the third tier with the first half link parameters for the second tier is used for the M/L area.
 - 2) The M/R area for the second tier and an empty area in the W0 to 3FF range is used for the M/R area.
 - (b) The range assigned to the master station for the third tier with the second half link parameters for the second tier is used for the second half link parameters for the third tier.



Fig 7.43 Assignment when the MELSECNET composite mode is used for the third tier

POINT

When the MELSECNET mode or the MELSECNET I composite mode is selected as the operation mode for the third tier, consider the following points when assigning device ranges.

- (1) The device range assigned to the master station for the third tier is within the B0 to 3FF range and the W0 to 3FF range. If the B400 to FFF range or W400 to FFF range is assigned to the master station for the third tier, the area cannot be used for the first half link parameters (including the MELSECNET mode link parameter) for the third tier.
- (2) When the M/R area is required for the third tier, either provide an empty area in the W0 to 3FF range with the link parameter for the second tier or use the M/R area for the second tier.

MELSECNET mode		MELSECNET		MELSECNET/B			
Operating Mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	
Applicability	0	0	0	0	0	0	

MELSEC-A

7.10.5 Link parameter setting example

The following describes the link parameters for the third tier, exemplifying the system configuration shown in Fig 7.44. (The link parameter setting for the second tier is the same as explained in Section 7.7 to 7.9.)

The number of assignment points for each station is the number of points shown in Table 7.6.



Fig 7.44 Three-tier system configuration example
			First half lin		Second half link parameter			
		M⇔L		M→P	M←P	B	W	
		В	W	IVI / IX	IVI NI N	D	**	
М		256	256			256	256	
R	1			34	32			
L	2	128	128			128 128		
	m	128	128			128	128	
1.3/m	l 1	128	128			128	128	
r2				34	32			
	l 3	128	128					
R4				34	32			

Table 7.6 Number of assignment point for each station

- (1) Checking the operation mode to be used
 - (a) Operation mode of the second tier: MELSECNET II composite mode This is because the slave stations (local stations and remote I/O stations) are connected to either MELSECNET mode-compatible station or MELSECNET II mode-compatible station.
 - (b) Operation mode of the third tier: MELSECNET II composite mode This is because the slave stations (local stations and remote I/O stations) are connected to either MELSECNET mode-compatible station or MELSECNET II mode-compatible station.
- (2) Checking the range assigned to the master station for the third tier with the link parameters for the second tier
 - (a) According to the number of assignment point in Table 7.6, 384 points for link relay (B) and 384 points for Link registers (W) are required for the first half link parameters, and 256 points for link relay (B) and 256 points for link register (W) are required for the second half link parameters.
 - (b) 34 points for M → R area and 32 points for M ← R area are required for the M/R area.

At least 66 points for the empty area of link register (W) is required for the assignment of the M/R area for the third tier with the first half link parameters for the second tier.

In the system configuration shown in Fig 7.44, since M/R area for the second tier also has 132 points, this range can also be used for assignment.

(3) Assignment of link relays (B)



Fig 7.45 Link relays (B) assignment example

Loca	l station No	2(L2)		Link	using local st	ation No. 3(L	L3/m) as the	master static	on for the thi	rd tier	
2000		_()	Mas	ter station (L	.3/m)	Local	station No.	1 (ℓ1)	Local	station No.	3 (ℓ3)
Coil	Contact	Used as internal relays (M) in the master station	Coil	Contact	Used as internal relays (M) in the master station	Coil	Contact	Used as internal relays (M) in the master station	Coil	Contact	Used as internal relays (M) in the master station
	0			0			0			0	
0	0			0				0			
	0		0	0			0			0	
	0			0		0	0			0	
	0			0			0		0	0	
		0			0			0			0
		0			0			0			
	0			0			0				
0	0			0				0			
	0		0	0			0				
	0			0		0	0				
		0			0			0			
		·			Coil: Contact: O:	Performs (Reads ON Usable rar	DN/OFF cc I/OFF data	ntrol of link using conta	relays (B) acts of link	relays (B).	

Fia	7.45	Link	relavs	(B)	assignment	example	(Continued)
9	1.40	_	roiuyo	ربا	assignment	example	(Continueu)

(4) Assignment of link registers (W)



Fig 7.46 Link registers (W) assignment example

(a) An empty area in the W0 to 5F range is used to assign an M/R area for the third tier with the link parameters for the second tier.
 The assignment can also be made by using the M/R area for the second tier W360 to 3E3.

Lo	ocal stat	ion		Link	using lo	cal statio	n No. 3 (l	_3/m) as	the maste	r station fo	or the thir	d tier		Remo	ote I/0
	No. 2(L2	2)	Maste	r station	(L3/m)	Local s	tation No	. 1 (ℓ1)	Remote I No. 2	/O station 2 (r2)	Local station No. 3(ℓ 1)		. 3(ℓ1)	station No. 4 (R4)	
*1	*2	*3	*1	*2	*3	*1	*2	*3	*4	*5	*1	*2	*3	*4	*5
		0						0		0			0		
		0						0	0				0		
		0			0			0					0		
0			0			0					0				
 0	0		0					0					0		
 0			0	0		0					0				
0			0			0	0				0	_			
0		-	0		-	0		-			0	0	-		
		0			0			0					0		
		0			0	-		0	-				0		0
		0			0			0					0	-	
	-	0			0			0					0	0	
		0			0			0					0		
		0			0			0							
0			0			0									
0	0		0					0							
0			0	0		0		_							
0			0			0	0								
		0			0			0							

*1 : Read Reading word data

*2 : Write Writing word data

*3 : Used as a data register (D) in the master station

*4 : Read from master station

*5 : Write from master station

O: Usable range

Fig 7.46 Link registers (W) assignment example (Continued)

- (5) Inputs (X) and outputs (Y)
 - (a) A memory map for the assignment example of inputs and outputs is shown in Fig 7.47.



Fig 7.47 Inputs/outputs assignment example

(6) Link parameter setting

The following shows the link parameters assigned as (1) to (4).

(a) First half link parameters for the second tier

			* MI	ELSECNET I	I MULTI MODI	ELINK +	M:B1 ↔ M:W1 ↔ M:B2 ↔	ALL L: ALL L: ALL L:	B1 000-2FF W1 060-35F B2 500-77F	
	MACTED	SLAVE	FIRST M	FIRST M → ALL L		INTER-	M : W2 ↔ M : W →	ALL L: ALL R:	W2 500-77F W 360-3A3	
	MASIER	STATIONS	В	w	10ms	10ms	M:W ← M:Y →	ALL R:	W 3A4-3E3 X 250-47F	
	м	4	000-0FF	060-15F	200	XXXX	$\begin{array}{c} M : Y \rightarrow \\ M : X \leftarrow \\ M \cdot Y \end{array}$	ALL R:	Y 4E0-7CF Y 1A0-36F	
						1	M : A -	ALL HI	× 460-70F	
	L/R	FIRST	M ← L 👘	M → R	M ← R	M →	▶ L/R	M ← L/R		
	NO.	В	w	w	w	Y	X/Y	×	Y/X	
	R 1 L 2 II L 3 II R 4	100-17F 180-2FF - -	160-1DF 1E0-35F	360-381 	3A4-3C3 	4E0-5DF 390-47F 250-36F 6D0-7CF - - -	030-12F 250-33F 1A0-2BF 080-17F - - - -	4B0-54F 1A0-25F 2A0-36F 650-70F - - - -	000-09F 210-2CF 1F0-2BF 000-0BF	
L F *	M : MASTER L : LOCAL R : REMOTE L : LOCAL PRESS <ssn> TO SELECT 1ST/2ND RANGE OF B/W * : MELSECNET-II (LOCAL)</ssn>									

WAATER	SLAVE	SECOND I	/ → ALL L	W.D.T.	INTER-	M : W2 ↔ M : W →	ALL L: ALL R:	W2 500-7 W 360-3	
MASTER	STATIONS	В	w	10ms	10ms	$\begin{array}{ccc} M : W & \leftarrow \\ M : Y & \rightarrow \end{array}$	ALL R: ALL L:	W 3A4-3 X 250-4	
м	4	500-5FF	500-5FF	200	xxxx	M:Y → M:X ← M:X ←	ALL R: ALL L: ALL R:	Y 4E0-7 Y 1A0-3 X 4B0-7	
L/R	SECON) M ← L	M←L M→R M←R			► L/R	M +-	M ← L/R	
ÑÔ.	В	w	w	w	Y	X/Y	х	Y/X	
R 1 L 2 II L 3 II R 4	600-67F 680-77F 	600-67F 680-77F - - -			······································	 - - - -	 - 		
t LOCAL REMOT	Έ		PR	ESS <ssn> T</ssn>	M : MAS	TER L : ST/2ND RANG	LOCAL SE OF B/W	R : REMO	

(b) Second half link parameters for the second tier

			* M	ELSECNET		E LINK +	M : B1 ↔ M : W1 ↔ M : B2 ↔	ALL L: ALL L:	B1 180-2FF W1 1E0-35F B2 680-77F		
	MAGTED	SLAVE	FIRST M	→ ALL L W.D.T.		INTER-	M:W2 ↔ M:W →	ALL L: ALL R:	W2 680-7FF W 000-021		
	MASIER	STATIONS	в	w	10ms	10ms	M:W ← M:Y →	ALL R: ALL L:	W 022-041 X 3B0-57F		
	м	3	180-1FF	1E0-25F	200	XXXX	M:Y → M:X ←	ALL R:	Y 600-78F Y 300-57F		
								ALL N.			
	$I/B \qquad FIRST M \leftarrow L \qquad M \rightarrow R \qquad M \leftarrow R \qquad M \rightarrow L/R \qquad M \leftarrow L/R$										
	NO.	В	w	w	w	Y	X/Y	х	Y/X		
	L 1 II R 2 L 3	200-27F 280-2FF	260-2DF 2E0-35F	000-021	022-041	3B0-4AF 600-78F 4B0-57F	250-34F 060-1EF 100-1CF	300-41F 5A0-65F 4B0-57F	1A0-2BF 000-0BF 100-1CF		
		-	-	-	-	-	-	-			
		-	-		1	1 -		-			
			•	-	-	-	-	•			
		-	-	-		-	-	-	-		
L F	M : MASTER L : LOCAL R : REMOTE L : LOCAL R : REMOTE PRESS <ssn> TO SELECT 1ST/2ND RANGE OF B/W * : MELSECNET-II (LOCAL)</ssn>										

(c) First half link parameters for the third tier

(d) Second half link parameters for the third tie	r
---	---

• .		* M	ELSECNET		ELINK +	M : B1 ↔ M : W1 ↔	ALL L: ALL L:	B1 180-2EF W1 1E0-35F		
MASTER	SLAVE	SECOND	/ → ALL L	W.D.T. FOR LINK	INTER-	M : B2 ↔ M : W2 ↔ M : W →	ALL L:	B2 680-77F W2 680-7FF W 000-021		
MAGTER	STATIONS	В	w	10ms	10ms	M:W ← M:Y →	ALL R:	W 022-041 X 3B0-57F		
м	3	680-6FF	680-6FF	200	XXXX	M:Y → M:X ←	ALL R:	Y 600-78F Y 300-57F		
						M:X ←	ALL R:	X 5A0-65F		
L/R	SECONE) M L	M → R	M ← R	M ← R M → L/R		M +	L/R		
NO.	В	W.	w	w	Y	X/Y	x	Y/X		
L 1 II	700-77F	700-7FF								
	-	-	-	-	-	-	-	-		
	-	-	•				-			
		-				-	-			
	-	-	•	-	-	-	-	-		
M : MASTER L : LOCAL R : REMOTE										
+ : REMO	R : REMOTE PRESS <ssn> TO SELECT 1ST/2ND RANGE OF B/W * : MELSECNET-II (LOCAL)</ssn>									

7. DATA LINK SETTINGS

MELSECNET mode		MELSECNET			MELSECNET/B	
Operating Mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode
Applicability	0		0	0		0

MELSEC-A

7.11 Assignment of Inputs and Outputs to the Master Station in a Remote I/O System

When a remote I/O system is configured with the MELSECNET data link system, there are restrictions on the I/O assignment for the master station.

The following describes the I/O assignment of master station which consists of a remote I/O station.

In case of the master station/local station which links to local stations only, I/O assignment can be executed in the same manner as for an independent system.

REMARK

I/O addresses are automatically assigned by the programmable controller CPU. A peripheral device is not particularly required for I/O assignment. However, I/O assignment with a peripheral device enables followings.

- Saving I/O points (16 points) occupied by an empty slot.
- Reserving number of I/O points (32, 48, or 64 points) at an empty area for the system expansion in the future.

7.11.1 I/O assignment restrictions

(1) I/O assignment must be made from the head address (X/Y0) to the final address assigned to the remote I/O station.

Failure to complete I/O assignment up to the final address will cause an error in the master station.

The I/O range to be assigned differs depending on the order of the M/L area and the M/R area set by the link parameters.

- (a) If the M/L area is assigned after the M/R area, it is not necessary to assign inputs and outputs to the local station setting range. (Refer to Fig 7.48 (a).)
- (b) If the M/L area is assigned after the M/R area, it is necessary to assign inputs and outputs to the local station setting range. (Refer to Fig 7.48 (b).)



Fig 7.48 I/O assignment setting range

REMARK

In the I/O assignment setting, the device range is set with the assumption that the input modules or output modules are installed in the area of "M/L area" or that empty slots exist in the "empty area (range marked by *)" Fig 7.48.

(2) As for I/O assignment of remote I/O station, a slot in which a module is installed cannot be assigned as an empty slot (S1: 0 point, S2: 16 points, S3: 32 points, S4: 48 points, S5: 64 points). If assigned to an empty slot, a "UNIT VERIFY ERROR" will occur.

	*1	*1	*1	*2	*1
AJ72P25	Input module 16 points	Output module 32 points	Special function module 32 points	Empty	Input module 16 points

• Slots marked with "1" cannot be assigned as an empty slot (S1, S2, S3, S4, or S5) because an input or output module is loaded.

• The slot marked with "*2" can be assigned as an empty slot.

(3) If slot 0 in a remote I/O station is empty, at least 16 points (S2, S3, S4, S5) must be assigned to an empty slot.

If S1 is set for an empty slot, a "UNIT VERIFY ERROR" will occur.

AJ72P25	Empty	Input module 16 points	Output module 32 points	Special function module 32 points	Input module 16 points
	L	1			

The empty slot cannot be set as a 0 point empty slot.

(4) When assigning inputs and outputs to a special function module, set the number of points of the module actually installed. If the wrong number of points is set, and the RFRP or RTOP instruction is executed, an error will occur.

POINT

I/O assignment cannot be used to change the number of I/O points for an input/ output module connected to the A0J2P25(S3)/R25 (compact type remote I/O station module).

Assign the same number of I/O points that is assigned to a remote I/O station configured with the A0J2P25(S3)/R25.

7. DATA LINK SETTINGS

MELSECNET mode		MELSECNET		MELSECNET/B			
Operating Mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	
Applicability	0		0	0		0	

MELSEC-A

7.11.2 I/O assignment example

The following describes the I/O assignment of I/O number for a remote I/O station by GPP. The I/O assignment of remote I/O station is set on the "I/O LOCATING" screen of GPP parameters and is store to the master station.

The following describes the procedure for setting "0 point" for an "empty slot" part (the shaded **WW** slots in Fig 7.49) in the master station and remote I/O stations, exemplifying the system shown in Fig 7.49.



Fig 7.49 System example

The I/O assignment is set from the head address (X/Y0) to the final address of M/R area. The I/O assignment range varies depending on the ranges assigned for the M/L area and the M/R area.

Refer to the I/O assignment example in Fig 7.49.

- (1) When the M/L area is assigned after the M/R area When the M/L area is assigned after the M/R area with the link parameters, the I/O assignment is as shown in the following example.
 - (a) Link parameter assignment example
 - Fig 7.50 shows the link parameter I/O assignment.



Fig 7.50 Assignment example

						., -										VACANC	V(e)
SLT NO.	I/O UNT	SLT NO.	I/O UNT	SLT NO.	I/O UNT	SLT NO.	I/O UNT	SLT NO.	I/O UNT	SLT NO.	I/O UNT	SLT NO.	I/O UNT	SLT NO.	I/O UNT	1 : 0 2 : 16 3 : 32	PT. PT. PT.
01234567890112345 10112345	X16 X16 X16 X16 X16 X16 Y32 Y16 S32 Y16 S32 Y16 S32 Y16 S32 S0	16 17 18 19 20 21 22 23 24 25 26 27 28 29 30 31	Y16 Y16 Y16 Y16 Y16 Y16 S64 S16 X16 X32 X16 Y32	32 33 34 35 36 37 38 39 40 41 42 43 44 45 46 47	Y16 F32 X16 Y16 S0 Y16 Y16 Y16 Y64 X32 X36 X16	48 49 50 512 53 54 55 56 57 58 59 60 61 62 63	F32 S 0 S 0 Y16 Y32 Y16 Y32 Y32 Y32 Y32 Y32 Y32	64 65 66 67 68 69 70 71 72 73 74 75 75 76 77 78 79		80 81 82 83 84 85 86 87 88 89 90 91 92 93 94 95		96 97 98 990 100 101 102 103 104 105 106 107 108 109 110		112 113 114 115 116 117 118 119 120 121 122 123 124 125 126 127		3 : 32 4 : 48 5 : 64 X(X) 6 : 16 7 : 32 8 : 48 9 : 64 Y(Y) A : 16 B : 32 C : 48 D : 64 S-UNIT(F E : 16 F : 32 G : 48 G : 48 C : 48	PT. PT. PT. PT. PT. PT. PT. PT. PT. PT.

(b) I/O assignment example

* I/O ALLOCATION *

PRESS <END>, WHEN SET

Slot numbers

0 to 23......Assignment of I/O modules in the master station

24 to 26......Range marked with an asterisk (the 170 to 1FF range)

- 27 to 42.....Assignment of remote I/O station No.1
 - 43......"Empty" area between remote I/O stations No.1 and

No.3 (shaded Zarea)

44 to 59......Assignment of remote I/O station No.3

- (2) When the M/R area is assigned after the M/L area When the M/R area is assigned after the M/L area with the link parameters, the I/O assignment is as shown in the following example.
 - (a) Assignment example by link parameter
 - Fig 7.51 shows the I/O assignment by link parameters.



Fig 7.51 Assignment example

					* I	/0 AL	LOCAT	ION	*									((0))
SLT NO.	I/O UNT	SLT NO.	I/O UNT	SLT NO.	I/O UNT	SLT NO.	I/O UNT	SLT NO.	I/O UNT	SLT NO.	1/O UNT	SLT NO.	I/O UNT	SLT NO.	I/O UNT	VAC 1 2 3	ANC1 : 0 : 16 : 32	PT. PT.
0 1 2	X16 X16 X16	16 17 18	Y16 Y16 Y16	32 33 34	S64 S64 S64	48 49 50	Y32 Y16 Y16	64 65 66	Y16 S 0 S 0	80 81 82		96 97 98		112 113 114		4 5 X(X)	: 48 : 64	PT. PT.
3 4 5	X16 X16 X16	19 20 21	Y16 Y16 Y16	35 36 37	S64 S64 S64	51 52 53	Y32 Y32 Y16 Y22	67 68 69 70	Y16 Y16 Y16 Y64	83 84 85		99 100 101		115 116 117		6 7 8 9	: 16 : 32 : 48 : 64	PT. PT. PT.
0 7 8 9	X16 X16 S 0	23 24 25	Y16 S64 S64	39 40 41	X32 X32 X16	55 56 57	X16 X32 X16	71 72 73 74	104	87 88 89		103 104 105		119 120 121		Y(Y) A B C	: 16 : 32 : 48	PT. PT. PT.
10 11 12 13	X32 X32 Y16 S 0	26 27 28 29	S64 S64 S64 S64	42 43 44 45	F32 S 0 S 0	50 59 60 61	Y 16 Y 16 F 32	75 76 77		90 91 92 93		108 107 108 109		123 124 125		D S-U	: 64 NIT(F) : 16	PT.
14 15	S 0 S 0	30 31	S64 S64	46 47	S 0 Y 16	62 63	X16 Y16	78		94 95		110		126 127		G H	: 48 : 64	PT. PT.

(b) I/O assignment example

PRESS <END>, WHEN SET

Slot numbers

nbers 0 to 23......Assignment of I/O modules in the master station

- 24 to 38......Range marked with an asterisk and "local station setting range" (the 170 to 4FF range)
- 39 to 54......Assignment of remote I/O station No.3
- 55 to 70......Assignment of remote I/O station No.1

8. PROCEDURES TO OPERATION

MELSECNET mode		MELSECNET		MELSECNET/B			
Operating Mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	
Applicability	0	0	0	0	0	0	

MELSEC-A

8 PROCEDURES TO OPERATION

This chapter describes procedures for data link and startup, station number setting, and precautions for wiring.

8.1 Preparatory Steps before Operation

This section describes a procedure for data link.



MELSECNET mode	MELSECNET			MELSECNET/B			
Operating Mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	
Applicability	0	0	0				

8.2 Setting the Link Module Station Numbers

This section describes the station number setting for link modules and precautions for it.

8.2.1 Setting the link module station numbers in the MELSECNET data link system

Assign station numbers from the master station (set "00" to the master station) in ascending order to the forward loop direction. The settable maximum station number is "64".

In a three-tier system, assign station numbers from the master stations of each tier (set "00" to the master stations) in ascending order to the forward loop direction. The settable maximum station number is "64".



Fig 8.1 Setting link module station numbers

REMARK

For station number setting of the link modules, refer to the manual for each link module.

Precautions for setting the link module station numbers

The following shows the precautions for station number setting.

Station numbers have to be consecutive.
 Station numbers cannot be skipped as Fig 8.2.

If station numbers are not set consecutively, the time taken for the system to switch to the loopback mode when the slave station is powered off is increased.

Therefore, the loopback processing is not performed within the monitoring time set in link parameter, which may cause the entire data link system to stop.



Fig 8.2 When not setting station numbers consecutively





Fig 8.3 When setting station numbers in descending order

(3) The station numbers in the same loop have to be all different. If the same station number is assigned to another station in the same loop, the link module closer to the receiving port of the master station is applied, and the link data in the other station is ignored. Therefore, the setting as Fig 8.4 cannot be made.



Fig 8.4 When setting the same station number

- (4) When the number of slave stations set with link parameter differs from the actual number of slave stations
 - (a) When the set number of slave stations is greater than the actual number of slave stations, the slave stations do not exist in the system are treated as communication faulty stations.
 - (b) If the set number of slave stations is less than the actual number of slave stations, the data link is only performed to the slave stations set with the link parameters.

All other slave stations are processed as off-line mode stations.

MELSECNET mode		MELSECNET		MELSECNET/B			
Operating Mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	
Applicability				0	0	0	

8.2.2 Setting the link module station numbers in the MELSECNET/B data link system

Assign station number from the master station (set "00" to the master station) in ascending order to the forward loop direction. The settable maximum station number is "31".



Fig 8.5 Setting link module station numbers

REMARK

For station number setting of the link modules, refer to the manual for each link module.

POINT					
In the MELSE descending or	CNET/B dat der as the fi	a link system, gure below.	the slave station	on numbers can be set in	ł
Ma	ster station	Slave station	Slave station No. 2	Slave station No. 3	
	Station No. 00	Station No. 01	Station No. 03	Station No. 02	

Precautions for setting the link module station numbers

The following shows the precautions for station number setting.

 Station numbers have to be consecutive.
 If station numbers are not set consecutively, the stations with incorrect station numbers are treated as communication faulty stations.



(2) The station numbers in the same loop have to be all different. If the same station number is assigned to another station in the same loop, since the stations with the same station number send data simultaneously, communications are failed.



- (3) When the number of slave stations set with link parameter differs from the actual number of slave stations
 - (a) When the set number of slave stations is greater than the actual number of slave stations, the slave stations do not exist in the system are treated as communication faulty stations.
 - (b) If the set number of slave stations is less than the actual number of slave stations, the data link is only performed to the slave stations set with the link parameters.

All other slave stations are processed as off-line mode stations.

8.3 Setting Communication Speed

The overall distance of the MELSECNET/B data link system depends on the communication speed.

Set the communication speed with the switch setting of link modules. (For the setting, refer to the manual for the link module.)

The relationship between set communication speeds and overall distances is shown in Table 8.1.

Communication speed (M bps)	Overall cable distance (m) (ft)
0.125	1200 (3937.2)
0.250	600 (1968.6)
0.500	400 (1312.4)
1.00	200 (656.2)

Table 8.1 Communication speeds and overall distances

REMARK

The overall distance refers to the distance between link modules connected to both ends.



MELSECNET mode		MELSECNET		MELSECNET/B			
Operating Mode	Operating Mode MELSECNET mode		MELSECNET II composite mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	
Applicability	0	0	0				

8.4 Optical Fiber Cable/Coaxial Cable Wiring

This section describes a method for connecting optical fiber cables or coaxial cables with link modules.

8.4.1 Precautions for wiring

The following describes wiring precautions for optical fiber cables and coaxial cables for MELSECNET.

(1) Securing space for the cables

The minimum allowable bend radius for optical fiber cables and coaxial cables are defined.

To connect a coaxial cable with a data link module, ensure the space for bending equal to or larger than the minimum allowable bend radius in Table 8.2.

For the connector A and allowable bend radius r of optical fiber cables, contact Mitsubishi Electric System & Service Co., Ltd.

Table 8.2 Minimum allowable bend radius

Data link A module	Cable		Connector A (mm)	Allowable bend radius r (mm)
	Coavial cable	3C-2V	30	23
		5C-2V		30

(2) Doubling link cables

Connect the optical fiber cables or coaxial cables in duplex loop.

If the cables are not connected using both the forward and reverse loop or if the final station is not connected with the master station and therefore the cables are not connected in duplex loop, the data link is maintained only at normal operation but disabled in case of an error.

(a) Connecting cables only in forward loop or reverse loop disables data link to all stations.



(b) If cables are not connected between F-RD on the master station and F-SD on the final station, and between R-SD on the master station and R-RD on the final station, the data link is established in the loopback mode. Therefore, in case of an error, data link between the faulty station and the final station is disabled.



- (3) Optical fiber cable wiring
 - (a) When wiring optical fiber cables, do not touch the optical fiber cores of the plugs or jacks and protect them from dust and dirt.

Attached dirt, dust, or oil may cause increase in transmission loss, resulting in data link fault.

In addition, do not remove the connector cover before cable connection. When storing the optical fiber cable, attach the connector cover to protect the connector from dust and dirt.

(b) When connecting/disconnecting an optical fiber cable, be sure to shut off all phases of the external power supply used by the system.



- (4) Coaxial cable wiring
 - (a) Keep a distance of 100 mm (3.94inch) or more between coaxial cables and other power and control cables.

Grounding FG of the power supply module of the base unit, where the link module is mounted, is effective for preventing noise.

(b) When connecting/disconnecting a coaxial cable, be sure to shut off all phases of the external power supply used by the system.

MELSECNET mode		MELSECNET		MELSECNET/B			
Operating Mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	
Applicability	0	0	0				

8.4.2 Optical fiber cables connections

This section describes methods for connecting optical fiber cables with link modules.

(1) Connecting link modules with optical fiber cables Connect an optical fiber cable to a link module from the OUT connector to IN connector of the next station as shown in Fig 8.6 (Connect the cable from the OUT connector of the final station to the IN connector of the master station).



Fig 8.6 Connecting link modules with optical fiber cables

(2) Connecting optical fiber cables
 The following flowchart shows a method for connecting optical fiber cables.



(3) Disconnecting optical fiber cables
 The following flowchart shows a method for disconnecting optical fiber cables.





MELSECNET mode		MELSECNET		MELSECNET/B			
Operating Mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	
Applicability	0	0	0				

8.4.3 Coaxial cable connections

This section describes methods for connecting coaxial cables with link modules.

(1) Connecting link modules with coaxial cables

Connect the F-SD connector on a link module to the F-RD connector on the next module and connect the R-RD connector on a link module to the R-SD connector of the next module. (Connect the F-SD connector and R-RD connector of the final station with the F-RD connector and R-SD connector of the master station, respectively.)



Fig 8.9 Connecting link modules with coaxial cables

(2) Connecting coaxial cables

The following flowchart shows a method for connecting coaxial cables.



(3) Disconnecting coaxial cable

The following flowchart shows a method for disconnecting coaxial cables.



8.5 Shielded Twisted Pair Cable Wiring

This section describes a connection method for shielded twisted pair cables.

8.5.1 Precautions for wiring

The following shows the precautions when wiring shielded twisted pair cables in MELSECNET/B data link system.

- Laying shielded twisted pair cables
 When laying shielded twisted pair cables, follow the precautions below to prevent extraneous noise and surge induction.
 - (a) Do not install shielded twisted pair cables together with the main circuit, high-voltage cable, or load carrying wire, or bring them close to each other (Keep a distance of 100 mm (3.94inch) or more between them).
 - (b) Connect remote module terminal block so that enough distance can be secured between shielded twisted pair cables and the module power or I/O signal cables.
 - (c) Do not use any part of shielded twisted pair cables (e.g. One pair of the cables from three pairs of them) to supply power.
 - (d) When connecting/disconnecting a shielded twisted pair cable, be sure to shut off all phases of the external power supply used by the system.

(2) Connecting a terminal resistor

Connecting a terminal resistor came with the data link module (110 Ω , 1/2 W) between SDA/RDA and SDB/RDB at both end stations of a MELSECNET/B data link system is required.



8.5.2 Connecting shielded twisted pair cables

Wire shielded twisted pair cables to link modules as shown in Fig 8.12. Connect terminal resistors to both end stations.



Fig 8.12 Connecting shielded twisted pair cables to link modules

REMARK

Use M4-size terminal screws for terminal blocks that connect shielded twisted pair cables. Select solderless terminals suitable for the terminal screws. The tightening torque range is from 78 to 118 N • cm.

MELSECNET mode	MELSECNET			MELSECNET/B		
Operating Mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode
Applicability	0	0	0	0	0	0

8.6 Startup Procedure

In MELSECNET data link system, the power supplies have to be turned on simultaneously or in the order of system from lowest to highest.

(1) Two-tier system

Power on the system in the following order: All slave stations \rightarrow Master station

(2) Three-tier system

Power on the system in the following order: All sub-slave stations (slave stations in the third tier) \rightarrow All slave stations in the second tier \rightarrow Master station for the second tier

REMARK

- (1) If the automatic return function is set for the master station and all of the slave stations, other startup procedures are also applicable.
 For example, if the automatic return function is set to the master station for the second tier and the master station for the third tier, the power supplies can be turned on in the following order: Master station for the second tier → Local stations in the second tier (including the master station for the third tier) → Local stations in the third tier.
- (2) If all stations are powered on simultaneously, a faulty station may be detected because of the difference on startup timing between the master station and slave stations. If this occurs when the automatic return function is not set for the master station, communications may not be started normally. If it is set, the number of retries is stored in D9210 for retry processing.

MELSECNET mode	MELSECNET			MELSECNET/B		
Operating Mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode
Applicability	0	0	0	0	0	0

8.7 Self-diagnostics Test

(1) The self-diagnostics test checks link module hardware and link cables for wire break. The following five items are checked with the self-diagnostics test:

Test name	Description	MELSECNET	MELSECNET/B
Forward loop test	Checks optical fiber cables or coaxial cables in the entire data link system. This test also checks the forward loop, through which data link is usually performed.	0	×
Reverse loop test	Checks optical fiber cables or coaxial cables in the entire data link system. This test also checks the reverse loop, which is used for loopback in the event of an error.	0	×
Station-to-station test (Testing station)	Checks line connection between two stations, by setting the lower-numbered station as the testing station and another station	0	0
Station-to-station test (Tested station)	as the tested station.	0	0
Self-loopback test	Checks the hardware of each link module, including the transmission and receive circuits.	0	0

O: Executable, ×: Not executable

8.7.1 Self-loopback test

- (1) Self-loopback test
 - (a) The self-loopback test checks the hardware of each link module, including the transmission and receiving circuits in transmission system.
 - In the MELSECNET data link system, connect the send and receive ends on the host station with an optical fiber cable or coaxial cables as shown in Fig 8.13.
 - 2) In the MELSECNET/B data link system, perform this test to a single link module.

(Connecting SDA/RDA and SDB/RDB are not required.)

(b) When the receive end cannot receive data sent from the send end in a given time, the loop is determined to be faulty.



Fig 8.13 Self-loopback test

(2) Test procedure

The following flowchart shows the operation procedure for self-loopback test.



(3) Test result

The LEDs on the link module indicate the test result.

The descriptions of the LEDs are as follows.

- (a) Normal...... The following LEDs flash in order:
 - CRC, OVER, AB.IF, TIME, DATA, UNDER
- (b) Error......The corresponding LEDs turn ON and the test is discontinued.
 - 1) When the F.LOOP, R.LOOP and TIME LEDs turn ON:
 - a) The forward loop cable is broken.
 - b) The sending and receiving ends of the forward loop are not connected with a cable.
 - c) The sending end of the forward loop is connected to the sending end of the reverse loop, and the receiving end of the forward loop is connected to the receiving end of the reverse loop.
 - (2) When the F.LOOP, R.LOOP, and DATA LEDs turn ON:
 - a) The reverse loop cable is broken.
 - b) The sending and receiving ends of the reverse loop are not connected with a cable.
 - (3) When an ERROR LED other than 1) and 2) turns ON:
 - a)Hardware error
 - b) The cable was disconnected during the test.
 - c) A cable was broken during the test.

MELSECNET mode	MELSECNET			MELSECNET/B			
Operating Mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	
Applicability	0	0	0	0	0	0	

8.7.2 Station to station test

(1) Station to station test

The station to station test checks the cable connections of two adjacent stations. When the data sent from the master station link module is not returned from the slave station link module in a given time, the loop is determined to be faulty.



Fig 8.14 Station to station test

(2) Test procedure

The following flowchart shows the operation procedure for station to station test.



(3) Test result

The LEDs indicate the test result.

The descriptions of the LEDs are as follows.

(a) Normal... The following LEDs flash in order:

CRC, OVER, AB.IF, TIME, DATA, UNDER

- (b) Error......The corresponding LEDs turn ON and the test is discontinued.
 - 1) When the F.LOOP and TIME LEDs turn ON:
 - a) The forward loop cable is broken.
 - b) The sending and receiving ends of the forward loop are not connected with a cable.
 - (2) When the F.LOOP, R.LOOP, and TIME LEDs turn ON:
 - a) The reverse loop cable is broken.
 - b) The sending and receiving ends of the reverse loop are not connected with a cable.
 - c) The sending end of the forward loop is connected to the sending end of the reverse loop, and the receiving end of the forward loop is connected to the receiving end of the reverse loop.
 - (3) When an ERROR LED other than 1) and 2) turns ON:
 - a) Hardware error
 - b) The cable was disconnected during the test.
 - c) A cable was broken during the test.

8. PROCEDURES TO OPERATION

MELSECNET mode	MELSECNET			MELSECNET/B		
Operating Mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode
Applicability	0	0	0			

MELSEC-A

8.7.3 Forward loop test and reverse loop test

POINT The forward/reverse loop tests require data link parameter. Write data link parameter setting of only the total number of slave stations to the master station.

- (1) Forward loop test
 - (a) The forward loop test checks the forward loop in MELSECNET data link system after optical fiber cables or coaxial cables are laid down.
 - (b) When the receive end of the forward loop in the master station cannot receive data sent from the send end of the forward loop in the master station, the loop is determined to be faulty.



Fig 8.15 Forward loop test

- (2) Reverse loop test
 - (a) The reverse loop test checks the reverse loop in MELSECNET data link system after optical fiber cables or coaxial cables are laid down.
 - (b) When the receive end of the reverse loop in the master station cannot receive data sent from the send end of the reverse loop in the master station, the loop is determined to be faulty.



Fig 8.16 Reverse loop test

(3) Test procedure

The following flowchart shows the operation procedure for forward loop test/reverse loop test.



(4) Test result

The LEDs on the link module of the master station or the GPP link monitor function show(s) the test result.

- (a) For GPP link monitor, refer to Section 10.1.
- (b) The descriptions of the LEDs are as follows.
 - 1) Normal.....The following LEDs flash in order:
 - CRC, OVER, AB.IF, TIME, DATA, UNDER
 - 2) Error...... The corresponding LEDs flash and the test is discontinued.
 - a) When the TIME, DATA and UNDER LEDs flash:
 - Hardware failure
 - Disconnection of the optical fiber/coaxial cables, or loopback due to a slave station error
 - The master station (00) setting is made for more than one station.
 - The monitoring time set is too short.

POINT

In case of forward/reverse loop error, the data link will be established in the reverse/forward loop or the loopback mode.

The forward/reverse loop data link is recovered when the loop returns to normal. The LEDs remain in the error status even after the recovery, reset the master station and execute the forward loop test/reverse loop test again.
MELSECNET mode		MELSECNET		MELSECNET/B			
Operating Mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	
Applicability	0	0	0	0	0	0	

MELSEC-A

9 PROGRAMMING

This chapter describes methods for programming the master and local stations to perform data link among CPU modules.

9.1 Precautions for Creating Programs

(1) Used link devices

The device numbers assigned to each station with link parameter can be used for link devices (B, W, X, Y) to be used in data link programs.

(2) Writing fail safe programs

Using special link relays (M9200 to M9255) and special link registers (D9200 to D9255), an interlock should be provided for data link programs among communication stations so that the other stations data can be used only when the data link is performed normally.

(3) Data link method for data of two or more words

When writing data of two or more words to the link registers, employ the following procedure to prevent sending/receiving the data in which the old and new data are mixed.

(a) When using a module that executes link refresh immediately after link scan (such as the AnNCPUP21/R21 and the A0J2CPUP23/R23), executing handshake processing with link relays are recommended so that the link register





In a program example above, if link refresh is performed when the D0 data is being transmitted to W0, the new data and old data is stored to W0 and W1 respectively, and sent to the receive station.

(b) Set link refresh disable area as shown below when using a module which can make rink refresh execution disable setting such as the AnNCPUP21/R21 by using ON/OFF of M9053 and the DI/EI instructions.



(1) The following shows a program for sending W0 and W1 data to another station



- (c) The module that performs link refresh only after the execution of the END instruction in the sequence program (A3HCPU, A3MCPU, AnACPU, AnUCPU, A2USCPU(S1), QnACPU, QnAS(H)CPU(S1), QCPU-A), even if data of two words or more are sent to link registers, the old and new data are not mixed.
- (4) Precautions when using the MELSECNET II mode or the MELSECNET II composite mode

The timing to be link refreshed may differ between the device range assigned with the first half of the link parameters and that assigned with the last half of the link parameters.

Handshake processing is required to receive the data written at the same timing.



- (5) Read/write of the special function module installed to the remote I/O station
 - (a) Execute the RFRP or RTOP instruction after initial communications of remote I/O stations.

The completion of initial communications can be checked with special link registers (D9224 to D9227). (Refer to the ladder examples in Section 9.8.) If the RFRP or RTOP instruction is executed before the completion of initial communications, an "OPERATION ERROR" occurs.

- (b) If a receive error occurs while the RFRP or RTOP instruction is being executed, the handshake signals (YnE, YnF) may remain ON. Therefore, create a ladder so that the handshake signals (YnE, YnF) and the special function module error signal (X(n+1)D) are turned OFF in case of a communication error. (Refer to the ladder examples in Section 9.8.)
- (6) Link data in cut-off stations If a local station and/or remote I/O station is cut-off during data link due to power-on or reset operation, the data immediately before the cut-off are retained in other stations.
- (7) Instructions that cannot be used in data link programs
 Pulse instructions (PLS, SFTP, etc.) cannot be used for outputting data to a remote
 I/O station or communications between the master station and local stations.
 Create a program with referring to the transmission delay time described in Section
 6.2.
- (8) Precautions for transient transmission execution using link instructions
 - (a) LRDP, LWTP.... These instructions can only be executed at one point in the same system. They cannot be executed simultaneously at two or more points.
 - (b) RFRP, RTOP.... These instructions cannot be executed at two or more points in a special function module.
- (9) If initial settings have been made in a program for the buffer memory in the special function module on the remote I/O station, create the program so that when only that remote I/O station is reset (by powering it off or by turning on the reset switch of the network module on the remote I/O station), the master station will detect the status and initial settings will be made to the special function module again. Initial settings are made to the special function module when:
 - (a) Setting sampling period specification and set data setting request with the A/D converter module, A616AD.
 - (b) Setting the number of channels and averaging processing specification with the A/D converter module, A68AD.



*: The operating status of the remote I/O station can be checked using remote I/O station error (D9228 to D9231) of special link registers. To use the remote I/O station error in a program, develop it in bit devices M, L, etc. using the MOV instruction. (Example: [MOV D9228 K4M1000], and the contents of D9228 are developed in M1000 to M1015.)

MELSECNET mode		MELSECNET		MELSECNET/B			
Operating Mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	
Applicability	0	0	0	0	0	0	

9.2 Special Link Relays

Special link relays are internal relays controlled by turning ON/OFF due to various causes during data link. Monitoring or using them in a sequence program can check data link errors.

9.2.1 Special link relays enabled only for the master station

Table 9.1 and Table 9.2 show the special relays controlled only when the host station is set to the master station.

Device Number	Name	Data	Description
M9200	LRDP instruction received	OFF : Unreceived ON : Received	 Turned ON when an LRDP (word device read) instruction is received. Used in a user program as an interlock for an LRDP instruction. Remains ON after the completion of word device read processing called by an LRDP instruction. Turned OFF with an RST instruction in a user program.
M9201	LRDP instruction completed	OFF : Uncompleted ON : Completed	 Turned ON after an LRDP (word device read) instruction has been executed. The execution results are stored in D9200. Used as a conditional contact to reset M9200 and M9201 after the completion of word device read processing called by an LRDP instruction. Turned OFF with an RST instruction in a user program after it has been turned ON.
M9202	LWTP instruction received	OFF : Unreceived ON : Received	 Turned ON when an LWTP (word device write) instruction is received. Used in a user program as an interlock for an LWTP instruction. Remains ON after the completion of word device write processing called by an LWTP instruction. Turned OFF with an RST instruction in a user program.
M9203	LWTP instruction completed	OFF : Uncompleted ON : Completed	 Turned ON after an LWTP (word device write) instruction has been executed. The execution results are stored in D9201. Used as a conditional contact to reset M9202 and M9203 after the completion of word device write processing called by an LWTP instruction. Turned OFF with an RST instruction in a user program after it has been turned ON.
M9206	Link parameter error in the host station	OFF : Normal ON : Error	 Turned ON when no link parameter of the host station is set or any of the settings is incorrect. To turn it OFF, use the RST instruction in the user program.
M9207	Link parameter inconsistency with the master station	OFF : Normal ON : Error	 Turned ON if a lower tier link uses device ranges (B, W) outside the range that is set to be used by the master station in the upper tier link. Check is executed only when M9209 is OFF.
M9208	B and W transmission range for the master station (lower tier master stations only)	OFF : Send to the second and third tiers ON : Send to the second tier only	 Sets whether the B and W data controlled by the master station in the upper tier is sent to the local stations (sub-slave stations) in the lower tier. OFF : B and W data in the master station is sent to the sub-slave stations. ON : B and W data in the master station is not sent to the sub-slave stations.

Table 9.1 MELSECNET special link relays list

Device Number	Name	Data	Description
M9209	Link parameter check instruction (lower tier link master stations only)	OFF : Check executed ON : Check not executed	 Turned ON when the link devices (B and W) used by the upper tier and lower tier are not compared for "match". When M9209 is OFF, the link parameters for the upper tier and lower tier are checked.
M9210	Link card error (master station)	OFF : Normal ON : Error	Turned ON when the link card hardware is faulty.
M9224	Link status	OFF : Offline ON : Online, station-to- station test, or self- loopback test	 Turned ON when the master station is offline, in the station-to- station test mode, or in the self-loopback test mode. Turned OFF when the master station is reset after being placed in the online mode.
M9225	Forward loop error	OFF : Normal ON : Error	 Turned ON when any of the following errors occurs on the forward loop line between the master station and the final station: Cable disconnection Forward loop receiver error of the master station link module Forward loop transmitter error of the link module on the final local station Turned ON when the station-to-station test, including the final station, is executed during the data link. Turned OFF automatically when the error state is eliminated.
M9226	Reverse loop error	OFF : Normal ON : Error	 Turned ON when any of the following errors occurs on the reverse loop line between the master station and station No.1 : Cable disconnection Reverse loop receiver error of the master station link module Reverse loop transmitter error of the link module on station No.1 Turned ON when the station-to-station test was executed at station No.1, during data link execution. Turned OFF automatically when the error state is eliminated.
M9227	Loop test status	OFF : Unexecuted ON : Forward loop test or reverse loop test being executed	 Turned ON when a forward loop test or reverse loop test is being executed for the master station.
M9232	Local station operating status	OFF : RUN or STEP RUN status ON : STOP or PAUSE status	 ON/OFF status depends on the operation status of the local station. Turned ON when the status of any local station in the loop changes to STOP or PAUSE. Turned OFF automatically when the status of all local stations changes to RUN or STEP RUN. (That is, M9232 is turned OFF when bits D9212 to D9215 are all OFF.)

Table 9.1	MELSECNET	special	link relavs	list	(Continued)
					(

Device Number	Name	Data	Description
M9233	Local station error detected	OFF : No error ON : Error detected	 Turned ON when a local station in the executed loop detects an error in another station (M9255 ON). Turned OFF automatically when the faulty station is returned to the normal state or the data link returns to the normal status by switching the loop line. (That is, M9233 is turned OFF when bits D9216 to D9219 are all OFF.)
M9235	Local station or remote I/O station parameter error detected'	OFF : No error ON : Error detected	 For local station: Parameter inconsistency was detected since devices other than the link relay (B) and link register (W)ranges, (which are assigned to the lower link master station) are assigned to link parameters for the lower link with link parameters for the upper link. For remote I/O station : Error in I/O assignment, or neither inputs (X) nor outputs (Y) are set with the link parameters. Turned OFF when the error is eliminated by correcting the link parameters. (That is, M9235 is turned OFF when bits D9220 to D9223 are all OFF.)
M9236	Local station or remote I/O station initial communication status	OFF : Not communi- cating ON : Communi- cating	 Turned ON while a local station and/or remote I/O station is communicating the initial setting data (link parameter) to the master station to execute data link processing. Automatically turned OFF when the communication for initial data setting has been completed. (That is, M9236 is turned OFF when bits D9224 to D9227 are all OFF.)
M9237	Local station or remote I/O station error	OFF : Normal ON : Error	 Turned ON when an error occurs with one local station or remote I/O station within the loop. (The relay is turned ON while a station-to-station test is being executed for a local station or a remote I/O station and the data link is operating.) Automatically turned OFF when the faulty station returns to the normal status or the data link returns to the normal status by switching the loop line. (That is, M9237 is turned OFF when bits D9228 to D9231 are all OFF.)
M9238	Local station or remote I/O station forward/reverse loop error	OFF : Normal ON : Error	• Turned ON when an error occurs in the forward loop line or reverse loop line of the local stations and remote I/O stations. (That is, M9238 is turned OFF when bits D9232 to D9239 are all OFF.)

Table 9.2 MELSECNET/B	special link relays list

Device Number	Name	Data	Description
M9200	LRDP instruction received	OFF : Unreceived ON : Received	 Turned ON when an LRDP (word device read) instruction is received. Used in a user program as an interlock for an LRDP instruction. Remains ON after the completion of word device read processing called by an LRDP instruction. Turned OFF with an RST instruction in a user program.
M9201	LRDP instruction completed	OFF : Uncompleted ON : Completed	 Turned ON after an LRDP (word device read) instruction has been executed. The execution results are stored in D9200. Used as a conditional contact to reset M9200 and M9201 after the completion of word device read processing called by an LRDP instruction. Turned OFF with an RST instruction in a user program after it has been turned ON.
M9202	LWTP instruction received	OFF : Unreceived ON : Received	 Turned ON when an LWTP (word device write) instruction is received. Used in a user program as an interlock for an LWTP instruction. Remains ON after the completion of word device write processing called by an LWTP instruction. Turned OFF with an RST instruction in a user program.
M9203	LWTP instruction completed	OFF : Uncompleted ON : Completed	 Turned ON after an LWTP (word device write) instruction has been executed. The execution results are stored in D9201. Used as a conditional contact to reset M9202 and M9203 after the completion of word device write processing called by an LWTP instruction. Turned OFF with an RST instruction in a user program.
M9206	Link parameter error in the host station	OFF : Normal ON : Error	 Turned ON when no link parameter of the host station is set, or any of the settings is incorrect. Turned OFF with an RST instruction in a user program.
M9207	Link parameter inconsistency with the master station	OFF : Normal ON : Error	 Turned ON if a lower tier link uses device ranges (B, W) outside the range that is set to be used by the master station in the upper tier link. Check is executed only when M9209 is OFF.
M9208	B and W transmission range for the master station (lower tier master stations only)	OFF : Send to the second and third tiers ON : Send to the second tier only	 Sets whether the B and W data controlled by the master station in the upper tier is sent to the local stations (sub-slave stations) in the lower tier. OFF : B and W data in the master station is sent to the sub-slave stations. ON : B and W data in the master station is not sent to the sub-slave stations.
M9209	Link parameter check instruction (lower tier link master stations only)	OFF : Check executed ON : Check not executed	 Turned ON when the link devices (B and W) used by the upper tier and lower tier are not compared for "match". When M9209 is OFF, the link parameters for the upper tier and lower tier are checked.
M9210	Link card error (master station)	OFF : Normal ON : Error	Turned ON when the link card hardware is faulty.
M9224	Link status	OFF : Offline ON : Online, station-to- station test, or self- loopback test	 Turned ON when the master station is offline, in the station-to- station test mode, or in the self-loopback test mode. Turned OFF when the master station is reset after being placed in the online mode.

П

Table 9.2 MELSECNET/B	special I	link relavs	list ((Continued)
				(

Device Number	Name	Data	Description
M9232	Local station operating status	OFF : RUN or STEP RUN status ON : STOP or PAUSE status	 ON/OFF status depends on the operation status of the local station. Turned ON when the status of any local station in the loop changes to STOP or PAUSE. Turned OFF automatically when the status of all local stations changes to RUN or STEP RUN. (That is, M9232 is turned OFF when bits D9212 to D9215 are all OFF.)
M9233	Local station error detected	OFF : No error ON : Error detected	 Turned ON when a local station in the executed loop detects an error in another station (M9255 ON). Automatically turned OFF when the faulty station is returned to the normal state. (That is, M9233 is turned OFF when bits D9216 to D9219 are all OFF.)
M9235	Local station parameter error detected	OFF : No error ON : Error detected	 Turned ON when the bit device range (link relays and link registers) outside the range allocated to a master station for the lower tier is allocated with the link parameters for the lower tier. Turned OFF when the error is eliminated by correcting the link parameters. (That is, M9235 is turned OFF when bits D9220 and D9221 are all OFF.)
M9236	Local station initial communications status	OFF : Not communi- cating ON : Communi- cating	 Turned ON while a local station is communicating the initial setting data (link parameters) to a master station to execute data link processing. Automatically turned OFF when the communication for initial data setting has been completed. (That is, M9236 is turned OFF when bits D9224 and D9225 are all OFF.)
M9237	Local station error	OFF : Normal ON : Error	 Turned ON when an error occurs at one local station within the loop. (The relay is turned ON while a station-to-station test is being executed for a local station and the data link is operating.) Automatically turned OFF when the faulty station returns to normal. (That is, M9237 is turned OFF when bits D9228 and D9229 are all OFF.)

MELSECNET mode		MELSECNET		MELSECNET/B			
Operating Mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	
Applicability	0	0	0	0	0	0	

MELSEC-A

9.2.2 Special link relays enabled only for local stations

Table 9.3 and Table 9.4 show the special relays controlled only when the host station is set to local station.

Table 9.3 MELSECNET special link relays list

Device Number	Name	Data	Description
M9204	LRDP instruction completed	OFF : Uncompleted ON : Completed	 Turned ON by a local station upon the completion of word device read processing called by an LRDP instruction.
M9205	LWTP instruction completed	OFF : Uncompleted ON : Completed	• Turned ON by a local station upon the completion of word device write processing called by an LWTP instruction.
M9211	Link card error (local station)	OFF : Normal ON : Error	Turned ON when the link card hardware is faulty.
M9240	Link status	OFF : Online ON : Offline, station-to- station test, or self- loopback test	 Turned ON when the host station is in offline, station-to-station test, or self-loopback test mode. Turned OFF when the host station is reset after being placed in the online mode.
M9241	Forward loop line error	OFF : Normal ON : Error	 Turned ON when any of the following errors occurs in the forward loop line between the host station and the preceding station: Cable disconnection Forward loop receiver error of the host station link module Forward loop transmitter error of the link module on the preceding station Turned OFF automatically when the error state is eliminated.
M9242	Reverse loop line error	OFF : Normal ON : Error	 Turned ON when any of the following occurs in the reverse loop line between the host station and the next station: Cable disconnection Reverse loop receiving part error in the data link module of the station itself Reverse loop sending part error in the data link module of the next station Turned OFF automatically when the error state is eliminated.
M9243	Loopback execution	OFF : Not-executed ON : Executed	Turned ON when loopback is executed by the host station.
M9246	Data unreceived	OFF : Received ON : Unreceived	Turned ON when the data from the master station has not been received.
M9247	Data unreceived	OFF : Received ON : Unreceived	 In the three-tier system, M9247 is turned ON when a sub-slave station has not received data from the master station for the third tier. (That is, M9247 is ON while M9208 is ON.)
M9250	Parameter unreceived	OFF : Received ON : Unreceived	 Turned ON when the link parameters have not been received from the master station. Automatically turned OFF when the link parameter is received. The master station sends the link parameters to each local station every time the loop line is switched. Only effective while the loop line in which the data link is executed is online.

П

Device Number	Name	Data	Description
M9251	Link halt	OFF : Normal ON : Halt	 ON/OFF status depends on whether the station itself stopped the data link. Turned ON when the data link is established in neither the forward loop line nor the reverse loop line. Automatically turned OFF when the data link returns to the normal state. Only effective while the loop line in which the data link is executed is online.
M9252	Loop test status	OFF : Not executed ON : Forward loop test or reverse loop test is being executed.	 Turned ON while the host station itself is in the forward loop test mode or the reverse loop test mode.
M9253	Master station operating status	OFF : RUN or STEP RUN status ON : STOP or PAUSE status	 Controlled according to the operation status of the master station. Turned ON when the status of a master station is either STOP or PAUSE. Turned OFF when the status of the master station changes to RUN or STEP RUN.
M9254	Operating status of local stations except host	OFF : RUN or STEP RUN status ON : STOP or PAUSE status	 Controlled according to the operation status of a local station other than the host station. Turned ON when the status of a local station other than the host station in the loop is either STOP or PAUSE. Not turned ON when the status of the host station itself is either STOP or PAUSE. Automatically turned OFF when the status of a local station other than the host station in the loop changes to RUN or STEP RUN. (That is, M9254 is turned OFF when bits D9248 to D9251 are all OFF.)
M9255	Error status of local stations except host	OFF : Normal ON : Error	 Controlled by detecting an error of a local station other than the host station. Turned ON if an error occurs in one local station other than the host station in the loop. Automatically turned OFF when the faulty station returns to the normal state or the data link returns to the normal state by switching the loop line. (That is, M9255 is turned OFF when bits D9252 to D9255 are all OFF.)

Table 9.4 MELSECNET/B	special link r	elavs list
	opoolai mini	onayo not

Device Number	Name	Data	Description
M9204	LRDP instruction completed	OFF : Uncompleted ON : Completed	 Turned ON by a local station upon the completion of word device read processing called by an LRDP instruction.
M9205	LWTP instruction completed	OFF : Uncompleted ON : Completed	 Turned ON by a local station upon the completion of word device write processing called by an LWTP instruction.
M9211	Link card error (local station)	OFF : Normal ON : Error	Turned ON when the link card hardware is faulty.
M9240	Link status	OFF : Online ON : Offline, station-to- station test, or self- loopback test	 Turned ON when the host station is in offline, station-to-station test, or self-loopback test mode. Turned OFF when the host station is reset after being placed in the online mode.
M9246	Data unreceived	OFF : Received ON : Unreceived	Turned ON when the data from the master station has not been received.
M9247	Data unreceived	OFF : Received ON : Unreceived	• Turned ON when a sub-slave station has not received data from the master station in the three-tire system. (That is, M9247 is ON while M9208 is ON.)
M9250	Parameter unreceived	OFF : Received ON : Unreceived	 Turned ON when the link parameters have not been received from the master station. Automatically turned OFF when the link parameter is received. Only valid while the loop line in which the data link is executed is online.
M9251	Link halt	OFF : Normal ON : Halt	 Controlled by whether the host station stopped the data link or not. Automatically turned OFF when the data link returns to the normal status. Only effective while the loop line in which the data link is executed is online.
M9253	Master station operating status	OFF : RUN or STEP RUN status ON : STOP or PAUSE status	 Controlled according to the operation status of a master station. Turned ON when the status of the master station is either STOP or PAUSE. Turned OFF when the status of the master station changes to RUN or STEP RUN.
M9254	Operating status of local stations except host	OFF : RUN or STEP RUN status ON : STOP or PAUSE status	 Controlled according to the operation status of a local station other than the host station. Turned ON when the status of a local station other than the host station in the loop is either STOP or PAUSE. Not turned ON even if the status of the host station is either STOP or PAUSE. Automatically turned OFF when the status of a local station other than the host station in the loop changes to RUN or STEP RUN. (That is, M9254 is turned OFF when bits D9248 and D9249 are OFF.)
M9255	Error status of local stations except host	OFF : Normal ON : Error	 Controlled by detecting by an error of a local station other than the host station. Turned ON if an error occurs in one local station other than the host station in the loop. Automatically turned OFF when the faulty station returns to the normal state or the data link returns to the normal state by switching the loop line. (That is, M9255 is turned OFF when bits D9252 to D9253 are all OFF.)

MELSECNET mode		MELSECNET			MELSECNET/B	3
Operating Mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode
Applicability	0	0	0	0	0	0

9.3 Special Link Registers

Special link registers store causes of errors that occur at the time of data link in the form of value. Monitoring special link registers enables to detect the area with error or the cause of error.

9.3.1 Special link registers enabled only for the master station

Table 9.5 and Table 9.6 show the special registers controlled only when the host station is set to the master station.

Device Number	Name	Data	Description
D9200	LRDP execution result	 0 : Normal 2 : LRDP instruction setting fault 3 : Corresponding station error 4 : LRDP cannot be executed in the corresponding station 	Stores the execution result of an LRDP (word device read) instruction (M9201 ON). • LRDP instruction setting fault:
D9201	LWTP execution result	 0 : Normal 2 : LWTP instruction setting fault 3 : Corresponding station error 4 : LWTP cannot be executed in the corresponding station 	Stores the execution result of an LWTP (word device write) instruction (M9203 ON). • LWTP instruction setting fault: Faulty setting of the LWTP instruction constant, source, and/or target • Corresponding station error: • LWTP instruction cannot be executed by the corresponding station: A remote I/O station is connected to the station designated with the LWTP instruction, or a local station of the QCPU specified by LWTP instruction is in STOP status.

Table 9.5 List of MELSECNET special link registers

Device Number	Name	Data	Description
D9202	Local station link type	Stores the status of No.1 to No.16	Stores whether a slave station is compatible with the MELSECNET mode or MELSECNET II mode. • MELSECNET II-compatible station : "1"
D9203		Stores the status of No.17 to No.32	MELSECNET-compatible station : "0" Bit DEVICE DEV
D9241		Stores the status of No.33 to No.48	 I Las Las Las Las Las Las Las Las Las Las
D9242		Stores the status of No.49 to No.64	above special registers is enabled.Even If the host station (master station) goes down, the data before the failure will be also held

Table 9.5 List of MELSECNE	special link registers	(Continued)
		(

MELSEC-A

Device Number	Name	Data	Description
D9204	Link status	0 : Data link in forward loop 1 : Data link in reverse loop 2 : Loopback in forward/ reverse direction 3 : Loopback in reverse direction 5 : Data link impossible	Stores the current path of the data link. (1) Forward loop

Table 9.5 List of MELSECNET special link registers (Continued)

Device Number	Name	Data	Description
D9205	Loopback execution station	Station executing forward loopback	Stores the number of local station or remote I/O station at which loopback is being executed.
D9206	Loopback execution station	Station executing reverse loopback	 Forward loopback Forward loopback Example: "1" is stored in D9205 and "3" is stored in D9206. The values stored in D9205 and D9206 are not reset to "0" when the data link returns to the normal state (data link in forward loop). Reset the programmable controller CPU to return the set values to "0".
D9207		Maximum value	 Stores the time used for data link processing (link scan time) by all of the local stations and remote I/O stations in the loop currently being used for data link. (in 10ms unit) Link scan time definition:
D9208	Link scan time	Minimum value	When M > LS When M < LS When M < LS When M < LS When M < LS
D9209		Current value	M : Sequence program scan time by master station LS : Link scan time (data link processing)
D9210	Retry count	Total number stored	 Stores the total number of retries conducted when a transmission error occurs. Definition of retry processing: If data is lost or becomes unreliable due to the occurrence of a data transmission processing error, the same data is sent again. Counting stops if the number of retries exceeds the maximum limit "FFFF_H". Execute reset operation to clear the data to "0".
D9211	Loop switching count	Total number stored	 Stores the total number of times that the forward loop is switched to a reverse loop or to loopback. Counting stops if the number of switches exceeds the maximum limit "FFFF_H". Execute reset operation to clear the data to "0".

Table 9.5 List of MELSECNET special link registers (Continued)

MELSEC-A

Device Number	Name	Data	Description
D9212		Stores the status of No.1 to No.16	Stores the station No. of STOP- or PAUSE-status local station(s) as shown below. DEVICE NUMBER b15 b14 b13 b12 b11 b10 b9 b8 b7 b6 b5 b4 b3 b2 b1 b0 Device 1 b14 b13 b12 b11 b10 b9 b8 b7 b6 b5 b4 b3 b2 b1 b0 Device 1 b14 b13 b12 b11 b10 b9 b8 b7 b6 b5 b4 b3 b2 b1 b0 Device 1 b14 b13 b12 b11 b10 b19 b8 b7 b6 b5 b4 b3 b2 b1 b0 Device 1 b14 b13 b12 b11 b10 b19 b8 b7 b6 b5 b4 b3 b2 b1 b0 Device 1 b14 b13 b12 b11 b10 b19 b8 b7 b6 b5 b4 b3 b2 b1 b0 Device 1 b14 b13 b12 b11 b10 b19 b8 b7 b6 b5 b4 b3 b2 b1 b0 Device 1 b14 b13 b12 b11 b10 b19 b18 b17 b10 b19 b18 b17 b10 b19 b11 b10 b19 b18 b17 b10 b19 b18 b18 b10 b19 b18
D9213	Local station	Stores the status of No.17 to No.32	Ds212 Lis Lis <thlis< th=""> <thlis< td="" td<=""></thlis<></thlis<>
D9214	operating status	Stores the status of No.33 to No.48	 relevant bit is "0", the corresponding bit in the above special registers is enabled. Even If the host station (master station) goes down, the data before the failure will be also held. When the status of a local station changes to STOP or PAUSE, the
D9215		Stores the status of No.49 to No.64	corresponding bit is "1". The bit status of remote I/O station always remains "0", indicating RUN. Example:When the operation status of No. 7 changes to the STOP, "1" is set to bit 6 of D9212. When D9212 is monitored, its value is "64 (40H)".
D9216		Stores the status of No.1 to No.16	Stores the numbers of the station that detect the occurrence of an error at another station. DEVICE DEVICE
D9217		Stores the status of No.17 to No.32	D9216 L10 L13 L12 L11 L10 L9 L5 L4 L3 L2 L1 D9217 L32 L31 L30 L29 L28 L27 L26 L25 L24 L23 L22 L21 L20 L19 L18 L17 D9218 L48 L47 L46 L45 L44 L43 L42 L41 L40 L39 L38 L37 L36 L35 L34 L33 D9219 L64 L63 L62 L61 L60 L59 L58 L57 L56 L55 L54 L53 L52 L51 L50 L49 • If a local station goes down, data before the failure will be held. When contents of D9224 to D9227 and D9228 to D9231 are ORed and the ORed and the D9218 L40
D9218	Local station error detection status	Stores the status of No.33 to No.48	 relevant bit is "0", the corresponding bit in the above special registers is enabled. Even If the host station (master station) goes down, the data before the failure will be also held. When a normally operating local station detects an error at another local
D9219		Stores the status of No.49 to No.64	 station, the bit corresponding to the normally operating station is set to "1". The bit status of remote I/O station always remains "0". Example:When No. 5 detects that No. 4 is faulty, "1" is set to bit 4 of D9216. When D9216 is monitored, its value is "16 (10H)". When the faulty station recovers normal operating status or when the loop line is switched so that the data link returns to normal operating status, the corresponding bit is automatically reset to "0".

Table 9.5 List of MELSECNET special link registers (Continued)

Device Number	Name	Data	Description					
D9220	Local station parameter mismatched or remote I/O station input/output allocation error	Stores the status of No.1 to No.16 Stores the status of NUMBER b15 b14 b13 b12 b11 b10 b9 b8 b7 b6 b5 b4 t						
D9221		Stores the status of No.17 to No.32	D9220 UR UR <thu< td=""></thu<>					
D9222		Stores the status of No.33 to No.48	D9223 L/R L/R </td					
D9223		Stores the status of No.49 to No.64	 enabled. Even If the host station (master station) goes down, the data before the failure will be also held. Example:When local station No. 5 is set as a remote I/O station, "1" is set for bit 4 of D9220. When D9220 is monitored, its value is "16 (10_H)". When the link parameter settings is corrected and the status of the master station is switched from STOP to RUN, the bit is automatically reset to "0". 					

Device Number	Name	Data	Description			
D9224	Initial communication between local and/ or remote I/O stations	Stores the status of No.1 to No.16	Stores station Nos. of local stations or remote I/O stations, which are exchanging initial setting data for data link processing, into the corresponding bits in D9224 to D9227 as shown below. DEVICE Bit NUMBER product by a back by the local back			
D9225		Stores the status of No.17 to No.32 Initial	D9224 L/R L/R <thl r<="" t<="" td=""></thl>			
D9226		Stores the status of No.33 to No.48	 UP UP U			
D9227		Stores the status of No.49 to No.64	 Example: When No. 23 and No. 45 are communicating initial setting data (link parameters), "1" is set to bit 6 of D9225 and bit 12 of D9226. When D9225 is monitored, its value is "64 (40н)"and when D9226 is monitored, its value is "4096 (1000н)". When the communication of initial setting data is complete, the bit is automatically reset to "0". 			
D9228	Local station or remote I/O station error	Stores the status of No.1 to No.16	 The master station detects the number of a faulty local or remote I/O station on data linking loop, and the information is stored in the corresponding bit of the data register as shown below. A station is determined to be faulty if the data returned from it to the master station is not received within the specified length of time. 			
D9229		Stores the status of No.17 to No.32	DEVICE NUMBER UR			
D9230		Stores the status of No.33 to No.48	 When data is not received within the specified length of time, the bit corresponding to the station number of the local station is set. Example: When an error at No. 3 causes it to fail to return the data to the master station, "1" is set for bit 2 of D9228. When D9228 is monitored, its value is "4". When the loop line becomes faulty, "1" is set for the bits if the 			
D9231		Stores the status of No.49 to No.64	 stations downstream the faulty point or for all local stations. When the master station becomes faulty or the setting for the monitoring time is too small, "1" is set for the bits corresponding to all local stations. When the faulty station recovers normal operating status or when the loop line is switched so that the data link returns to normal operating status, the bit is automatically reset to "0". 			

Device Number	Name	Data	Description			
D9232	Local station or remote I/O station loop error	Stores the status of No.1 to No.8	Stores the station number of the local and remote I/O stations a which an error is detected in the forward loop line or reverse loc line. DEVICE DEVICE DEVICE DIS_DIA_DI3_DI3_DI3_DI3_DI3_DI3_DI3_DI3_DI3_DI3			
D9233		Stores the status of No.9 to No.16	D9232 R F R R R R			
D9234		Stores the status of No.17 to No.24	D9237 R F			
D9235		Stores the status of No.25 to No.32	 F: Forward loop line R: Reverse loop line If a local station goes down, data before the failure will be held. When contents of D9224 to D9227 and D9228 to D9231 are ORed and the relevant bit is "0", the corresponding bit in the above special registers is enabled. Even If the host station (master station) goes down, the data before the failure will be also held. When an error is detected at a local station and/or remote I/O station in the forward loop line or the reverse loop line, the corresponding bit is set to "1". Example: When an error is detected in the forward loop line at No. 			
D9236		Stores the status of No.33 to No.40				
D9237		Stores the status of No.41 to No.48	 5, "1" is set for bit 8 of D9232. When D9232 is monitored, its value is "256 (100_H)". This error will have been caused by one of the following: (a) A faulty connection of the forward loop cable connecting No. 4 and No. 5 (b) A fault of the forward loop receiver of link module on 			
D9238		Stores the status of No.49 to No.56	 No.5 (c) A fault of the forward loop transmitter of link module on No.4 With errors other than loop line errors, such as hardware errors and data communication errors, only the error involved with the loop line currently being used will be detected. The error status is retained. When data link is executed again with the loop line in which an error was detected, the bit data is automatically reset to "0" provided that the fault has been removed. 			
D9239		Stores the status of No.57 to No.64				
D9240	Receive error detection count	Stores the total number of receive error occurrences	 Stores the number of times that the following errors are detected in the loop line currently being used: "CRC", "AB.IF", "OVER" Counting stops if the number of receive error occurrences exceeds the maximum limit "FFFFH". Execute the reset operation to clear the data to "0". 			

Device Number	Name	Data	Description		
D9200	LRDP execution result	 0 : Normal 2 : LRDP instruction setting fault 3 : Corresponding station error 4 : LRDP can be executed in the corresponding station 	Stores the execution result of an LRDP (word device read) instruction (M9201 ON). • LRDP instruction setting fault:		
D9201	LWTP execution result	 0 : Normal 2 : LWTP instruction setting fault 3 : Corresponding station error 4 : LWTP cannot be executed in the corresponding station 	Stores the execution result of an LWTP (word device write) instruction (M9203 ON). • LWTP instruction setting fault:		
D9202	Local station link	Stores the status of No.1 to No.16	Stores whether a slave station is compatible with the MELSECNET mode or MELSECNET II mode. • MELSECNET II-compatible station : "1" • MELSECNET-compatible station : "0"		
D9203	type	Stores the status of No.17 to No.31	DEVICE Bit Bit<		
D9204	Link status	0 : Data link 5 : Data link impossible	Stores the current path of the data link. (1) Forward loop		

Device Number	Name	Data	Description			
D9207		Maximum value	 Stores the time used for data link processing (link scan time) by all of the local stations and remote I/O stations in the loop currently being used for data link. (in 10ms unit) Link scan time definition: 			
D9208	Link scan time	Minimum value	When M > LS M END 0 END 0 When M < LS			
D9209		Current value	M : Sequence program scan time by master station LS : Link scan time (data link processing)			
D9210	Retry count	Total number stored	 Stores the total number of retries conducted when a transmission error occurs. Definition of retry processing: If data is lost or becomes unreliable due to the occurrence of a data transmission processing error, the same data is sent again. Counting stops if the number of retries exceeds the maximum limit "FFFFH". Execute reset operation to clear the data to "0". 			

Table 9.6 List of MELSECNET/B special link registers (Continued)

MELSEC-A

Table 9.6 List of MELSECNET/E	special link registers	(Continued)
-------------------------------	------------------------	-------------

Device Number	Name	Data	Description					
D9212	Local station operating status	Stores the status of No.1 to No.16	Stores the station No. of all local stations that are in a STOP or PAUSE status. DEVICE NUMBER b15 b14 b13 b12 b11 b10 b9 b8 b7 b6 b5 b4 b3 b2 b1 b0 D9212 L16 L15 L14 L13 L12 L11 L10 L9 L8 L7 L6 L5 L4 L3 L2 L1 D9213 0 L31 L30 L29 L28 L27 L26 L25 L24 L23 L22 L21 L20 L19 L18 L1					
D9213		Stores the status of No.17 to No.31	 When the status of a local station changes to STOP or PAUSE, the corresponding bit is set to "1". Example: When the operation status of No. 7 changes to the STOP status, "1" is set to bit 6 of D9212. When D9212 is monitored, its value is "64 (40H)". 					
D9216	216 Local station error detection status 217	Stores the status of No.1 to No.16	• Stores the numbers of the station that detect the occurrence of an error at another station. $\begin{array}{r rrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrr$					
D9217		Stores the status of No.17 to No.31	 operating station is set to "1". The bit status of remote i/O statio always remains "0". Example: When No. 5 detects that No. 4 is faulty, "1" is set to bit 4 of D9216. When D9216 is monitored, its value is "16 (10H)". When the faulty station recovers normal operating status or whe the loop line is switched so that the data link returns to normal operating status, the corresponding bit is automatically reset to "0". 					
D9220	Local station parameter mismatched or	Stores the status of No.1 to No.16	Stores the numbers of the stations at which a link parameter error sent from the master station is detected by another local or remote I/O station. DEVICE DEVICE					
D9221	remote I/O station input/output allocation error	Stores the status of No.17 to No.31	 D9221 0 31 30 29 28 27 26 25 24 23 22 21 20 19 18 17 Example: When local station No. 5 is set as a remote I/O station in the link parameter settings "1" is set to bit 4 of D9220. When D9220 is monitored, its value is "16 (10_H)". When the link parameter settings is corrected and the status of the master station is switched from STOP to RUN, the bit is automatically reset to "0". 					

Device Number	Name	Data	Description		
D9224	Initial communication between local stations/remote I/O stations	Stores the status of No.1 to No.16	Stores station Nos. of local stations or remote I/O stations communicating initial setting data for data link processing into the corresponding bits in D9224 to D9227 as shown below. DEVICE Bit DEVICE Bit DEVICE DEVICE DIS224 URIGLR15LR14LR13LR12LR11LR10LR9 UR8 UR7 UR8 UR5 UR3 UR		
D9225		Stores the status of No.17 to No.31	setting data (link parameters), the bit corresponding to the station number is set. Example: When No. 23 is communicating initial setting data (link parameters), 1 is set to bit 6 of D9225. When D9225 is monitored, its value is 64 (40H). • When the initial setting data has been communicated, the bit is automatically reset to "0".		
D9228	Local station/	Stores the status of No.1 to No.16	 Stores the number of the local stations/remote I/O stations in the data link that is determined by a master station to be faulty. A station is determined to be faulty if the data returned from it to the master station is not received within the specified length of time. DEVICE Bit NUMBER bis bi4 bi3 bi2 bi1 bi0 bi8 bi7 bi8 bi5 bi4 bi3 bi2 bi1 bi0 bi9 bi8 bi7 bi8 bi5 bi4 bi3 bi2 bi1 bi0 bi9 bi8 bi7 bi8 bi5 bi4 bi3 bi2 bi1 bi0 bi9 bi8 bi7 bi8 bi7 bi8 bi5 bi4 bi3 bi2 bi1 bi0 bi9 bi8 bi7 bi8 bi7 bi8 bi5 bi4 bi3 bi2 bi1 bi0 bi9 bi8 bi7 bi8 bi7 bi8 bi7 bi8 bi7 bi8 bi7 bi8 bi8 bi7 bi8 bi7 bi8 bi8 bi4 bi3 bi2 bi1 bi0 bi9 bi8 bi7 bi8 bi7 bi8 bi8 bi4 bi3 bi2 bi1 bi0 bi9 bi8 bi7 bi8 bi7 bi8 bi8 bi7 bi8 bi8 bi4 bi3 bi2 bi1 bi0 bi9 bi8 bi7 bi8 bi4 bi3 bi2 bi1 bi0 bi9 bi8 bi7 bi8 bi8 bi4 bi3 bi2 bi1 bi0 bi9 bi8 bi7 bi8 bi8 bi4 bi3 bi2 bi1 bi0 bi9 bi8 bi7 bi8 bi8 bi8 bi4 bi3 bi2 bi1 bi0 bi9 bi8 bi7 bi8 bi8 bi4 bi3 bi2 bi1 bi0 bi9 bi8 bi7 bi8 bi8 bi4 bi3 bi2 bi1 bi0 bi9 bi8 bi7 bi8 bi8 bi8 bi8 bi8 bi8 bi8 bi8 bi4 bi8		
D9229	error	Stores the status of No.17 to No.31	 remote I/O station is set. Example: When an error at No. 3 causes it to fail to return the data to the master station, "1" is set for bit 2 of D9228. When D9228 is monitored, its value is "4". When the loop line becomes faulty, "1" is set for the bits of the stations after the fault or for all local stations/remote I/O stations. When the master station becomes faulty or the setting for the monitoring time is too small, "1" is set for the bits corresponding to all local stations/remote I/O stations. When the faulty station returns to normal, the bit is automatically reset to "0". 		
D9240	Receive error detection count	Stores the total number of receive errors	 Stores the number of times that the following errors are detected in the loop line currently being used: "CRC", "AB.IF", "OVER" Counting stops if the number of receive error occurrences exceeds the maximum limit "FFFFH". 		

Table 9.6 List of MELSECNET/B special link registers (Continued)

• Execute the reset operation to clear the data to "0".

MELSEC-A

MELSECNET mode	MELSECNET			MELSECNET/B		
Operating Mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode
Applicability	0	0	0	0	0	0

MELSEC-A

9.3.2 Special link registers effective only for local stations

Table 9.7 and Table 9.8 show the special registers controlled only when the host station is set to the master station.

Table 9.7 List of MELSECNET special link registers

Device Number	Name	Data	Description
D9243	Station number data of the station itself	Stores a station number (0 to 64)	Stores the station number assigned to the host station.Used by a local station to check the host station number.
D9244	Slave station number data	Stores a slave station number	Used by a local station to check the total number of slave stations in the loop.
D9245	Receive error detection count	Stores the accumulated total number of receive error occurrences	 Stores the accumulated number of times that the following errors are detected in the loop line currently being used: "CRC", "OVER", "AB.IF" Errors are counted up to "FFFF_H" and then counting is stopped. Execute the reset operation to clear the data to "0".
D9248		Stores the status of No.1 to No.16	Stores station numbers of the local stations, excluding the host station, whose status is either STOP or PAUSE. DEVICE NUMBER b15 b14 b13 b12 b11 b10 b9 b8 b7 b6 b5 b4 b3 b2 b1 b0 D9248 L16 L15 L14 L13 L12 L11 L10 L9 L8 L7 L6 L5 L4 L3 L2 L1
D9249		Stores the status of No.17 to No.32	D9249 L32 L31 L30 L29 L28 L27 L26 L25 L24 L22 L21 L20 L19 L18 L17 D9250 L48 L47 L46 L45 L44 L42 L41 L40 L39 L38 L37 L36 L35 L34 L33 D9251 L64 L63 L62 L61 L60 L59 L58 L57 L56 L54 L53 L52 L51 L50 L49 • If a local station except for the host station goes down, data before the failure will be held. Edd. L64
D9250	Local station operating status	Stores the status of No.33 to No.48	 When the corresponding bit in D9252 to D9255 is "0", the relevant bit in the above special registers is validated. Even If the host station (master station) goes down, the data before the failure will be also held. When the status of a local station is either STOP or PAUSE, the corresponding bit is set to "1". When the status of the local station changes
D9251		Stores the status of No.49 to No.64	 to RUN or STEP RUN, the bit is automatically reset to "0". The bit status of remote I/O station always remains "0". Example:When the statuses of local stations No.7 and No.15 are either STOP or PAUSE, "1" is set to bit 6 and bit 14 of D9248. When D9248 is monitored, its value is "16448 (4040_H)". The bit corresponding to the host station is not set to "1" regardless of the status of the status of the status of stored)
D9252		Stores the status of No.1 to No.16	 Stores station Nos. of faulty local stations except for the host station in the loop, into the corresponding bits in the special registers (for link) as shown below Only a faulty local station can be detected by another local station. The bit status of remote I/O station always remains "0".
D9253		Stores the status of No.17 to No.32	DEVICE Bit NUMBER b15 b14 b13 b12 b11 b10 b9 b8 b7 b6 b5 b4 b3 b2 b1 b0 D9252 L16 L15 L14 L13 L12 L11 L10 L9 L8 L7 L6 L5 L4 L3 L2 L1
D9254	Local station error status	Stores the status of No.33 to No.48	D9253 L32 L31 L30 L29 L28 L27 L26 L25 L24 L22 L21 L20 L19 L18 L17 D9254 L48 L47 L46 L45 L44 L42 L41 L40 L39 L38 L37 L36 L35 L34 L33 D9255 L64 L63 L62 L61 L60 L59 L57 L56 L54 L53 L52 L51 L50 L49 • Bits corresponding to the station Nos. of faulty local stations (except for host station) turn to "1". "1"."
D9255		Stores the status of No.49 to No.64	 Example:When local station No. 12 is faulty, "1" is set to bit 11 of D9252. When D9252 is monitored, its value is "2048 (500H)". When the faulty station recovers normal operating status or when the loop line is switched so that the data link returns to normal operating status, the bit is automatically reset to "0".

Table 9.8 List of MEL	SECNET/B	special	link reaisters
Table the Electric Hille	OLONE I/D	opoolai	mint regiotore

Table 9.8 List of MELSECNET/B special link registers								
Device Number	Name	Data	Description					
D9243	Station number data of the station itself	Stores a station number (0 to 64)	 Stores the station number assigned to the host station itself. Used by a local station to check the host station number. 					
D9244	Slave station number data	Stores a slave station number	Used by a local station to check the total number of slave stations in the loop.					
D9245	Receive error detection count	Stores the accumulated total number of receive errors	 Stores the accumulated number of times that the following errors are detected in the loop line currently being used: "CRC", "OVER", "AB.IF" Errors are counted up to "FFFF_H" and then counting is stopped. Execute the reset operation to clear the data to "0". 					
D9248		Stores the status of No.1 to No.16	 Stores station numbers of the local stations, excluding the host station , whose status is either STOP or PAUSE. DEVICE Bit D9248 L16 L15 L14 L13 L12 L11 L10 L9 L8 L7 L6 L5 L4 L3 L2 L1 D9249 0 L31 L30 L29 L28 L27 L26 L25 L24 L23 L22 L21 L20 L19 L18 L17 If a local station except for the host station goes down, data before the failure will be held. When the corresponding bit in D9252 to D9255 is "0", the relevant bit in the above special registers is validated. Even If the host station (master station) goes down, the data 					
D9249	Local station operating status	Stores the status of No.17 to No.31	 before the failure will be also held. When the status of a local station is either STOP or PAUSE, the corresponding bit is set to "1". When the status of the local station changes to RUN or STEP RUN, the bit is automatically reset to "0". The bit status of remote I/O station always remains "0". Example: When the statuses of local stations No.7 and No.15 are either STOP or PAUSE, "1" is set to bit 6 and bit 14 of D9248. When D9248 is monitored, its value is "16448 (4040H)". The bit corresponding to the station itself is not set regardless of the status of the status of the station itself. (always "0" is stored) 					
D9252	Local station error	Stores the status of No.1 to No.16	 Detects station Nos. of faulty local stations except for the host station in the loop and then stores the data into the corresponding bits in the special registers (for link) as shown below. Error detection is performed only for local stations other than the host station. The status of remote I/O stations remains "0". 					
D9253	status	Stores the status of No.17 to No.31	 Bits corresponding to the station Nos. of faulty local stations (except for host station) turn to "1". Example:When local No. 12 is faulty, "1" is set to bit 11 of D9252. When D9252 is monitored, its value is "2048 (500H)". When the faulty station recovers normal operating status or when the loop line is switched so that the data link returns to normal operating status, the bit is automatically reset to "0". 					

MELSECNET mode	MELSECNET			MELSECNET/B		
Operating Mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode
Applicability	0		0	0		0

MELSEC-A

9.4 Data Link Program Using Link Inputs (X) and Link Outputs (Y)

This section explains the programming method for data link between the master station and local station and between the master station and remote I/O station using link inputs (X) and link outputs (Y).





Fig 9.1 System configuration

[Link device assignment]

		*	LINK *			M : B ↔	ALL L:	B 000-15
MASTER	SLAVE	$M \rightarrow A$		W.D.T.	INTER-	M:W ↔ M:W →	ALL L: ALL R:	W 200-18 W 200-29 W 300-30
MAGTER	STATIONS	в	w	10ms	10ms	$\begin{array}{ccc} M & Y & \rightarrow \\ M & Y & \rightarrow \\ M & Y & \rightarrow \end{array}$	ALL L:	X 260-47 Y 580-7F
м	4	000-05F	000-083	20	xxxx	M:X ← M:X ←	ALL L ALL R	Ý 1A0-38 X 500-76
L/R	M +	- L	M → R	M ← R	M →	L/R1	M ←	· L/R
NO.	В	w	w	w	Y	X/Y	×	Y/X
R 1 L 2 L 3 R 4	060-18F 0E0-15F	0A0-0FF 100-186	200-23F 	300-33F 340-3C1	700-7FF 390-47F 260-36F 580-6AF	030-12F 250-33F 1B0-2BF 080-1AF	6D0-76F 1A0-25F 2A0-3BF 500-5BF	000-09F 210-2CF 300-41F 000-0BF
	-	- - -				- - -		- - -
L : LOCA		1	L		M : MA	ASTER L	: LOCAL	R : REMC

Fig 9.2 Link device assignment

[Program example 1] Data link between the master station and a local station

Master station program Turn Y260 (X1B0 of local station No.3) (Y390 (X250 of local station No.2) ON w Turn YC0 ON while X1A0 (Y210 of local while X2A0 (Y300 of local station No.3)	ON while the T4 contact is OFF and turn hile the T4 contact is ON. I station No.2) is ON, and turn YC1 ON is ON.
	Turns ON/OFF the T4 contact every 2 seconds. (It stops when both X1A0 and X2A0 turn ON.) Turns Y390 ON while the T4 contact is ON.
Jr (Y280) X1A0 (Y000)	 Turns Y260 ON while the T4 contact is OFF. Turns YC0 ON while X1A0 is ON.
Local station No 2 program	Turns YC1 ON while X2A0 is ON.
Turn Y1E0 ON while X250 (Y390 of the Turn Y210 (X1A0 of the master station)	master station) is ON. ON when X250 is ON for 20 times.
X250 (C10)	Counts 20 times that X250 turns from OFF to ON with C10.
	 Turns Y1E0 ON while X250 is ON. T11 times out 5 seconds after C10 counts up.
T11 T11 HET C10	Turns Y210 ON after C11 counts up. Resets C10 when T11 times out.
Local station No.3 program Turn Y80 ON while X1B0 (Y260 of the r Turn Y300 (X2A0 of the master station)	naster station) is ON. ON when X1B0 is turned ON for 20 times.
X1B0 -	Counts 20 times that X1B0 turns from OFF to ON with C20.
C20 K50	Turns Y80 ON while X1B0 is ON.
(Y300)	Turns Y300 ON when C20 counts up.
T21 	Resets C20 when T21 times out.

[Program example No.2]..... Data link between the master station and a remote I/O station

Master station program When X0 (X6D0 of master station) of remote I/O station No.1 is turned ON, Y120 (Y7F0 of the master station) of remote I/O station No.1 and Y170 (Y670 of the master station) of remote I/O station No.4 flash every 1 second. In addition, flashing stops when X0 (X500 of the master station) of remote I/O station No.4 is turned ON. XeDo Stores when X6D0 (X0 of rem 4 SET M10 station 1) is turned ON. K15 M10 Τ7 # Т6 H۲ Turns ON/OFF the T6 contact at intervals K15 T6 of 1.5 seconds while M10 is ON. 17 H۲ Turns Y7F0 (Y120 of remote I/O station Y7F0 4) ON while the T6 contact is ON. Т6 Turns Y670 (Y170 of remote I/O station 1) H Y670 ON while the T6 contact is OFF. X500 Turns M10 OFF when X500 (X0 of RST M10 -11remote I/O station 4) is turned ON.

MELSECNET mode		MELSECNET			MELSECNET/B)
Operating Mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode
Applicability	0	0	0	0	0	0

MELSEC-A

9.5 Data Link Program Using Link Relays (B)

This section presents a program for sequentially turning ON "Y140" of the master station, "YC0" of local station No.2, and "Y70" of local station No.3 by link relays.



Fig 9.3 System configuration

[Link device assignment]

		*	LINK *			M:B. ↔	ALL L:	B 000-15F
MASTER	SLAVE	M -→ A		W.D.T.	INTER-	$\begin{array}{ccc} M : W & \leftrightarrow \\ M : W & \rightarrow \\ M : W & \leftarrow \end{array}$	ALL L: ALL R:	W 000-186 W 200-294 W 200 201
	STATIONS	В	w	10ms	10ms	$\begin{array}{ccc} M & \cdot & \mathbf{V} & \rightarrow \\ M & \cdot & \mathbf{Y} & \rightarrow \\ M & \cdot & \mathbf{Y} & \rightarrow \end{array}$	ALL L:	X 260-301 X 260-47F X 580-7FF
м	4	000-05F	000-083	20	XXXX	M : X + M : X +	ALL L: ALL R:	Y 1A0-3BF X 500-76F
L/R	M +-	- L	M → R	M ← R	M →	L/R1	M ←	· L/R
NO.	В	w	w	w	Y	X/Y	x	Y/X
R 1 L 2 L 3 R 4	060-18F 0E0-15F - -	0A0-0FF 100-186 - - - -	200-23F 	300-33F 	700-7FF 390-47F 260-36F 580-6AF - - - -	030-12F 250-33F 1B0-2BF 080-1AF - - -	6D0-76F 1A0-25F 2A0-3BF 500-5BF - - - -	000-09F 210-2CF 300-41F 000-0BF - - - -
L : LOCA R : REMO	NL DTE				M : MA	ASTER L	: LOCAL	R : REMOTE

Fig 9.4 Link device assignment

[Program example]



MELSECNET mode		MELSECNET			MELSECNET/B	
Operating Mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode
Applicability	0	0	0	0	0	0

MELSEC-A

9.6 Data Link Program Using Link Registers (W)

The following describes a program where the link register contents (0 to 10) written by the master station are read by the local station No.2 and "YD0" to "YD2" are turned ON/OFF according to the contents.



Fig 9.5 System configuration

[Link device assignment]

	SLAVE	M → A	+ LINK	* W.D.T.	INTER-	M:B M:W M:W	ALL L ALL L ALL B	: B 000-15F : W 000-186 : W 200-294
MASTER	PC STATIONS	В	W	FOR LINK 10ms	MITTENT 10ms	M:W + M:Y -	ALL R	W 300-3C1 X 260-47F
M	4	000-05F	000-083	20	XXXX	M:X M:X	- ALL L - ALL R	Y 1A0-3BF X 500-76F
L/R	M +	- L	M → R	M ← R	M -	• L/R1	M +	- L/R
NO.	в	w	w	w	Y	X/Y	x	Y/X
R 1 L 2 L 3 R 4	060-18F 0E0-15F - - -	0A0-0FF 100-186 - - - -	200-23F 	300-33F 	700-7FF 390-47F 260-36F 580-6AF - - -	030-12E 250-33F 1B0-2BF 080-1AF - - - - -	6D0-76F 1A0-25F 2A0-3BF 500-5BF - - - - -	000-09F 210-2CF 300-41F 000-0BF - - - - -
Î L : LOCA R : REMO				, , , , ,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	M : MA	STER L	: LOCAL	R : REMOTE

Fig 9.6 Link device assignment

[Program example]

Master station program	
A program to store "0 to 10" to W0.	
(B10 and B11 are used for handshaking of the	e master station and local station No.2.)
	Furns the initialization flag (M1) of W0 ON at he start of CPU module run and 10 < W0.
М1 	Initializes W0 data to "0".
(B010)	Turns B10 ON when the initialization of the W0 data has been completed.
T2	The value of W0 is increased by "1" every
M2 M1	6 seconds.
	Turn B11 ON when an INCP instruction has been executed (addition of "1" to W0 data).
	Turns M1 and M2 OFF when the W0 data has been read by local station No. 2 (B70 turned ON)
Local station No.2 program	
Turn YD0, YD1, and YD2 ON when the W0 d	ata is transferred from the master
station	
ON YD0 $VV0 < 5$	
ns: YD2 W0 > 5	
B010 	Transfers W0 data to D0 by initialization of W0 on the master station or upon completion of INCP instruction execution.
B011 B070	Turns B70 ON upon completion of transferring W0 data to D0.
- <u>> к5 D0</u> - (7000-	Turns YD0 ON if D0 < 5.
	Turns YD1 ON if D0 = 5.
	Turns YD2 ON if D0 > 5 (both YD0 and YD1 OFF).

Operating Mode MELSECNET MELSECNET I MELSECNET I MELSECNET I MELSECNET I mode mode mode mode composite mod	MELSECNET mode		MELSECNET			MELSECNET/B	
	Operating Mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode
Applicability O O O O O	Applicability	0	0	0	0	0	0

MELSEC-A

9.7 Read/Write Program for a Word Device from the Master Station to a Local Station

The following describes a program where data is read from/written to the word device of the local station from the master station using the LRDP/LWTP instruction.

[System configuration] [For MELSECNET data link system] Remote I/O station Local station Local station Remote I/O station Master station No. 1 No. 2 No. 3 No. 4 XO XAO X30 Y30 X80 Y80 хo X20 X80 Yn XAC YO Y70 CPI P2 J7: P25 P25 XAF XF 22 XF/ X2F X4F Y4F Y7F Y9F X1 X9F Y9F XA0 YA0 YB0 Y130 Y120 Y150 Y120 YAC YCC 110 **Y**12F 16F Y12F Y 13F X12F XBF YBF Y130 Y190 Y1A0 Y140 170 Y200 Y130 Y190 Y13F Y1AF 15 Y19 Y17F Y20F Y14F Y19F I O S Input module : : : Output module Special function module [For MELSECNET/B data link system] Master station Remote I/O station Local station Local station Remote I/O station No. 1 No. 2 No. 3 No. 4 ×30 χo X20 ¥30 X80 ¥80 **X8**0 Y8C A.17 XA T25 T25E 2 XF X4F X9F Y4F Y7F Y9F XB Y71 X11 X9F Y9F X2F XA0 130 Y120 YC Y150 X110 YA 100 Y12F YAF XBF YBF YBF Y13F YCF Y16F Y8F X12F Y12F Y200 1A0 Y17F 156 Y19 Y20F Y14F Y19F Y13F Y1AF Input module Output module Special function module : : : Ò S

Fig 9.7 System configuration
[Link device assignment]

 			* LINK	*		M : B +	ALL L	: B 000-15F		
MASTER	SLAVE	M → A		W.D.T.		M:W -	ALL R	: W 200-294 : W 300-3C1		
MASTEN	STATIONS	В	w	10ms	10ms	M:Y -	→ ALL L → ALL R	: X 260-47F : Y 580-7FF		
м	4	000-05F	000-083	20	xxxx	M:X + M:X +	- ALL L - ALL R	Y 1A0-3BF X 500-76F		
					· · · · · · · · · · · · · · · · · · ·					
L/R	M ←	- L	M → H	M + H	M -	· L/H1	M ←	L/M		
NO.	В	w	w	w	Y	X/Y	×	Y/X		
R 1 L 2 L 3 R 4	060-18F 0E0-15F - - -	0A0-0FF 100-186 - - -	200-23F 	300-33F 	700-7FF 390-47F 260-36F 580-6AF - - -	030-12F 250-33F 1B0-2BF 080-1AF - - - -	6D0-76F 1A0-25F 2A0-3BF 500-5BF - - - -	000-09F 210-2CF 300-41F 000-0BF - - - -		
M : MASTER L : LOCAL R : REMOTE R : REMOTE										

Fig 9.8 Link device assignment

 Read program (LRDP instruction) The following describes program where the present value of C50 to C55 of local station No.2 is read to D100 to D105 of the master station. (For system configuration and link parameter setting, refer to Fig 9.7 and Fig 9.8.)





(2) Write program (LWTP instruction) The following describes the program where the present value of C0 to C9 of the master station is written to C0 to C9 of the local station No.3. (For system configuration and link parameter setting, refer to Fig 9.7 and Fig 9.8.)

[Program example]



MELSECNET mode		MELSECNET		MELSECNET/B			
Operating Mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	
Applicability	0		0	0		0	

MELSEC-A

9.8 Read/Write Program from a Remote I/O Station to a Special Function Module

This section describes the programming method to read/write data of a special function module installed to a remote I/O station from the master station.

[System configuration]



Fig 9.9 System configuration

[Link device assignment]

			+ LINK	*		M : B ↔	ALL L:	B 000-15F		
MARTER	SLAVE	$M \rightarrow A$	LLL W.D.T.		INTER-	$M: W \rightarrow M: W \rightarrow M: W \leftarrow$	ALL R:	W 200-294 W 300-3C1		
MASIEN	STATIONS	В	w	10ms	10ms	$ \begin{array}{ccc} M & Y & \rightarrow \\ M & Y & \rightarrow \\ M & Y & \rightarrow \end{array} $	ALL L: ALL R:	X 260-47F Y 580-7FF		
м	4	000-05F	000-083	20 XXXX		M:X ← M:X ←	ALL L: ALL R:	Y 1A0-3BF X 500-76F		
L/B			M→R M←R		м –	• L/R1	M ← L/R			
ÑÓ.	В	w	w	w	Y	X/Y	х	Y/X		
R 1 L 2 L 3 R 4	060-18F 0E0-15F - -	0A0-0FF 100-186 - -	200-23F 	300-33F 	700-7FF 390-47F 260-36F 580-6AF - - -	030-12F 250-33F 1B0-2BF 080-1AF - - - - -	6D0-76F 1A0-25F 2A0-3BF 500-5BF - - - -	000-09F 210-2CF 300-41F 000-0BF - - - - -		
M : MASTER L : LOCAL R : REMOTE R : REMOTE										

Fig 9.10 Link device assignment

REMARK

In the M \rightarrow R area of remote I/O stations No.1 and No.4, two points (W200 and W201, W250 and W251) from the head device are used by the system. These points cannot be used for a user program. (Refer to Section 7.6.3.)

MELSECNET mode		MELSECNET		MELSECNET/B			
Operating Mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	
Applicability	0		0	0		0	

MELSEC-A

9.8.1 Read program (RFRP instruction)

This is a program to read the data of the special function module mounted on the remote I/O station No. 1.

(For system configuration and link parameter setting, refer to Fig 9.9 and Fig 9.10.)







NOTE

(1) To execute an RFRP instruction, always interlock with YnE and X(n+1) E (n: the device number of the master station that corresponds to the I/O number of the slot in which a special function module is installed. It is first 2 digits of the 3-digit representation for the first half link parameters 16 points of 32 occupied points.), and at the same time, do not execute an RFRP or RTOP instruction at two or more positions within a single special function module.

Turn ON/OFF YnE and X(n+1) E as shown below:



- (2) The RFRP execution start signal must be always turned ON by the SET instruction. If an OUT or PLS instruction is used, the RFRP instruction will not be executed correctly.
- (3) Reset and initialize YnE and the RFRP execution start signal after the execution is completed.

When failing to do so, reading cannot be executed again.

(4) Refer to the manual for each special function module for addresses where each data of the special function module is stored.

- (5) Write an error detection circuit in a sequence program to check the operation from error occurrence to initial communication completion when an error occurs to a local or remote I/O station.
 - (a) Whether an error has occurred in a local or remote I/O station or not can be determined by 1/0 of bit corresponding to the specified station of D9228 to D9231. (When the bit corresponding to the specified station is "1", it means that an error occurs.)
 - (b) Whether the initial communication is being executed or not can be determined by 1/0 of bit corresponding to the specified station of D9224 to D9227. (When the bit corresponding to the specified station is "1", it means that initial communication is being executed.)
 - (c) The occurrence of an error and the execution of initial communication in a local station or a remote I/O station are detected in the following timing.



- (d) The error detection program must be written before the initial communication detection program.
 If initial communication detection program is written first, the occurrence of errors and the execution of initial communication may not be detected.
- (6) When RFRP and RTOP instructions cannot be executed because the special function module is faulty, X(n+1)D is turned ON.
 - (a) When YnD is turned ON, X(n+1)D is turned OFF.
 - (b) If X(n+1)D is turned ON, a special function module might be faulty or the module might not be mounted correctly. Check the special function module at which the error occurred.
- (7) Write the following timing circuit to turn YnD ON/OFF with SET/RST instruction.
 - (a) Turn YnD ON, when X(n+1)D is turned ON.
 - (b) Turn YnD OFF only once, when X(n+1)D is turned OFF

MELSECNET mode		MELSECNET		MELSECNET/B			
Operating Mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	
Applicability	0		0	0		0	

MELSEC-A

9.8.2 Write program (RTOP instruction)

The following shows the program where data is written to the special function module installed to remote I/O station No.4.

(For system configuration and link parameter setting, refer to Fig 9.9 and Fig 9.10.)







PRECAUTIONS

(1) To execute an RTOP instruction, always interlock with YnF and X(n+1)F (n: the device number of the master station that corresponds to the I/O number of the slot in which a special function module is installed. It is first 2 digits of the 3-digit representation for the first half link parameters 16 points of 32 occupied points.), and at the same time, do not execute an RFRP or RTOP instruction at two or more positions within a single special function module.

Turn ON/OFF YnF and X(n+1)F as shown below:



- (2) The RTOP execution start signal must be always turned ON by the SET instruction. If an OUT or PLS instruction is used, the RTOP instruction will not be executed correctly.
- (3) Reset and initialize YnF and the RTOP execution start signal after the execution is completed.

When failing to do so, reading cannot be executed again.

(4) Refer to the manual for each special function module for addresses where each data of the special function module is stored.

- (5) Write an error detection circuit in a sequence program to check the operation from error occurrence to initial communication completion when an error occurs to a local or remote I/O station.
 - (a) Whether an error has occurred in a local or remote I/O station or not can be determined by 1/0 of bit corresponding to the specified station of D9228 to D9231. (When the bit corresponding to the specified station is "1", it means that an error occurs.)
 - (b) Whether the initial communication is being executed or not can be determined by 1/0 of bit corresponding to the specified station of D9224 to D9227. (When the bit corresponding to the specified station is "1", it means that initial communication is being executed.)
 - (c) The occurrence of an error and the execution of initial communication in a local station or a remote I/O station are detected in the following timing.



(d) The error detection program must be written before the initial communication detection program.

If initial communication detection program is written first, the occurrence of errors and the execution of initial communication may not be detected.

- (6) When RFRP and RTOP instructions cannot be executed because the special function module is faulty, X(n+1)D is turned ON.
 - (a) When YnD is turned ON, X(n+1)D is turned OFF.
 - (b) If X(n+1)D is turned ON, a special function module might be faulty or the module might not be mounted correctly. Check the special function module at which the error occurred.
- (7) Write the following timing circuit to turn YnD ON/OFF with SET/RST instruction.
 - (a) Turn YnD ON, when X(n+1)D is turned ON.
 - (b) Turn YnD OFF only once, when X(n+1)D is turned OFF.

MELSECNET mode		MELSECNET		MELSECNET/B			
Operating Mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	
Applicability	0	0	0	0	0	0	

MELSEC-A

9.9 Fault Detection Program

The following shows a program with which a faulty station is detected by the master station when an error occurs in any of the local/remote I/O stations No. 1 to 4.



Fig 9.11 System configuration

[Link device assignment]

				+ LINK	*		M:B ↔	ALL L:	B 000-15F	F
	MAGTED	SLAVE M→AL		LL L W.D.T.		INTER-	M:W →	ALL R:	W 200-294 W 300-301	4 1
	MASIEN	STATIONS	В	w	10ms	10ms	M:Y → M:Y →	ALL L:	X 260-47F Y 580-7FF	F F
	M	4	000-05F	000-083	20	XXXX	M:X M:X	ALL L: ALL R:	Y 1A0-3BI X 500-76F	F
	L/B	 M ← L		M→R M←R		M -	► L/R1	M +	- L/R	
	ÑO.	В	w	w	w	Y	X/Y	x	Y/X	
	R 1 L 2 L 3 R 4	060-18F 0E0-15F - - -	0A0-0FF 100-186 - -	200-23F 250-294	300-33F 340-3C1	700-7FF 390-47F 260-36F 580-6AF	030-12F 250-33F 1B0-2BF 080-1AF - -	6D0-76F 1A0-25F 2A0-3BF 500-5BF - - -	000-09F 210-2CF 300-41F 000-0BF - -	
	t L : LOCAI R : REMO	ŤE				M : MA	STER L	: LOCAL	R : REMOTI	E

Fig 9.12 Link device assignment

MELSEC-A

[Program example]



10. TROUBLESHOOTING

MELSECNET mode		MELSECNET		MELSECNET/B			
Operating Mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	
Applicability	0	0	0	0	0	0	

MELSEC-A

10 TROUBLESHOOTING

To improve the reliability of the system, it is very important to use reliable equipment. In addition, the other very important factor is whether or not to recover from failure immediately and certainly.

If a problem occurs during the data link operation, check the link status, following the steps below:

- Perform link monitor by GPP and check the faulty parts When the A7PU is used, monitor the special link relays and special link registers to check the faulty parts.
- (2) Check the LED indicator of link module The "ERROR" LED on the link module turns ON when an error occurs. Check the content of error by "ERROR" LED which is turned ON.
- (3) Check the connection status of data link cable Refer to Section 8.2 to check whether the station number setting and cable connection order are not wrong.

10.1 Link Monitor by GPP

The link status of the data link system loop state, master or slave station status, and the scan time can be checked by using GPP.

Link monitor includes the following three types.

(a)	Master station link monitor	Link monitor when connecting GPP to the master station. Refer to Section 10.1.1
(b)	Local station link monitor	Link monitor when connecting GPP to local stations. Refer to Section 10.1.2
(c)	Remote I/O station link monitor	Link monitor when connecting GPP to remote I/O stations. Refer to Section 10.1.3

10. TROUBLESHOOTING

MELSECNET mode		MELSECNET		MELSECNET/B			
Operating Mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	
Applicability	0	0	0	0	0	0	

MELSEC-A

10.1.1 Master station link monitor

The following describes the link monitor when connecting GPP to the master station.



Fig 10.1 Master station link monitor screen (when connecting to master station)



- (1) Displaying host station operation mode (M9224, M9227)
 - (a) Displays operation status of the master station
 - 1) ON-LINE: Master station mode setting is ON-LINE (with/without automatic return function).
 - 2) OFF-LINE: Master station mode setting is OFF-LINE, SELF-LOOPBACK TEST, or STATION-TO-STATION TEST.
 - LOOP TEST: Master station mode setting is FORWARD LOOP TEST or REVERSE LOOP TEST.
- (2) Displaying loop line status of the host station (M9225, M9226)
 - (a) Displays the forward loop line (F loop) and reverse loop line (R loop) status of the master station.
 - 1) OK: Loop line is normal.
 - 2) NG: Loop line is faulty.
- (3) Displaying link scan time (D9207 to D9209)
 - (a) Displays the time required for data link between the master station and all slave stations.
 - 1) MAX.: Displays the maximum value of time required for data link.
 - 2) MIN.: Displays the minimum value of time required for data link.
 - 3) PRESENT: Displays the present value of time required for data link.
- (4) Displaying loop status of data link system (D9204 to D9206)
 - (a) The Table 10.1 shows the present loop status.



- (b) The loop status and loopback execution station are the same as the following registers:
 - 1) Loop status:..... D9204
 - 2) Loopback execution station:... D9205, D9206
- (5) Operation status of all slave stations (local and remote I/O stations) The status of all of the slave stations in the system is displayed as follows:
 - "L" column: Describes status of present data link
 - "C": Communicating normally (D9224 to D9227)
 - "D": Disconnected due to communication stop (D9224 to D9227) The possible causes to be disconnected are as follows:
 - 1) The power supply to the disconnected station is OFF.
 - 2) The disconnected station is reset.
 - 3) An error that causes operation stop to the programmable controller CPU.
 - A MELSECNET-compatible local or remote I/O station is connected to a station number where MELSECNET II -compatible station (local station) is set with link parameters.
 - <u>The station is disconnected due to loopback processing.</u> Check the loopback with the display of loop status of the data link system described in (4).
 - 6) Incorrect cable wiring (For the optical fiber cable, the IN/OUT connection is incorrect.)
 - "P":Parameter communication with master station(D9228 to D9231) Link parameter communication is executed only once at the time of starting communications.

Possible causes to display "P" continuously are as follows:

- 1) A remote I/O station is connected to a station number set as the MELSECNET mode local station with the link parameters.
- A local station is connected to a station number set as a remote I/O station.
- A remote I/O station is connected to a station number set as a MELSECNET-compatible local station in the MELSECNET II composite mode with link parameters.
- "S" column: Displays present CPU operation status (D9212 to D9215)
 - "R": Run status
 - "S": Stop status

Only "R" is displayed for a remote I/O station.

- "A" column: An error will occur to the master station for the third tier if there is an error in the third tier link parameters set for the corresponding station.(D9220 to D9223)
 When the both inputs (X) and outputs (Y) are not set to a remote I/O station with link parameters or when I/O assignment of the master station and installation position of the I/O module of the host station are wrong, an error will occur.
 "0": Normal
 "E": Error
- "B" column: Displays whether a local station has detected an error at other local stations.(D9216 to D9219)
 - "0": Error has not been detected.
 - "E": Error has been detected.
 - Only "0" is displayed for a remote I/O station.
- "C" column: Displays the forward loop line status of each slave station.(D9232 to D9239) "0": Normal
 - "E": Error
- "D" column: Displays the reverse loop line status of each slave station.(D9232 to D9239) "0": Normal
 - "E": Error



MELSECNET mode		MELSECNET		MELSECNET/B			
Operating Mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	
Applicability O		0	0	0	0	0	

MELSEC-A

10.1.2 Local station link monitor

The following describes the link monitor when connecting GPP to the local station.



Fig 10.2 Local station link monitor screen (when connecting to local station)



- (1) Displaying operation mode of the host station
 - (a) Displays the operation status of host station
 - 1) ON-LINE......Host station mode setting is ON-LINE (automatic return function set/not set).
 - 2) OFF-LINE......Host station mode setting is OFF-LINE, SELF-LOOPBACK TEST, or STATION-TO-STATION TEST.
 - 3) LOOP TEST...Host station mode setting is FORWARD LOOP TEST or REVERSE LOOP TEST.
 - (b) The operation display is the same as contents in M9240 and M9252.
- (2) Displaying communication status of the host station
 - (a) Displays the communication status of the host station.
 - 1) P-MTR WAITING......Awaiting parameter data to be sent from the master station.
 - 2) OK.....Normal communication is being executed.
 - 3) STOP.....The host station is disconnected and communication has been stopped.
 - (b) The communication status is the same as contents in M9250 and M9251.
- (3) Displaying loop line status of the host station
 - (a) Displays the forward loop line (F loop) and reverse loop line (R loop) status of the host station.
 - Only "OK" is displayed when a MELSECNET/B data link system is used.
 - 1) OK.....Loop line is normal.
 - 2) NG.....Loop line is faulty.
 - (b) The loop line status is the same as contents in M9241 and M9242.
- (4) Displaying loopback execution status
 - - 2) NG (not executed)...Loopback has not been executed by the host station.
 - (b) The loopback execution status is the same as content in M9243.
- (5) Displaying B/W/Y receive status
 - (a) Displays whether the link relays (B), link registers (W), and link outputs (Y) are being received from the master station.
 - 1) OK (receiving)......B, W, and Y are being received from the master station by the cyclic communication.
 - 2) NG (not receiving)...B, W, and Y cannot be being received from the master station due to disconnection of host station.
 - (b) The B/W/Y receive status is the same as content in M9246.

- (6) Displaying B/W receive status (local station in three-tier system) (M9247)
 - (a) Displays whether a local station in the three-tier system is receiving the link relays (B) and link registers (W) from the master station in the two-tier system.
 1) OK (receiving)......B and W are being received from the master station in
 - the two-tier system by the cyclic communication.
 - NG (not receiving)...The local station is in a status in which B and W are not being received from the master station in the two-tier system.

Reception will be disabled when M9247 is turned ON.

- (7) Displaying local station operation status (D9248 to 9251, D9252 to D9255)
 - (a) Displays the operation status of the local station.
 - 1) RUN: Station in RUN status
 - 2) STOP: Station in STOP status
 - 3) DOWN: Station is disconnected from the link due to power supply OFF
 - (b) The operation status of the station number part of remote I/O stations will always be RUN regardless of normal status or power supply OFF etc.

MELSECNET mode		MELSECNET		MELSECNET/B			
Operating Mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	MELSECNET mode	MELSECNET II mode	MELSECNET composite mo	
Applicability	0		0	0		0	

10.1.3 Remote I/O station link monitor

This section describes the following link monitor when connecting GPP to a remote I/O station.

- (a) Loop monitor..... Displays the data link status of the host station.
- (b) Batch monitor..... Can batch monitor the devices that communicate data by host station.

Loop Monitor

The following describes the loop monitor when connecting GPP to a remote I/O station.



Fig 10.3 Loop monitor screen

- (1) Displaying operation mode of the host station
 - (a) Displays the operation status of the host station
 - 1) ON-LINE......Mode setting of host station is ON-LINE (with/without automatic return function).
 - 2) OFF-LINE......Mode setting of host station is OFF-LINE, SELF-

LOOPBACK TEST, or STATION-TO-STATION TEST.

- 3) LOOP TEST.......Mode setting of host station is FORWARD LOOP TEST or REVERSE LOOP TEST.
- (2) Displaying communication status of the host station
 - (a) Displays the communication status of the host station.
 - 1) P-MTR WAITING...Awaiting parameter data to be sent from the master station.
 - 2) OK.....Normal communication is being executed.
 - 3) STOP......The host station is disconnected and communication has been stopped.

- (3) Displaying loop line status of the host station
 - (a) Displays the forward loop line (F loop) and reverse loop line (R loop) status of the host station.
 - 1) OK.....Loop line is normal.
 - 2) NG.....Loop line is faulty.
- (4) Displaying loopback execution status
 - (a) Displays whether loopback has been executed by the host station or not.
 - 1) OK (executed)......Loopback has been executed by the host station.
 - 2) NG (not executed)....Loopback has not been executed by the host station.
- (5) Displaying I/O assignment status
 - (a) Displays whether I/O assignment has been performed by the master station or not.
 - 1) YES......I/O assignment has been performed by the master station.
 - 2) NO.....I/O assignment has not been performed by the master station.
- (6) Displaying error numbers
 - (a) Displays error numbers when the following errors occur.

Error Number	Error Description	Counter Measure			
	A special function module other than 32-point	Remove the special function module (any other			
10	modules is loaded on a remote I/O station.	than 32-point modules) from the remote I/O			
		station.			
11	Special function module hardware error.	Replace the special function module.			
	The number of words read by an RFRP	Change the link register (W) range set with the link parameters			
12	instruction exceeds the link register (W) range	Change the number of words to be read by the			
	specified with the link parameters.	RFRP instruction.			
	The number of words to be written by the	Change the link register (W) range set with the			
13	BTOP instruction exceeds the link register (W)	parameters.			
10	range specified with the parameters	Change the number of words to be written by			
		the RTOP instruction.			
14	The RFRP instruction was executed when a	Special function module hardware failure			
	special function module has a hardware failure.				
15	The RTOP instruction was executed when a	Special function module hardware failure			
	special function module has a hardware failure.				
20	Fuse blown in an I/O module	Replace the fuse in the I/O module loaded to the			
	1/0 module verification error (The 1/0 module	remote I/O station.			
	information of the remote I/O station is different				
	from that recognized at power up.)	Check and/or replace the I/O module.Reset the master station or the remote			
21	(1) An I/O modulo is not socuro				
	(1) An I/O module was removed, or another	I/O station.			
	I/O module was loaded during operation				
	Neither inputs (X) nor outputs (Y) have been	Check the I/O modules ON the remote I/O			
22	specified with the parameters.	station and set the parameters again.			
23	I/O assignment error in GPP A7I MS/GPP	Check the link assignment for the I/O modules			
	function)	and the master station and correct the I/O			
		assignment.			
		Check if local station data are set for a remote			
24	Remote I/O station specification error.	I/O station number with link parameters, and			
		correct the settings.			

- (7) Displaying station number Displays the station number of the remote I/O station connected to the GPP.
- (8) Displaying number of slave stations
 Displays the total number of local and remote I/O stations connected in the loop.

Batch Monitor

The following describes the batch monitor when connecting GPP to a remote I/O station.



Fig 10.4 Batch monitor screen

REMARK

Use SW SRV-GPPA/SW IVD-GPPA when connecting a peripheral device to the remote I/O station.

GX Developer cannot be connected to the remote I/O station.

- (1) Displaying ON/OFF status of inputs (X)
 - (a) Displays the ON/OFF status of inputs (X) sent from a remote I/O station to the master station.
 - 1)
 ON status
 - 2) □.....OFF status
 - (b) If the remote I/O station is online, the device numbers at the master station are used for the display.If the remote I/O station is offline, the device numbers at the host station are used for the display.
- (2) Displaying ON/OFF status of outputs (Y)
 - (a) Displays the ON/OFF status of outputs (Y) sent from the master station to a remote I/O station.
 - 1)
 ON status
 - 2) □.....OFF status
 - (b) If the remote I/O station is online, the device numbers at the master station are used for the display.If the remote I/O station is offline, the device numbers at the host station are

used for the display.

- (3) Displaying link registers (W) for RFRP instructions
 - (a) Displays the data in the area set for data transmission from the host station to the master station by link parameter setting.
- (4) Displaying link registers (W) for RTOP instructions
 - (a) Displays the data in the area set for data transmission from the master station to the host station by link parameter setting.

10. TROUBLESHOOTING

MELSECNET mode	MELSECNET			MELSECNET/B		
Operating Mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode
Applicability	0	0	0	0	0	0

MELSEC-A

10.2 Monitoring the Special Link Relays and Special Link Registers

Faulty stations in the MELSECNET data link system can be detected by using the A7PU to monitor the special link relays and special link registers.

When the GPP/IBM PC/AT is available, detect them by using the link monitoring in Section 10.1.

Refer to Section 9.2 and Section 9.3 for details on special link relays and special link registers.



ľ	MELSECNET mode	MELSECNET			MELSECNET/B		
	Operating Mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode
l	Applicability	0	0	0	0	0	0

10. TROUBLESHOOTING

ME	LSE	C-A
----	-----	-----

10.3 Procedure for Troubleshooting

The following describes the simple procedure for troubleshooting.

10.3.1 Troubleshooting flowchart



*: When mode setting, station number setting, or baud rate setting for link module is not normal, 'SP UNIT LAY ERR' causes by switching a CPU module from STOP to RUN.

MELSECNET mode	MELSECNET			MELSECNET/B			
Operating Mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	
Applicability	0	0	0	0	0	0	

MELSEC-A



10.3.2 Flowchart for when "the data link is disabled throughout the entire system"



POINT

(1) In the MELSECNET data link, if the power supply of the local or remote I/O station on both sides of a normally operating local or remote I/O station is turned off simultaneously (within 100 ms), the data link for the entire system might be disabled.

If the "automatic return function" is set for these stations, data communication resumes immediately. However, if the automatic return function is not set for these stations, they will remain disconnected. To return them to the data link, perform reset operation.

Example:



If the power supply to L1 and R3 is turned off simultaneously (within 100 ms) while R2 is operating normally, the data link for the entire system might be disabled.

10. TROUBLESHOOTING

MELSECNET mode	MELSECNET			MELSECNET/B			
Operating Mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	
Applicability	0	0	0	0	0	0	

MELSEC-A

10.3.3 Flowchart for when "the data link is disabled at a specific station"





MELSECNET mode	MELSECNET			MELSECNET/B			
Operating Mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	
Applicability	0	0	0	0	0	0	



10.3.4 Flowchart for when "the data communication error occurs"



MELSEC-A


MELSEC-A



MELSEC-A



MELSEC-A



10. TROUBLESHOOTING

I	MELSECNET mode	MELSECNET			MELSECNET/B		
	Operating Mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode
	Applicability	0	0	0			

MELSEC-A

10.3.5 Flowchart for when "the unspecified number of slave stations become faulty"



10. TROUBLESHOOTING

MELSECNET mode	MELSECNET			MELSECNET/B		
Operating Mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode	MELSECNET mode	MELSECNET II mode	MELSECNET II composite mode
Applicability	0	0	0	0	0	0

MELSEC-A

10.4 ERROR LED

The following describes LED which turns ON when an error occurs during data link execution.

Indication	Name	Error- detected status	Description
CRC	CRC error (Cyclic redundancy check)		Code check error of receive data <cause> • Data-sending station was disconnected at the timing. • Cable fault, noise, etc.</cause>
OVER	Overrun error		Received data were overwritten with another data received next due to delay in loading. <cause> Hardware failure in receiving part of link module In the system composites with local stations and remote I/O stations, the "OVER" LED of remote I/O stations are turned ON dimly. However, note that the module is not faulty.</cause>
AB.IF	Abort invalid frame error	ON	 "1" has been received consecutively more than stipulated times. Receive data length is shorter than stipulated length. <cause></cause> Data-sending station was disconnected at the timing. Monitoring time too short, cable fault, noise, etc.
TIME	Time check error		Data link monitoring time is over. <cause> Monitoring time too short, cable fault, noise, etc.</cause>
DATA	Data check error		Data containing erroneous code have been received. <cause> Cable fault, noise, etc.</cause>
UNDER	Underrun error		Internal processing of send data is not executed constantly. <cause> Hardware failure in sending part of link module</cause>
F.LOOP	Forward loop error		Forward loop line has an error. Adjacent station was powered OFF. <cause> Forward loop cable disconnection, or incomplete cable connection</cause>
R.LOOP	Reverse loop error		Reverse loop line has an error. Adjacent station was powered OFF. <cause> Reverse loop cable disconnection, or incomplete cable connection</cause>

10.5 Replacing a Faulty Station of Link Module

This section explains how to replace the link module of a faulty station in data link system.

When a MELSECNET data link system is used

Since the link cables are doubled in a MELSECNET data link system, the loopback function can continue data link operations even if power supply to one station (local or remote I/O station) is turned OFF.

Replace the link module of a faulty station as shown below:



When a MELSECNET/B data link system is used

In a MELSECNET/B data link system, since the link cables for both send and receive operations are connected to the same terminal block, data link operations can be continued even if power supply to one of the local stations is turned OFF. Replace the link module of a faulty station as shown below:



MEMO		

INDEX

[A]

Assignment of link devices for a Three-Tier S	ystem
	. 7-61
Automatic return function5-4	,5-19

[C]

Coaxial cable	5-37,8-12
Communication speed	8-7
Comparison (MELSECNET mode, MEL	SECNET II
mode, MELSECNET II composite mode	e) 1-6
Comparison (MELSECNET, MELSECN	IET/B)
	1-5
Cyclic transmission function	3-2,5-6

[D]

Data link modules	
Data link system when the Second tier is	
MELSECNET and the Third tier is MELSE	CNET/
В	. 4-13
Data link system when the Second tier is	
MELSECNET/B and the Third tier is	
MELSECNET	. 4-22
MELSECNET data link system	. 2-10
MELSECNET/B data link system	3-7
Three-tier System using the MELSECNET	data
link system	4-6
Three-tier System using the MELSECNET/	B data
link system	. 4-29
Differences between QCPU and A/QnACPU	local
stations	. 1-11

[E]

ERROR	LED	 10-25

[F]

Fault detection function	5-24
Forward loop test	8-21
Functions	5-4

[L]

Link data communication processing	6-1,6-17
Link data during a communication error	6-5
Link parameters	7-11
Using MELSECNET II composite mode	•
	7-14,7-48

Using MELSECNET II mode	7-12,7-42
Using MELSECNET mode	7-11,7-24
Link refresh execution timing	6-2
Link refresh time	6-12
Link register	5-29,7-21
Link relay	5-29,7-20
Local station	1-3
Local/remote I/O system assignment.	7-52
Loopback function	5-20

[M]

. 1-3,2-1,3-1
5-2,7-19
. 1-1,1-4,2-1
5-34
5-32
. 1-1,1-4,3-1
7-17

[N]

Network refresh parameters	7-4
----------------------------	-----

[0]

Operation mode	1-1
Optical fiber cable	5-36,8-10

[P]

Performance Specifications	5-2
Precaution	
Data link system when the Second tier is	
MELSECNET and the Third tier is MELSE	CNET/
В	4-11
Data link system when the Second tier is	
MELSECNET/B and the Third tier is	
MELSECNET	4-20
MELSECNET data link system	2-9
MELSECNET/B data link system	3-6
Three-tier System using the MELSECNET	data
link system	4-4
Three-tier system using the MELSECNET/I	3 data
link system	4-28
Preparatory steps before operation	8-1
Programming	9-1

[R]

Remote I/O station	1-3
Remote I/O system assignment	7-50
Reverse loop test	8-21

[S]

Self-diagnostics function	. 5-28,8-17
Self-loopback test	8-17
Setting of the Number of Modules	
Setting the link module station numbers	8-2
Slave stations	2-1,3-1
Special link register	9-12
Special link relay	9-4
Specification	5-1
Station to station test	8-19

[T]

Three-tier system	4-1
Transient transmission function	3-3,5-16
Transmission delay time	6-7,6-20
Troubleshooting	10-1
Communication error occurrence in the	е
unspecified number of slave stations	10-24
Data link disabled at a specific station	10-17
Data link disabled throughout the entir	e system
	10-15
Error occurrence in data communication	on 10-19
Twisted pair cable	5-41,8-14
Two-tier system	2-7,3-5

[W]

Niring 8	3-8
----------	-----

WARRANTY

Please confirm the following product warranty details before using this product.

1. Gratis Warranty Term and Gratis Warranty Range

If any faults or defects (hereinafter "Failure") found to be the responsibility of Mitsubishi occurs during use of the product within the gratis warranty term, the product shall be repaired at no cost via the sales representative or Mitsubishi Service Company.

However, if repairs are required onsite at domestic or overseas location, expenses to send an engineer will be solely at the customer's discretion. Mitsubishi shall not be held responsible for any re-commissioning, maintenance, or testing on-site that involves replacement of the failed module.

[Gratis Warranty Term]

The gratis warranty term of the product shall be for one year after the date of purchase or delivery to a designated place.

Note that after manufacture and shipment from Mitsubishi, the maximum distribution period shall be six (6) months, and the longest gratis warranty term after manufacturing shall be eighteen (18) months. The gratis warranty term of repair parts shall not exceed the gratis warranty term before repairs.

[Gratis Warranty Range]

- (1) The range shall be limited to normal use within the usage state, usage methods and usage environment, etc., which follow the conditions and precautions, etc., given in the instruction manual, user's manual and caution labels on the product.
- (2) Even within the gratis warranty term, repairs shall be charged for in the following cases.
 - 1. Failure occurring from inappropriate storage or handling, carelessness or negligence by the user. Failure caused by the user's hardware or software design.
 - 2. Failure caused by unapproved modifications, etc., to the product by the user.
 - 3. When the Mitsubishi product is assembled into a user's device, Failure that could have been avoided if functions or structures, judged as necessary in the legal safety measures the user's device is subject to or as necessary by industry standards, had been provided.
 - 4. Failure that could have been avoided if consumable parts (battery, backlight, fuse, etc.) designated in the instruction manual had been correctly serviced or replaced.
 - 5. Failure caused by external irresistible forces such as fires or abnormal voltages, and Failure caused by force majeure such as earthquakes, lightning, wind and water damage.
 - 6. Failure caused by reasons unpredictable by scientific technology standards at time of shipment from Mitsubishi.
 - 7. Any other failure found not to be the responsibility of Mitsubishi or that admitted not to be so by the user.

2. Onerous repair term after discontinuation of production

- (1) Mitsubishi shall accept onerous product repairs for seven (7) years after production of the product is discontinued.
 - Discontinuation of production shall be notified with Mitsubishi Technical Bulletins, etc.
- (2) Product supply (including repair parts) is not available after production is discontinued.

3. Overseas service

Overseas, repairs shall be accepted by Mitsubishi's local overseas FA Center. Note that the repair conditions at each FA Center may differ.

4. Exclusion of loss in opportunity and secondary loss from warranty liability

Regardless of the gratis warranty term, Mitsubishi shall not be liable for compensation of damages caused by any cause found not to be the responsibility of Mitsubishi, loss in opportunity, lost profits incurred to the user by Failures of Mitsubishi products, special damages and secondary damages whether foreseeable or not, compensation for accidents, and compensation for damages to products other than Mitsubishi products, replacement by the user, maintenance of on-site equipment, start-up test run and other tasks.

5. Changes in product specifications

The specifications given in the catalogs, manuals or technical documents are subject to change without prior notice.

Microsoft, Windows, Windows NT, and Windows Vista are registered trademarks of Microsoft Corporation in the United States and other countries.

Pentium is a trademark of Intel Corporation in the United States and other countries.

Ethernet is a trademark of Xerox Corporation.

All other company names and product names used in this manual are trademarks or registered trademarks of their respective companies.

Type MELSECNET, MELSECNET/B Data Link System

Reference Manual

MODEL MELSECNET/B-R-E

13JF70

MODEL CODE

IB(NA)-66350-G(1202)MEE

MITSUBISHI ELECTRIC CORPORATION

HEAD OFFICE : TOKYO BUILDING, 2-7-3 MARUNOUCHI, CHIYODA-KU, TOKYO 100-8310, JAPAN NAGOYA WORKS : 1-14 , YADA-MINAMI 5-CHOME , HIGASHI-KU, NAGOYA , JAPAN

When exported from Japan, this manual does not require application to the Ministry of Economy, Trade and Industry for service transaction permission.